26959

of Canada



National Library Bibliotheque nationale du Canada

CANADIAN THESES ON MICROFICHE

THÈSES C'AN ADIENNES SUR MIEROFICHE

	•
NAME OF AUTHOR/NOM DE L'AUTEUR_ IAN MICHAEL WRIGHT	
TITLE OF THESIS/TITRE DE LA THÈSE MORAL REASONING AND CONDUCT OF	•
SELECTED ELEMENTARY SCHOOL CHILDREN	**
•	
UNIVERSITY/UNIVERSITE ALBERTA	
DEGREE FOR WHICH THESIS WAS PRESENTED/ GRADE POUR LEQUEL CETTE THESE FUT PRÉSENTÉE	
YEAR THIS DEGREE CONFERRED/ANNÉE D'OBTENTION DE CE GRADE 1975	
NAME OF SUPERVISOR/NOM DU DIRECTEUR DE THÈSE Dr. D. MASSEY	

Permission is hereby granted to the NATIONAL LIBRARY OF CANADA to microfilm this thesis and to lend or sell copies of the film.

The author reserves other publication rights, and neither the thesis nor extensive extracts from it may be printed or otherwise reproduced without the author's written permission.

L'autorisation est, par la présențe, accordée 🖗 la BIBLIOTHÈ-QUE NATIONALE DU CANADA de microfilmer cette thèse et de prêter ou de vendre des exemplaires du film.

L'auteur se réserve les autres droits de publication; ni la thèse ni de longs extraits de celle-ci ne doivent être imprimés u autrement reproduits sans l'autorisation écrite de l'auteur.

SIGNED/SIGN DATED PERMANENT ADDRESS/RESIDENCE FIXE

THE UNIVERSITY OF ALBERTA

ر ج

MORAL REASONING AND CONDUCT OF SELECTED

ELEMENTARY SCHOOL CHILDREN

by

IAN MICHAEL WRIGHT

A THESIS

SUBMITTED TO THE FACULTY OF GRADUATE STUDIES AND RESEARCH IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

DEPARTMENT OF ELEMENTARY EDUCATION

EDMONTON, ALBERTA

FALL, 1975

THE UNIVERSITY OF ALBERTA

FACULTY OF GRADUATE STUDIES AND RESEARCH

The undersigned certify that they have read, and recommend to the Faculty of Graduate Studies and Research, for acceptance, a thesis entitled "Moral Reasoning and Conduct of Selected Elementary School Children" submitted by Ian Michael Wright in partial fulfilment of the requirements for the degree of Doctor of Philosophy.

) med IV Supervisor

Charles Cham Ittimment.

n. M. Jonna

External Examiner

September 29", 1975-Date ...

ABSTRACT

The major purpose of this exploratory study was to investigate the relationship between moral reasoning and conduct. A second purpose was to study the effect of participation in moral discussions on moral reasoning. Kohlberg's cognitive-developmental theory of moral development provided the conceptual framework.

In three designated schools, Grade Five and Six teachers identified, according to 'a researcher designed instrument, students who displayed overt 'delinquent' or 'non-delinquent' behavior. From these the researcher selected eighteen 'delinquents' and twenty 'nondelinquents' as representing the most overt anti-social or sociable, behavior. Subjects ranged in age from ten years and five months to thirteen years and one month. All attended schools in low, or lowmiddle, socio-economic areas.

All subjects were administered Form A of Kohlberg's Moral Judgment Instrument (1973 version), and a Flavell cognitive roletaking task in an interview situation. Canadian Lorge-Thorndike intelligence quotients were obtained also. Subjects were then randomly placed into an experimental treatment group which participated in six, one-half hour moral discussions held over a six week period; a placebo group which played Social Studies games over the same time period; or a control group which continued with normal classroom activities. A, posttest consisting of Form B of the Kohlberg instrument was administered, followed a month later by the readministration of Form A and the Flavell task.

Analysis of variance statistics indicated that 'delinquents'

and 'non-delinquents' differed significantly (p = .001) on moral reasoning scores at all test administrations. Product-moment correlations on modal stage scores among three judges were over .88. Verbal intelligence quotients were significantly related (p = .01) to moral reasoning scores (r = .51) but the product-moment correlation was only .15 when 'delinquents' scores were analysed separately. Role-taking was apparently not related to moral judgment (r = .02). Analysis of variance statistics indicated that the treatments had no significant effect on any groups' moral reasoning acores (p = .13). There was, however, a slight indication that 'non-delinquents' in the experimental treatment group were affected; this group made the most advances between pre and follow-up tests.

On the basis of the findings it was hypothesized that moral reasoning was related to conduct, and that the effect of participation in moral discussions might be a function of conduct. Implications discussed included (1) the need for teachers to reason with 'delinquents' in terms of the latters' stages of development; (2) the need to examine values education, as part of the Social Studies program, in light of Kohlberg's theory; and (3) the need for program developers to be aware of children's stages of moral and role-taking development. Further studies exploring the relationship between moral reasoning and conduct appeared warranted.

ACKNOWLEDGEMENTS

I am greatly indebted to my supervisor, Dr. D. Massey, for his encouragement, guidance and friendship; to Dr. C. Chamberlin and Dr. G. Fitzsimmons for their assistance and support; to Dr. J. Young of my examining committee and to Dr. P. Martofella, my external examiner, for their participation; to Dr. W. Hague for his assistance in the planning stages; to the Edmonton Public School Board for permission to conduct the study; and to the principals, teachers and students who made the study possible.

Appreciation is also expressed to Mr. R. Mahen, Dr. K. Nixon, Dr. F. Simon and Dr. M. Smith for checking reliability, and to the graduate students in Elementary Education and Secondary Social Studies Education for their support and friendship.

Last, but not least, I am greatly indebted to my wife, Annette, for her encouragement, support and patience.

TABLE OF CONTENTS

.

Ċ,

х.

.

Sample Limitations	PTER	P	AGE
THE PROBLEM	I.	INTRODUCTION AND STATEMENT OF THE PROBLEM	1
STATEMENT OF THE PROBLEMS 8 DESIGN OF THE STUDY 10 The Sample 10 Instrumentation 10 Experimental Design 10 DEFINITIONS 13 Functional 13 Operational 13 ASSUMPTIONS 14 LIMITATIONS 15 Theoretical Limitations 15 Sample Limitations 15 SIGNIFICANCE OF THE STUDY 16 SUMMARY 18 II. RATIONALE AND RELEATES STUDIES 20 PINGET 20 PINGET 20 PINGET 20 II. RATIONALE AND RELEATES STUDIES 20 PINGET 20 PINGET 20 PINGET 33 Intelligence 33		THE PROBLEM	2
STATEMENT OF THE FRODUCTS		THEORETICAL OVERVIEW	دي ⁵
The Sample 10 Instrumentation 10 Experimental Design 10 DEFINITIONS 13 Functional 13 Operational 13 ASSUMPTIONS 14 LIMITATIONS 15 Theoretical Limitations 15 Sample Limitations 15 Significance OF 15 SUMMARY 16 SUMMARY 18 HI RATIONALE AND RELATES STUDIES 20 PINGET 20 PINGET 20 Intelligence 33	· .	STATEMENT OF THE PROBLEMS	8
Instrumentation 10 Experimental Design 10 DEFINITIONS 13 Functional 13 Operational 13 Operational 13 ASSUMPTIONS 14 LIMITATIONS 15 Theoretical Limitations 15 External Limitations 15 Sample Limitations 15 Significance of The Study 16 SUMMARY 18 II. RATIONALE AND REINTER STUDIES 20 PINGET 20 KOHLBERG 24 FACTORS INFLUENCING MORAL DEVELOPMENT 33 Intelligence 33		DESIGN OF THE STUDY	10
Instrumentation 10 Experimental Design 13 DEFINITIONS 13 Functional 13 Operational 13 Operational 13 ASSUMPTIONS 14 LIMITATIONS 15 Theoretical Limitations 15 External Limitations 15 Sample Limitations 15 Significance OF THE STUDY 16 SUMMARY 18 H. RATIONALE AND RELATED STUDIES 20 PINGET 20 KOHLBERG 24 FACTORS INFLUENCING MORAL DEVELOPMENT 33 Intelligence 33		The Sample	10
DEFINITIONS		Instrumentation	10
Jacobian Jacobian Functional Jacobian Operational Jacobian ASSUMPTIONS Jacobian ASSUMPTIONS Jacobian ASSUMPTIONS Jacobian ASSUMPTIONS Jacobian ASSUMPTIONS Jacobian ASSUMPTIONS Jacobian Imitations Jacobian Sample Limitations Jacobian Sample Limitations Jacobian 'Internal Limitations Jacobian 'Intelligence Jacobian 'Intelligence Jacobian		Experimental Design	10
Operational 13 ASSUMPTIONS 14 LIMITATIONS 15 Theoretical Limitations 15 External Limitations 15 Sample Limitations 15 Sample Limitations 15 Significance OF4 THE STUDY 16 SUMMARY 18 II. RATIONALE AND RELATED STUDIES 20 PIAGET 20 KOHLBERG 24 FACTORS INFLUENCING MORAL DEVELOPMENT 33 Intelligence 33		DEFINITIONS	13/
ASSUMPTIONS		Functional	1/3
ASSUMPTIONS		Operational \ldots \ldots \ldots \ldots \ldots \ldots \ldots	/13
LIMITATIONS 15 Theoretical Limitations 15 External Limitations 15 Sample Limitations 15 'Internal Limitations 15 SIGNIFICANCE OF THE STUDY 16 SUMMARY 18 U. RATIONALE AND RELATED STUDIES 20 PIAGET 20 KOHLBERG 24 FACTORS INFLUENCING MORAL DEVELOPMENT 33 Intelligence 33	. • ₹	ASSUMPTIONS	14
Theoretical Limitations 15 External Limitations 15 Sample Limitations 15 Internal Limitations 15 SIGNIFICANCE OF THE STUDY 16 SUMMARY 18 I. RATIONALE AND RELATED STUDIES 20 PIAGET 20 KOHLBERG 24 FACTORS INFLUENCING MORAL DEVELOPMENT 33 Intelligence 33		LIMITATIONS	15
External Limitations		Theoretical Limitations	15
Sample Limitations 15 Internal Limitations 16 SIGNIFICANCE OF THE STUDY 16 SUMMARY 18 I. RATIONALE AND RELATED STUDIES 20 PIAGET 20 KOHLBERG 24 FACTORS INFLUENCING MORAL DEVELOPMENT 33 Intelligence 33		External Limitations	15
SIGNIFICANCE OF THE STUDY		Sample Limitations	15
SUMMARY 18 II. RATIONALE AND RELATED STUDIES 20 PIAGET 20 KOHLBERG 20 FACTORS INFLUENCING MORAL DEVELOPMENT 33 Intelligence 33		• Internal Limitations	
II. RATIONALE AND RELATED STUDIES 20 PIAGET 20 KOHLBERG 24 FACTORS INFLUENCING MORAL DEVELOPMENT 33 Intelligence 33		SIGNIFICANCE OF THE STUDY	16
II. RATIONALE AND RELATED STODIES 20 PIAGET 20 KOHLBERG 24 FACTORS INFLUENCING MORAL DEVELOPMENT 33 Intelligence 33		この時間 ぬない 読んしょ 🥊 出入 行き データー オードレーズ しいせいしょう しんしんしょう	
PLAGET 24 KOHLBERG 33 FACTORS INFLUENCING MORAL DEVELOPMENT 33 Intelligence 33	II.	RATIONALE AND RELATED STUDIES	
FACTORS INFLUENCING MORAL DEVELOPMENT		PIAGET	
Intelligence	· · · ·	그는 것은 것을 가장 같은 것을 하는 것을 하는 것 같은 것을 가지 않는 것을 하는 것을 수 있다. 이렇게 나는 것을 하는 것을 하는 것을 하는 것을 하는 것을 수 있다. 이렇게 나는 것을 하는 것을 수 있는 것을 수 있는 것을 하는 것을 하는 것을 수 있는 것을 수 있는 것을 수 있는 것을 수 있다. 이렇게 나는 것을 수 있는 것을 수 있다. 것을 것 같이 않는 것을 수 있는 것을 수 있는 것을 수 있는 것을 수 있는 것을 것 같이 않는 것을 수 있는 것을 수 있는 것을 것 같이 않는 것 같이 않는 것 같이 않는 것 않는 것 않는 것 같이 않는 것 않는 것 않는 것 같이 않는 것 않는	
		에 가장 이 것 되었다. 한 것 같은 것 되었을 것 않는 것 이 것 같은 것 같은 것 같은 것 같이 ??	
			35

vii

CHAPTER

III.

ŋ

Instructional Processes	•••• 39
KNOWLEDGE AND ACTION	•••• 45
SUMMARY	•••• 51
THE RESEARCH DESIGN	•••• 53
THE SAMPLE	•••• 53
INSTRUMENTATION	•••• 61
The Student Behavior Rating Instrument	• • • • 62
Validity	• • • • 64
Reliability	• • • • 64
The Fairmindedness and Trustworthiness Instrument	t 68
Kohlberg's Moral Judgment Instrument	• • • • 68
Administration	70
The Interview	• • • 71
Validity	
Construct Validity	• • • 74
Reliability	••• 77
	••• 81
Procedure	• • • 83
Flavell's Role-Taking Instrument	• • • 85
Administration	••• 86
Validity	••• 87
Reliability	• • 88
The Intelligence Test,	• • • 88
TREATMENTS	••• 90
The Treatment Group	• • /• 90

IV.

The Placebo Group	• • 95
The Control Group \ldots \ldots \ldots \ldots \ldots \ldots	•••• 95
ANALYSIS OF THE DATA	• • • 95
Statistical Procedures	•••97
THE PILOT STUDY	
Instrumentation	
	*
şummary	
FINDINGS OF THE STUDY	· · ·
Introduction	104
FINDINGS OF THE STUDY	••• 105
Question 1	•••105
Discussion	107
Question 2	••• 110
Discussion	•••119
Question 3	••• 119
Discussion	• • 123
Question 4	• • • 124
Discussion	• • • 126
CONCLUSIONS	••• 127
SUMMARY	130
SUMMARY, IMPLICATIONS AND SUGGESTIONS FOR FURTHER RESEARCH	131
Introduction	131
SUMMARY OF THE STUDY	• • • 132
Summary of the Research Procedures	• • 132

PAGE

X

			•	-
• CHAPTER			PAC	ĴE .
	Summary of the Finding	s	13	35
•	IMPLICATIONS			39
۰ ۱ ۱ ۱ ۱ ۱ ۱ ۱ ۱ ۱ ۱ ۱ ۱ ۱ ۱ ۱ ۱ ۱ ۱ ۱	Implications for the C	ognitive-Development. lopment	· · · · · · · · 14	40
i i	Implications for Class		ч х	44
	Implications for Socia		. · · · · · · · · · · · · · · · · · · ·	45
	Implications for Curri		14	
	SUGGESTIONS FOR FURTHER	•	14	49
BIBLIOG	RAPHY	••••	1	51
APPENDI	CES		•	
Α.	DEFINITION OF MORAL STAGES	· · · · · · · · · ·	1	62
в.	THE TEN MORAL ISSUES '		••••••	65
c.	THE STUDENT BEHAVIOR RATIN	NG INSTRUMENT	1	67
D.	THE FAIRMINDEDNESS AND TRU	JSTWORTHINESS INSTRUM	ENT 1	71
E .	THE KOHLBERG MORAL JUDGMEN	NT INSTRUMENT	1	73
F.	THE ROLE-TAKING INSTRUMENT	6 • • • • • • • • • • • • • • • • • • •		79
G۱	THE FILMSTRIP DILEMMAS USE EXPERIMENTAL TREATMENT GRO		~ • • • • • • • • 1	.81
H.	GAMES PLAYED BY THE PLACE		·s 1	84 ·)
· · · · · · · · · · · · · · · · · · ·	SAMPLE PRETEST RESPONSES I RESPONSE TO THE ISSUES OF PUNISHMENT IN THE HEINZ DI	BY STAGE OF LIFE AND	1	.87
	SAMPLE RESPONSES TO THE R		· · · · · · · · · · · · · · · · · · ·	.89
•	NUMBER OF SCORABLE RESPON	SES AT EACH MORAL UBJECT ON THE PRE,		.91
	POST AND FOLLOW-UP KOHLBE	rg moral Judgment Tes		· · · ·
X				

LIST OF TABLES

. .

٠

٠,

•

•

	LIST OF TABLES
- TABLE	PAGE
. 1.	Categorization by Teachers on the SBRI of 'Delinquent' Subjects
2.	Categorization by Teachers on the SBRI of 'Non-Delinquent Subjects
3.	Ratings of 'Delinquent' and 'Non-Delinquent' Subjects by Teachers on the Fairmindedness and Trustworthiness Instrument
4.	Subjects by Age, as of April 1st, 1975, Grade and Sex
5.	Number of Subjects by School and by Category 61
6. 1	Percentage Agreement of Teacher Rating on the SBRI
7.	Judges' Scoring on the Kohlberg Moral Judgment Instrument
8.	Percentage Agreements Between Judges on Overall Moral Reasoning Scores on the Post and
. 6	Follow-up Tests
9.	Percentage Interjudge Agreement on Pretest Role-Taking Scores
10.	Test-Retest Role-Taking Scores for Éach Subject 89,
11.	Composition of the Three Experimental Groups by Student Identification Number and Category,
	and by School
12.	Summary of the Characteristics and Location of the Sample 102
13.	Pretest Moral Maturity Scores for the "Delinquent' and 'Non-Delinquent' Groups 106
14.	Two-Way Analysis of Variance Results on Pre, Post and Follow-Up Test Moral Maturity
	Scores of 'Delinquents' and 'Non-Delinquents' in the Three Treatment Groups

xi

анна 1999 - Марияна 1999 - Марияна

--

. •

.

ß

15.	Pre, Post and Follow-Up Test Moral Maturity Scores for 'Delinquent' and 'Non-Delinquent' Subjects in the Experimental Treatment, Placebo and Control Groups
1,6.	Mean Moral Maturity Scores for 'Delinquents' and 'Non-Delinquents' in Each of the Treatment Groups at Each Test Administration 114
17.	Product-Moment Correlations Between Pre, Post and Follow-up Test Moral Maturity Scores, for the Entire Sample and for the 'Delinquents' and 'Non-Delinquents' by Treatment Group 115
18.	Analysis of Variance Statistics on Pre and Follow- Up Test Moral Maturity Scores for 'Delinquents' and 'Non-Delinquents' in the Experimental, Placebo and Control Groups
19.	<pre>Pre and Follow-Up Test Role-Taking Scores and Moral Maturity Scores for the 'Delinquent' and 'Non-Delinquent' Groups</pre>
20.	Moral Maturity Scores and Verbal and Non-Verbal . Intelligence Quotients for the 'Delinquent's and 'Non-Delinquent' Groups
21	Product-Moment Correlations Between Moral Maturity Scores and Intelligence Quotients

x

TABLE

LIST OF FIGURES /

FI	GURE		1	PAGE
	1.	The Research Design	• .	12
•	2.	Stage Level Relative to Assigned Stage	•	30
	3.	Relations Between Piaget Logical Stages and Kohlberg Moral Stages	•	, 35 35
	4.	Social Perspective for the First Four Stages of Moral Reasoning	•	37
	5 .	A Comparison Between the Moral Judgment Instruments Used in This Research and Those Now Suggested by Kohlberg	•	69
	6.	The Pattern of Moral Success	•	75
÷	7.	The Purpose Administration, Validity, Reliability and Type of Data Gained, of Each of the Research Instruments	-	91
·	8.	The Relationship Between Role-Taking Scores and Moral Maturity Scores	•	122
	9.	The Relationship Between Moral Maturity Scores and Verbal Intelligence Quotients	•	129
•			Ł	•

.

Chapter I

INTRODUCTION AND STATEMENT OF THE PROBLEM

ĩ

As a result of the past decade of ferment in the Social Studies, most teaching-learning programs in this subject area have moved beyond the emphasis on facts and the transmission of the cultural heritage. Recent literature on the teaching of Social Studies in the schools stresses not only the intellectual development but also the personal and social development of students (Joyce, 1972; Fraenkel, 1973). Many writers (Oliver and Shaver, 1966; Newman and Oliver, 1970; Simon, 1970) have emphasized that Social Studies education should focus, as the Alberta Social Studies program (Alberta Department of Education, 1971) does, upon social '' and personal issues. These social issues ". . . are fundamentally moral problems—and moral problems of which we can for the most part see no clear solution . . . (Jeffre¶s, 1968, p. 40)."

As the school has the function of teaching skills, developing intelligence and forming character (Lester Smith, 1964, p. 29), as formal schooling not only teaches people, but pre-eminently teaches people how to live with people (Dewey, 1959, 1966), and as people "need to be trained not only in scientific but human terms (Snow, 1962, p. 47)," monsiderations of morality cannot be ignored.

> For questions of private and public behaviour and of social relationships, and the exercise of moral judgment, are matters of practical necessity for all children as for all adults, in their studies as well as in everyday living (May, 1971, p. 39).

Moral education has been of concern to educators since the time of Plato's <u>The Republic</u>. During the last decade there has been a deepening interest within education in morality, moral development and moral education. This interest is attested to in the over one thousand books and articles contained in a bibliography (Association for Values Education and Research, 1972) compiled at the University of British Columbia.

However, the term 'moral' is subject to many interpretations. In fact, Warnock (1967, p. 73) states that there is no commonly accepted definition of the term 'moral.' In psychology varying paradigms give rise to differing concepts of the nature of man and the ways in which he learns moral ideas. However, the fundamental question of morality and moral education was recognized long ago in Meno's question to Socrates:

> Can you tell me Socrates—can virtue be taught? Or if not, does it come by practice? Or does it come neither by practice nor by teaching, but do people get it by nature, or in some other way? (Rouse, 1963, p. 28)

This question has been 'answered' by Lawrence Kohlberg whose theory was the subject of this study.

1. THE PROBLEM

One of the paradoxes of human behaviour is that one often knows what one should do but acts otherwise. The question then becomes, "Is one acting morally?" Hare (1952, p. 1) says:

> If we were to ask a person, 'What are his moral principles?' the way in which we could be most sure of a true answer would be by studying what he DID.

One might be inclined prima facie to agree with this dictum,

but it must be realized that actions may not tell one what a man's moral principles are. One needs to not only observe action, but also to realize the reasoning behind the action. Kohlberg (1964, p. 408)) argues that ". . . the development of judgment and development of conduct do not go along on two independent tracks." He reiterated this position when he stated (Kohlberg, 1971b, p. 78), "Fortunately, moral maturity in judgment and in action are closely related." This view is subscribed to by Porter (1972, p. 126), who states:

> Generally it has been found that persons at a higher level of moral judgment not only reason better, but they act in accordance with their judgments.

Kohlberg's position is that:

Although more research needs to be conducted, existing evidence clearly supports a positive relationship between stage of reasoning, and moral behavior. The goal of facilitating more mature forms of moral thinking is certainly important to the long range goal of promoting social action (Kohlberg and Selman, 1972, p. 41).

Whereas it may be true that moral knowledge and understanding are prerequisites for authentic moral action and that, for example, one must learn respect for the property of others if one is to know that one should resist the temptation to take it even when safe opportunity offers, other writers question the knowledge-action concomi-

tance claim. Bull (1969, p. 5) posits that:

No claim could be made for strong correlation between moral judgment and moral behavior . . Even if the individual knows the right thing to do, so much depends upon motivation, the actual situation, and—according to some adolescent girls—even mood.

Kay (1970, p. 39 and p. 112) also argues that there is no necessary connection in every situation between moral reasoning and

conduct, and Graham (1972, p. 16), when discussing studies involving guilt and resistance to temptation, says:

Their value may be primarily in demonstrating that judgment and action are not always related—if indeed any such demonstration is necessary.

On the one hand, therefore, we see support from Kohlberg, Porter and Selman that "reasoning and behaviour are linked because maturemoral action requires mature forms of moral thought as prerequisites (Kohlberg and Turiel, 1971, p. 457)," and that "if we know a child's moral-judgment level we should be able to predict a good deal of his moral behavior (Ibid.)." On the other hand we have support for the view that knowledge and action are not necessarily linked in that an individual may know the right thing to do but acts contrary to this knowledge. This knowledge-action question is of prime importance for moral education. If a program in moral education can raise a student's level of moral reasoning but it is unknown as to whether or not there will be any concomitant change in the maturity of moral conduct, then moral education can be questioned. One would wish children to behave as well as reason at morally mature levels.

This study examined, therefore, the moral reasoning and conduct of selected elementary school children in light of Kohlberg's theory of moral development. The study was exploratory in that it attempted

to:

... discover significant variables in the field situation, to discover relations among variables, and to lay the groundwork for later, more systematic and rigorous testing of hypotheses (Kerlinger, 1973, p. 406).

As far as is known this study was unique in that it explored the moral reasoning of students who were deemed by teachers to display

'delinquent' or 'non-delinquent' characteristics in a classroom situation. Although studies have been carried out on the moral reasoning of legal delinquents in corrective institutions, no one has studied the moral reasoning of Grade Five and Six 'delinquents' in a public school setting. Whether the moral reasoning variable is related to 'delinquent' conduct can, at present, only be speculated. This study should, therefore, help to lay the groundwork for hypothesis building as to the relationship between moral reasoning and conduct among 'delinquent' students in the upper elementary school. Although previous research has studied the effects of exposure to higher stage moral reasoning on legal delinquents and 'normal' elementary school students, this study explored the effects of participation in moral discussions on 'delinquent' and 'non-delinquent' students. Results might form a basis for discovering the relationship between conduct and the effects of participation in moral discussions, leading to possible hypotheses which might form the framework for future research.

2. THEORETICAL OVERVIEW

In order to investigate the problem of whether or not there is concomitance between how one reasons about moral problems and how one acts in concrete moral situations, the cognitive-developmental theory of Lawrence Kohlberg was used. Based on empirical evidence Kohlberg has developed a six stage model of moral development (Appendix A). Progression through these stages results from organismenvironmental interaction with the resulting mental structure being a product of this interaction rather than a direct reflection of either

innate neurological or external environmental patterns (Kohlberg and Mayer, 1972, p. 457). Like Piaget, Kohlberg views cognitive structures as internally organized wholes or systems of internal relations. These structures are rules for processing information or connecting events. Kohlberg postulates that:

1. The stages are universal. They are not culturally determined and particular religions make no difference.

2. All stages have to be moved through; there is an invariant sequence, order or succession and no stage may be skipped. However, speed of development will vary with the individual, fixation may occur at a given stage, and individuals may be partly in one stage and partly in neighbouring stages.

3. Each stage reflects different motives and a differentiation of life over other values.

4. Each stage shows a concern for welfare and justice.

There is reasonably consistent evidence that moral development is closely related to intelligence either as measured by intelligence quotients or by Piagetian tasks. However, intelligence is a necessary, not a sufficient condition for moral development. Another factor which appears to influence development is role-taking ability. According to Selman (1971) only reciprocators [the understanding that others can take one's own perspective simultaneously with one's taking of others' perspectives (Selman, Gordon and Damon, 1973)] attain the conventional level of morality.

Kohlberg postulates that upward movement occurs through stimulation of cognitive conflict. Studies by Turiel (1966), Turiel and Rothman (1972), Blatt and Kohlberg (1973), Beck, Sullivan and Taylor (1972), and Lieberman and Selman (1974) all found that movement was created when lower stage subjects were pitted against higher stage subjects or experimenters either in discussion or role-playing contexts. Moral development, it would appear, is facilitated there-

1. Provision of enhanced opportunities for role-taking.

2. Exposure to cognitive conflict, to contradictions in one's own moral views.

Exposure to moral reasoning one stage above one's own. 3. When it comes to putting moral reasons into action, Kohlberg posits that "the influence of judgment on action should be characterizable in cognitive-structural terms (Kohlberg, 1969, p. 390)." Several research studies (Kohlberg, 1971a, p. 228; Haan et al, 1968; Kohlberg and Turiel, 1971, p. 459) have apparently demonstrated a congruence between actions and stage of moral reasoning. One of the most promising avenues of research into the thought-action paradox appears to be with prisoners in corrective institutions and with juvenile delinquents. Kohlberg, LaCross and Ricks (1971) state that repetitive stealing or bullying are possible indicators of immaturity in children as young as seven or eight and that these are predictive of adult maladjustment, delinquency and anti-social behavior. Kohlberg (1958) found that delinquency was a sign of preconventional moral judgment, although non-delinquency is not a sign of conventional

moral reasoning.

Delinquency is the most extreme form of consistent disregard of approved behavior. In this connection it

has been found that the large majority (eighty-three percent) of fifteen to seventeen year old working class delinquents, are preconventional . . . (Kohlberg and Turiel, 1971, p. 460).

As Kohlberg and Turiel (1971, p. 460) state that:

, . . We may expect a typical child to have reached the conventional level of moral judgment (Stage's 3 and 4) by early preadolescence, and to reflect this level in behavior consistently showing a decent regard for the core expectations and approval of parents, peers and outside authorities.

it can be hypothesized that pre-adolescents who repetitively violate these basic expectations have failed to reach the conventional level. Although there are other sociological and psychological factors involved, delinquency does appear to indicate immature moral development. This is evidenced by the Kohlberg, Scarf and Hickey (1972) prison study and studies by Fodor (1972), Hudgins (Hudgins and Prentice, 1973) and Ewanyk (1973) in which adult inmates or adolescent delinquents invariably scored at Stages 1 or 2.

According to Kohlberg and Turiel (1971, p. 448) one of the critical transition periods is between the ages of ten and thirteen. If a child has not begun to reason at the conventional level by age thirteen he is unlikely to develop principled morality in adulthood. Therefore this period appears to be crucial for moral development and it is this period which was examined in this study.

3. STATEMENT OF THE PROBLEMS

The major purposes of this study were to investigate the following questions.

1. Is there congruence between how a person reasons about

moral, problems and how he acts in actual moral situations?

2. Will a situation in which students are exposed to moral reasoning at one stage above their initial modal level influence their level of moral reasoning towards the higher stage?

In order to answer these questions the following specific problems were investigated.

 Do Grade Five and Six 'delinquents' and 'non-delinquents' differ in the scores obtained on Kohlberg's Moral Judgment Instrument?

2. Will a situation in which a random selection of Grade Five and Six 'delinquents' and 'non-delinquents' are exposed to moral reasoning at one stage above their pretest stage influence their reasoning to the extent that their stage of reasoning is higher on the posttest than it was on the pretest?

2 (a). Will any changes be reflected on a follow-up administered one month after the posttest?

2 (b). Will there be any differences between pretest, posttest and follow-up scores on the Kohlberg Moral Judgment Instrument between 'delinquents' and 'non-delinquents' in experimental, placebo and control groups?

Additional questions researched were:

Taking Instrument (Flavell et al, 1968) related to stage or moral reasoning? For 'delinguents //? For 'non-delinguents'?

3 (a). Do 'delinquents' and 'non-delinquents' differ on role-taking ability?

Is intelligence quotient as assessed by the Canadian

Lorge Thorndike Intelligence Tests (Lorge et al, 1967) related to stage of moral development? For 'delinquents'? For 'non-delinquents'?

4. DESIGN OF THE STUDY

In order to investigate the problems posed the following research design was formulated.

The Sample

One hundred and fifty-seven students in Grades Five and Six in three Edmonton public schools, chosen by the Edmonton Public School Board, were selected by teachers and/or principals as displaying 'delinquent' or 'non-delinquent' characteristics. A Student Behavior Rating Instrument (Appendix C) was used for selection purposes. From this sample, forty subjects were chosen by the researcher as demonstrating the most extreme characteristics of 'delinquency' (n = 20) and 'non-delinquency' (n = 20). During the course of the research two 'delinquent' subjects became unavailable for study.

Instrumentation

Role-Taking Instrument. All subjects were administered Flavell's Role-Taking Instrument (Flavell et al, pp. 70-81) (Appendix F).

Kohlberg's Moral Judgment Instrument. (Appendix E) All subjects were administered Form A of this instrument as a pretest, Form B as a posttest and Form A again as a follow-up test one month after the posttest. Each subject was interviewed separately and all interviews were tape-recorded. Intelligence Quotient. All Grade Six subjects were administered Form D of the <u>Canadian Lorge-Thorndike Intelligence Test</u> and all Grade Five subjects were administered Form C of the same test.

Experimental Design

All subjects were randomly assigned to one of three groups.

1. Experimental Group (n = 13). These subjects were involved in six, one-half hour sessions held once per week in each of the three schools, over a six week period. Five of the sessions comprised the viewing of one of the First Things filmstrips (Guidance Associates, 1972) followed by a discussion. The final session comprised the discussion of moral dilemmas which subjects felt were relevant to their own lives.

2. <u>Placebo Group</u>. This group (n = 12) was involved in six, one-half hour sessions held once per week in each of three schools, over a six week period. During each session a Social Studies related game was played.

3. <u>Control Group</u>. This group (n = 13) continued with normal classroom activities during the experimental and placebo group sessions.

The Design

Figure 1 provides a summary of the research design.

Teacher and others' ratings on the Student Behavior Rating Instrument

Selection of the

'Delinquent' sample

'Non-delinquent' sample

12

Kohlberg Moral Judgment Instrument (Form A)

Flawell's Role-Taking Instrument

Canadian Lorge-Thorndike Intelligence Test

Random selection into

Experimental Treatment Placebo

Control

Posttest 🖌

Kohlberg Moral Judgment Instrument (Form B)

Follow-up test

Kohlberg Moral Judgment Instrument (Form A) Flavell's Role-Taking Instrument

FIGURE 1

THE RESEARCH DESIGN

1.

2:

3.

5.

5. DEFINITIONS

13

Functional

Value: Any characteristic deemed important because of psycho-/ logical, social, moral or aesthetic considerations (Good, 1973, p. 636).

Moral: That which a person considers to be right or wrong and that which he considers he ought or ought not to do.

Moral Judgment: This involves a choice between principles, policies or courses of action based on a criterion or criteria for right action.

<u>Conduct</u>: The actual behaviors demonstrated by someone in a real world concrete situation.

<u>Delinquent</u>: Behavior that is in violation of the conduct norms of society; usually based on legal norms but can be broadened to include all^o anti-social acts whether or not these are brought to the attention of a court of law (Good, 1973, p. 170).

Role-Taking Ability: The facility of seeing something from somebody else's perspective; the ability to disassociate one's own viewpoints from those of others.

Operational

Rating Instrument.

Moral and Moral Judgment: The 'score' obtained from the analysis of subjects' responses to Kohlberg's Moral Judgment Instru-

Conduct: The ratings obtained from the Student Behavior

Role-Taking, Ability: The score obtained on Flavell' Fle-Taking Instrument.

Delinquent: Those subjects identified by teachers and others on the Student Behavior Rating Instrument as displaying overt anti-

<u>Non-Delinquent</u>: Those subjects identified by teachers and others on the Student Behavior Rating Instrument as displaying overt social behavior.

Intelligence: The score obtained on the <u>Canadian Lorgé</u>-Thorndike Intelligence Tests.

6. ASSUMPTIONS

1. That Social Studies education not only has intellectual growth functions but is also concerned with children's personal and social behavior and moral development.

2. That the cognitive-developmental theory of moral development provides a sound theoretical base for understanding moral ', reasoning.

3. That the Kohlberg Moral Judgment Instrument will accurately reflect the moral reasoning of subjects.

4. That the 'stage level' of each subject can be accurately ascertained from responses given to the dilemmas posed in the Kohlberg Moral Jüdgment Instrument.

5. That Flavell's Role-Taking Instrument will accurately reflect the cognitive role-taking ability of subjects.

6. That the Student Behavior Rating Instrument will accurately

reflect the classroom conduct of the subjects.

10

7. That subjects in the experimental treatment situation will initially use the stage(s) of reasoning assessed from the pretest score on the Kohlberg Moral Judgment Instrument.

8. That any researcher effect on the experimental group posttest scores will be controlled by the incorporation in the design of the placebo group, and by the administration of a follow-up test.

7. LIMITATIONS

The following limitations appear to apply to this study.

Theoretical Limitations

The study was limited to one theory of moral development. Although the theory appears to be based on empirical evidence and has a reasonable philosophical basis, it is but one theory among several.

External Limitations

As the study was exploratory, generalizations to other populations could not be made.

Sample Limitations

These limitations included the small size of the sample, the assignment of schools by the School Board and the difficulties involved in obtaining the sample.

Internal Limitations

Caution was used in the analysis of results due to the internal limitations of the research design. Even though a number of subjects were categorized as 'delinquent,' treating these subjects as a group was carried out cautiously for, although the group shared a common characteristic (i.e., anti-social behavior), the specific quantitative and qualitative degree to which the individuals in the group were equivalent was questionable. Also no attempt was made in this study to account for the home background or past experiences of the subjects.

8. SIGNIFICANCE OF THE STUDY

The study was deemed significant in the following ways:

I. Knowledge was gained regarding how 'delinquents' reasoned about concerns of morality. Although studies on legal delinquents in corrective institutions have been carried out (Kohlberg, 1958; Ewanyk, 1973), this study appeared to be unique in that it researched the moral reasoning of Grade Five and Six students who were deemed by teachers to display overt anti-social behavior in 'normal' public school classrooms.

2. Information gained from this study may have significance for program development and classroom practice. Although several studies, such as those by Turiel (1966), Blatt and Kohlberg (1973) and Beck, Sullivan and Taylor (1972), have apparently demonstrated that a student's stage of moral reasoning can be raised there is still a question regarding the success of the intervention technique on all students. This study, therefore, examined the efficacy of the Kohlberg technique on 'delinguent' and 'mon-delinguent' Grade Five and Six students.

The Alberta Social Studies program (Alberta, Department 3. of Education, 1971) stresses the values clarification approach designed by Raths et al (1966). Presumably this approach can be used with students at any age. Yet the question can be raised as to whether, for example, students can identify and choose freely from among all known alternatives (Steps 1, 2 and 3 of the Raths model) unless they are cognitively and morally mature enough to do The Kohlberg theory when put into instructional practice does so. attempt to match the student's stage of development to the instruction. If students are only capable of reasoning at the stage they are in, or the one stage above, then this developmental principle can be used as a basis for instruction by "creating dissatisfaction" in the student about his present knowledge of the good (Kohlberg, 1970, p. 308)," by exposing the student to +1 reasoning.

Values clarification strategies do not stress role-taking; role-taking is included as one part of the overall appraoch. Yet according to Kohlberg (1971b, pp. 51-52), moral reasoning is closely related to role-taking as mature moral reasoning involves considering the welfare of others. This necessitates the ability of 'taking another person's point of view,' It would appear that, although roletaking ability is developmental in nature, it needs to be practised. Woral discussions which involve such questions as, "How would you feel if you were A?" involve the practising of seeing others' perspectives either within a dyad, group or societal framework. The importance of role-taking has not been adequately explasized either in the values clarification approach or in the Alberts Social Staties program.

Also values clarification centers on the personal values of the student; the end result is unknown and should not be inculcated. Kohlberg, however, maintains that although values (content) may be \mathcal{O} relative, the reasons for holding values are not. Kohlberg argues that the notion of relativity will have to be rethought as it is both philosophically and psychologically unsound.

Therefore this study had significance in that it examined the moral reasoning of elementary school students, creating a possible basis for moral or values education programs in the school.

4. The conduct aspect of morality is of import. One would wish students not only to reason at a mature level but also to act at a morally mature level for it is the child's conduct which the teacher evaluates. In this regard all teachers are values educators. Yet the relationship between thought and conduct is problematic. As research evidence regarding the congruence between moral reasoning and moral conduct is scarce this study was regarded as both unique and necessary. As Wright (1971, p. 172) says:

We need to know how people's theoretical morality relates to the rest of their moral lives.

9. SUMMARY

This chapter has introduced the major research questions and has outlined the procedures used to provide answers to these questions. The major problem was seen to be the hypothesized relationship between moral researching and moral action, with a secondary concern relating to the efficacy of the Kohlberg intervention technique designed to raise a person's stage of moral ressoning. Chapter II examines research studies related to the problems posed. Chapter III outlines the research design, and the procedures used to analyse the data. Chapter IV contains the results of the data analysis and Chapter V summarizes the study, discusses the implications and provides ~ suggestions for further research.

Chapter II

RATIONALE AND RELATED STUDIES

This chapter reviews the cognitive theory of moral development as formulated by Piaget and modified and extended by Kohlberg. Relevant research and theory in terms of moral and cognitive stages, decentration (role-taking ability), instructional processes and actual moral conduct are analysed in order to explicate the operational problems and design of this study.

1. PIAGET

Jean Piaget postulated that children move through invariant stages in cognitive operations—the sensori-motor, pre-operational, concrete operational and formal operational. He posited (1960,

pp. 13-14) that these stages had a constant order of succession, that the elements of one stage were integrated into the next higher stage and that lower stage elements were the basis for the higher stage. Factors which influenced development were maturation, experience (physical and logico-mathematical), social interaction and transmission, and equilibrium. Piaget's main work has been in the nature of physical and qualitative thinking (Phillips, 1969, p. 129) but in 1932 he more one boot on children's much reasoning, <u>The Moral</u> **Matural State Control**. By stateting the reasoning of lower class,

pattern consisting of three stages.

- 1. Pre-moral Stage
 - No concept of right or wrong.
- 2. Heteronomous Stage

Unlitaral respect for adults leads to heteronomous attitude towards adult rules as being sacred and unchanging because of:

- a. Egocentrism-confusion of one's own perspective
- with his own-inability to see moral values as relative to various persons and ends.
- b. Realism—confusion of subjective phenomena with objective things—moral rules as fixed rather than psychological expectations.

This leads to:

- c. Objective responsibility—physical consequences are used as the basis for judgment rather than intents.
- d. Unchangeability of rules.
- e. Absolution of values—'right' and 'wrong' are absolute and in conflict situations the parent is 'right.'
- f. Wrongness is defined by sanctions, wrong = punishment.
- g. Duty as being obedience to authority.
- h. Belief in explatory punishment.
- i. Ignoring reciprocity in defining obligations.
- j. Belief in immanent justice.
- k. Belief in collective responsibility.
- 1. Punishment by authority rather than retaliative reciprocity by victim.

3. Autonomous Stage

Growth of sense of rights (justice) and duty (obligation) as co-operation increases and egocentrism declines means that the above 'beliefs' will disappear and more subjective and autonomous thinking will replace them.

However, it must be noted that Plaget did not conceptualize these

latter two moralities as being discrete. He states (1932, p. 78):

There exists in the child certain attitudes and beliefs which intellectual development will more and news word to eliminate there are others which will development the second second second and in the second second second second is the child and in the second second second second is the one, the other in and a basic developmental pattern has, in the main, been substantiated. However, moral development has been seen to be more complex than Piaget realized in 1932. Questions have been raised regarding the consistency of development across all components of morality. Johnson (1962), testing for immanent justice, moral realism, punishment and responsibility concepts with eight hundred and seven subjects in Grades Five, Seven, Nine and Eleven, found development with age in all areas. However, correlations between the various moral judgment tests used ranged from .68 to .15 demonstrating, as did Brennan (1962) in part, that various areas of judgment may not be highly related. MacRae (1954) also found development with age but like the aforementioned researchers he could find no common entity in moral judgment and McKechnie (1971) concluded that specificity in reasoning was related to the behavioral area involved and the structure of the

protocols presented.

Other questions have also been raised regarding whether or not every component of moral judgment, as described by Piaget, is developmental in nature. The following general characteristics are deemed to be the criteria for stage developmental theory (Kohlberg and Mayer, 1972, p. 458):

> 1. Stages imply distinct or qualitative differences in children's modes of thinking or of solving the same scohlem.

These different modes of thought form an invariant sequence, order, or succession in individual development.
While contrast factors may speed up, slow down, or stop development, they do not change its sequence.
Base of these different and sequential modes of the secuence.

1. Ship of shares differings and segmential modes of the state a further a first and ships a given stagethe state of a first down with the state state a specific the state of a state of the state of a state an underlying thought organization.

4. Cognitive stages are his rarchical integrations. Stages form an order of increasingly differentiated and integrated structures to fulfill a common function.

Whereas developmental theory has been substantiated according to the above criteria in the areas of objective responsibility, fixity of rules, absolutism of value, wrong defined as punishment, explative justice and immanent justice (Kohlberg, 1963, p. 319), the more socioemotional aspects of Piaget's theory have been questioned. For example, peer group participation, although an important aspect of overall moral development, has not been found to be specifically associated with advance on measures of reciprocity or intentionality. Kugelmass and Breznitz (1967) found that children raised on an Israeli kibbutz were not more intentionally oriented than children raised in a conventional Israeli family. Reciprocity appears to increase in children aged between six and nine but then stays the same or declines (Kohlberg, 1963, p. 319).

A second factor of Piaget's theory relates to the existence of a democratic home atmosphere. Yet MacRae (1954) and Johnson (1962) found that parental democracy did not relate to development on the Piagetian component of objective responsibility. According to Lydiat

(1973, p. 376), the non-developmental dimensions are:

Duty defined as obedience to authority, reciprocity in rights, punishment by authority versus retaliation, individual and collective responsibility, and favoritism in the distribution of reverds and punishments.

Accordingly it can be stated that developmental theory holds in certain dimensions of Finget's theory. However, development is det definitively/age related and variation due to many factors can
occur. Piaget himself stated (1932, p. 78):

A given individual may, for example, have reached the state of autonomy with regard to a certain group of rules while his consciousness of these rules, together with the practice of certain more subtle rules, will still be colored with heteronomy.

2. KOHLBERG

Lawrence Kohlberg has extended and modified Piaget's original thesis by developing an ideal typological model to explain the development of moral reasoning. Kohlberg's theory differs from Piaget's on several major points. The first is the starting point. Whereas Piaget begins with respect for sacred rules, Kohlberg (1973c, p. 191) commences with the pragmatic concern of punishment, finding that ten year olds who displayed Stage 1 thinking (Appendix A) did not show strong respect for adult authority. In fact, young children appear to judge acts by whether or not the act is rewarded or punished. A 'bad' act which was accorded parental reward was deemed to be 'good' by young children (Kohlberg, 1968, p. 489). Secondly, Piaget's theory contained three stages, whereas Kohlberg incorporates six (Appendix Thirdly, whereas Piaget stresses peer-group participation as a A) 👾 major factor facilitating development, Kohlberg (1968, p. 489) does not. Yet both agree that there exists:

> A culturally universal age development of a sense of justice, involving progressive concern for the needs and feelings of others and eleborated conceptions of reciprocity and equality (Kohlberg, 1968, p. 489).

Basically the Kohlberg theory states that:

The conditive-developmental metaphor is not material, it is diffectical; it is a model of the progression of ideas in discourse and conversation. The dialectical metaphor was first elaborated by Plato, given new meaning by Hegel and finally stripped of its metaphysical claims by John Dewey and Jean Piaget, to form a psychological method. In the dialectical metaphor a core of universal ideas are redefined and reorganized as their implications are played out in experience and as they are confronted by their opposites in argument and discourse. These organizations define qualitative levels of thought, levels of increased epistemic adequacy (Kohlberg and Mayer, 1972, p. 456).

Kohlberg sees this progression as a reorganization of psychological structures resulting from organism-environmental interaction, with the mental structure being a product of this interaction rather . than a direct reflection of either innate neurological patterns or external environmental patterns (Kohlberg and Mayer, 1972, p. 457). Like Piaget, Kohlberg views cognitive structures as internally organized wholes or systems of internal relations. These structures are rules for processing information or connecting events, and these events are organized actively through these cognitive connecting processes, not passively through external association and repetition. The core of development is not the unfolding of instincts, emotions or sensori-motor patterns but instead is cognitive change in distinctively human general patterns of thinking about the self and the world. The development of moral reasoning has therefore a cognitive core, and the cognitive and affective are parallel aspects of the structural transformation which takes place in development (Kohlberg and Mayer, 1972, p. 457). Kohlberg admits (1971b, p. 44) that moral judgments involve strong emotional components but he states that emotions will not reduce the cognitive core.

Development is therefore movement towards greater differentiation, integration and adaptation; it is progression from a less

adequate psychological state to a more adequate one (Kohlberg and Mayer, 1972, p. 483). It is a change in form, pattern and organization of behavior, not just a change in frequency or intensity of responses.

Knowledge is thus viewed as functional or pragmatic. It is an equilibrated or resolved relationship between an inquiring human mind and a problematic situation. The child's epistemology is both unique to the individual at his stage of development and universal as all individuals progress through stages of development.

Evidence for the existence of the six stage model (Appendix A) was based initially on analysis carried out on the responses given to moral dilemmas by eighty-four male subjects ranging in age from nine years and six months to sixteen years and four months (Kohlberg, 195

quasi-simplex correlate more statements made hundred and eight ever, whereas age in the research (G ledges:

ng to Kohlberg (1958), these stages form a ent stages to the subjects' dominant stage each other than do distant stages. All ts could be assigned to one of the one (thirty moral aspects by six stages). Howin moral development have been substantiated , 1972, p. 182), this does not validate the notion of sequenting nvariance. As Kohlberg (1973c, p. 189) acknow-

> While the age trends indicate that some modes of thought are generally more difficult or advanced than other modes of thought, they do not demonstrate that attainment of each mode of thought is prerequisite to the attainment of the next higher in a hypothetical sequence.

This evidence is provided in a longitudinal study which followed some of Kohlberg's original 1958 subjects. According to the results of this study, "Each individual child must go step by step through each of the kinds of moral judgment . . . (Kohlberg, 1971b, p. 36)." This longitudinal data, however, has demonstrated that post-conventional reasoning is not reached until adulthood, if at all. ". . . our longitudinal data now suggests that Stage 5 is a stage reached in adulthood, not in adolescence . . . Thinking we labelled Stage 6 in high school was misclassified (Kohlberg, 1973a, p. 31)." In fact since the conception of the six stage model in 1958, further modifications have been made on the basis of research evidence. Kramer (Kohlberg and Kramer, 1969) discovered what at first was assumed to be a regression in the moral reasoning of college students. These students apparently were reasoning at Stage 2, yet on further analysis Kohlberg posited that this reasoning was characteristic of a transition between conventional and post-conventional thinking, and has thus labelled it as Stage 4 - 1/2. This has, as it were, 'saved' the sequential and hierarchial aspect of the theory. However, there is a statement by Kohlberg (1973b, p. 501) which reads:

> A pilot cross-sectional study of moral judgment in the aged (unpublished) suggested that some, but far from all, aging people regressed to childish pre-conventional patterns of moral thought.

This suggests that the non-regression aspect of the cognitivedevelopmental theory may not be valid for all age groups. Other changes have included the reclassification of the various aspects of moral judgment. Whereas in 1958 thirty aspects were conceptualized, these have been reclassified into ten major issues (Appendix B). Also

substages within each stage have been formulated based on evidence from a longitudinal study. The A substage is based on concerns of normative order and utilitarianism and the B substage on justice and ideal-self concepts. According to Kohlberg, Kauffman, <u>et al</u> (1974, p. 25), the longitudinal data indicates that the B substage is more amature in the sense that a 3A thinker moves to 3B, but a 3B never moves to 3A although he may move to 4A. However, the B substage can be skipped. This study was only concerned, however, with the six stage model and not with the substages.

Cross-cultural research (Kohlberg, 1971b, p. 37) indicates that people progress through the stages regardless of culture. However, evidence for this universality of development is still open to question. Graphs of stage usage of subjects in five countries (Kohlberg, 1971b, p. 37) do not indicate how many subjects were tested, which dilemmas were used or what the scoring reliability was, and evidence for the existence of post-conventional reasoning is absent in two out of the five samples.

Further evidence for the developmental theory is provided by Rest[°] (Rest <u>et al</u>, 1969) and by Selman (Kohlberg and Turiel, 1971, p. 451). Rest demonstrated with a sample of forty-five Fifth and Bighth Grade boys and girls who were asked to recapitulate statements at each of the stages, that these subjects could recapitulate statements at one stage below their own modal stage with a high degree of accuracy. Recapitulation of +1 statements yielded a moderate degree of accuracy with +2 statements being the least accurate. Subjects apparently understood their own stage and lower stages but did not

comprehend reasoning which was too developmentally advanced. Rest also found, when subjects were asked to state which of the statements taken from each stage they preferred and which they thought were the worst advice, that:

> The -1 statements were chosen as the worst advice with significantly more frequency than either the +1 or +2 statements (p < .01). The +1 and +2 statements were also judged as being better than the -1 statements (p < .01), and the results indicate also that these two statements were equally preferred (Turiel, 1969, p. 104).

The same pattern was demonstrated with a group of forty-seven Twelfth Graders (Rest, 1973). This group preferred higher stage reasoning, but could only comprehend and recapitulate reasoning at their own stage, lower stages or the +1 stage.

Ability to comprehend moral reasoning was also shown by Selman to be a function of developmental principles. Kohlberg and Turiel (1971, p. 451) report that Selman, in a research study which involved ten to sixteen year old children's interpretations of the 'Golden Rule,' found only Stage 3 children translating the rule correctly. Those at Stage 2 used a Stage 2 orientation of actual exchange or revenge, not a Stage 3 ideal reciprocal judgment involving a consideration of what you would wish if you were in the other's place. 'Yet almost all children could correctly verbalize the formula of 'Do unto others as you would have them do unto you.'

Further evidence of the cognitive-developmental theory of moral development is provided in studies which attempted to induce changes in moral reasoning, and in studies which related behavior to

moral judgment. 'This evidence will be discussed later in this chapter.

Each of the six stages represents a structured whole but this does not mean "at any given time a child functions entirely on one stage and that change involves movement from one such discrete stage to another (Turiel, 1969, p. 113)." In fact the evidence suggests that the majority of subjects tested respond at various stages.

> Typically, as children develop they are partly in their major stage (about 50% of their ideas) partly in the stage into which they are moving and partly in the stage they have just left behind (Kohlberg, 1971b, p. 38).

The actual usage pattern may be as diagrammed below in Figure 2.



(Mohlawrg, 1969, p. 387)

This above pattern is borne out by the correlations existing among the various dilemmas presented to subjects. These range between .75 and .31 with the median being .51 (Kohlberg, 1969, p. 388). One of the problems presented in reviewing studies pertinent to this particular research study is that it is often unknown as to which dilemmas were presented to subjects and which questions were posed. Therefore, the 'degree of fit' between reasons given to one dilemma and those given to another is problematic. One reason for the fluctuating correlations between various dilemmas may also be due to actual differences in the dilemmas. Leming (1974) posits that there are two modes-one a judgment one in which a response is made to an act which has already been performed, the other a deliberation mode in which a correct course of action is chosen for the dilemma charac-For example, in the Heinz story (Instrument I, Appendix E) ter. Heinz has actually broken into the store to steal the drug, but in the dilemma involving Joe (Instrument II, Appendix B) Joe has to decide whether or not to give his money to his father. Secondly, Leading suggests that classical moral dilemmas which are removed from the life space of the individual (such as the aforementioned Heinz story), differ from practical dilement in which there is a femiliar character in a familiar situation. Leaing in fact found differences in moral maturity accores in his sample of sixty, Seventh and Twelfth Grade subjects according to the mode and type of dilumna presented. According to Mobilisers, Scarf and Micky (1972), prison inmates used Lower modes of thought os primes dilements as compared with the

standard newspiriters o

Fluctuations in scores, therefore, may be due to the different types of dilemmas posed. Haan <u>et al</u> (1968), however, found that fortysix per cent of their adolescent sample could not even be classified in a modal stage. This 'inability' to find a modal stage may be due to a number of factors, such as scoring imprecision or to the lack of differentiation between learned verbal responses and 'truly structural' responses. But measurement imprecision may not necessarily account for all the variation.

> Some of the variation can be explained by Piaget's concept of 'decalage' . . . Decalage exists because some concepts are more difficult than others and because a child has had more experience in some realms than in others (Turiel, 1969, p. 115)..

Two concepts can be introduced here. One is that of 'mental structures' which refer to forms of thought that may function in more or less cohesive patterns; the second is 'structuring process' which refers to the mode of transition from less mature to more mature forms of functioning. Variation will typify changes in the structuring process, fixity will be typified by consistency. This consistency could be a function of age—the young child is progressing very slowly in moral development, the adult at the higher stage has reached optimal equilibrium. On the basis of these hypotheses, Turisl (1969) analyzed the data from three studies and computed variation scores. He found (1969, p. 125):

Annual dillighted they also continued of the seciry denses provide an exclusion of the seciry doubted the seciry second of the second second of the second doubted the second second doubted to be a second of the second doubted to be a second of the second second second doubted to be a second doubted to be second as a second second second doubted to be a second doubted to be second as a second second second second doubted to be a second doubted to be second as a second second second second second doubted to be a second second second second second second second doubted to be a second greatest amount of mixture because at that point higher stage functioning is very unstable.

On the basis of both theoretical considerations and empirical vidence Kohlberg states:

1. The stages have universal application. They are not culturally determined although the cultural environment may help to speed up the developmental process or slow it down (Kohlberg, 1971b, p. 38).

2. All stages have to be moved through; there is an invariant sequence and no stage may be skipped. However, speed of development will vary with the individual, fixation may occur at a given stage, and individuals may be partly in one stage and partly in neighboring stages.

3. Each stage reflects, (a) different motives for action, (b) differentiation of life over other values, and (c) a concern for welfare and justice.

4. Each stage is a cognitive structure, an internally organized whole or system of internal relations.

3. FACTORS INFLUENCING HORM. DEVELOPMENT

there is according consistent evidence that and i devidence. As closely addresses (2007), Solid-to-ty body (2007), Annow Sales, Address (2007), Solid-to-types (2004), Witness, and Makes Sales, Weight, Weight, Solid-to-types, (2004), Matama, and Makes Sales, Makes (2006), Solid-to-types, (2004), Matama, and Makes Sales, Makes (2006), Solid-to-types, (2004), Matama, and Makes Makes (2006), Makes (2006), Solid-to-types, (2004), Matama, and p. 391) reports that he found "a curvilinear relation between I.Q. and moral maturity. In the below-average range, a linear correlation (r = .53) is found between I.Q. and moral maturity." However, in the above average group a low correlation of .16 was found between

intelligence quotient and moral maturity.

In the above mentioned studies, the intelligence tests used varied as did the correlations with moral maturity. It must be realized that it may not be true to say that those with high intelligence quotients will necessarily be morally mature for, "Cognitive structure development depends upon massive general experience, a requirement which an 'innately' bright child cannot short circuit (Kohlberg, 1968a, p. 1030)." Far more relevant, therefore, are studies which utilize Piagetian cognitive tasks and correlated results of these with stages of moral reasoning. Hypothetically the relationship should be as shown in Figure 3.

Jantz (1973) found a positive relationship between intellectual development, as measured by conservation ability, and moral development. Sixty-eight per cent of his sample fell in the pattern of non-conservation/morality of constraint or conservars/morality of co-dpension. Jes (1971) found that the transition to found operations air a prod linking of an interse in Stage 4 thicking, and that the transition for site of a spectrum is there i thicking, and that the transition of a billing operations was a good indicator of a billing in State 5 founding. Side DBNI abtenues a consistion of detailed in State 5 founding. Side DBNI abtenues a consistion of

Moral Stage

Symbolic, intuitive thought.

Logical Stage

Concrete operations. Substage 1. Formulation of stable, categorical classes.

Reversible concrete thought.

Formal operations. Substage 1. Relations involving the inverse of the reciprocal.

Formal operations. Substage 2. Depacity to other triads of propositions or relations.

10.3.

Stage 0. The good is what I want and like.

Punishment-obedience Stage 1. orientation.

Concrete operations. Substage 2. Stage 2. Instrumental hedonism and concrete reciprocity.

> Stage 3. Orientation to interpersonal relations of mutuality.

Maintenance of social Stade 4. order, fixed rules and authority.

Social contact, Stage 5. utilitarian law making tage 6. Universal ethical primitie printerion.

Q.

and Tomlinson-Keasey (1972) also showed that formal operations thinking was necessary for principled morality. Although not clear cut, the relationship between Piagetian cognitive tasks and stage of moral reasoning appears to be quite well substantiated, in that "a given logical stage is a necessary but not sufficient condition for the parallel moral stage (Kohlberg, 1973, p. 13)."

Role-Taking

Q.

Another way of assessing intellectual ability and moral development is to focus on decentration and role-play skills, both vital according to Piaget and Kohlberg, for development. In fact, one of the assumptions of the cognitive developmental theory is:

> There is a fundamental unity of personality organization and development termed the ego or the self. While there are various strands of social development (psychosexual development, moral development, etc.) these strands are united by their common reference to a single concept of self in a single social world. Social development is, in essence, the restructuring of the (1) concept of self, (2) in its relationship to concepts of other people, (3) conceived as being in a common social world with social standards (Nohlberg, 1969, p. 349).

Therefore, the ability to role-take is necessary for mature. Morel juigeants as moral juigeants at the higher stages involve considering the perspective of other people. In this regard Mohlberg has not Mediated athers of social perspective taking in the most second section memoirs (Mohlberg, 1975). The desilopsent of social second for Mohlberg to Shops & is wellings in Figure 4.

and a second second

Stage 1. Doesn't consider the interests of others or recognize they differ from actor's. Doesn't relate two points of view. Actions are considered physically rather than in terms of psychological interests of others. Confusion of authorities perspective with one's own.

Stage 2. Separates own interests and point of view from those of authorities and others. Aware everybody has their own interests to pursue and these conflict, so that right is relative (in the concrete individualistic sense). Integrates or relates conflicting interests to one another through instrumental exchange of services, through instrumental need for the other and the other's good will, or through fairness in treating each individuml's interest as equal.

Stage 3. Aware of shared feelings, agreements and expectations which take primacy over individual interests. Relates points of view through the 'concrete Golden Rule,' putting yourself in the other person's shoes. Does not consider generalized 'system' perspective.

Stage 4. Differentiates societal point of view from interpersonal approximation or motives. Takes the point of view of the system which defines roles and rules. Considers individual relations in taxes of place in the system.

label it as a summary variable because it contains within it various properties. Globally, role-taking is equated with the general ability of taking the role of another person and assessing his responses, capacities and tendencies. The ability includes literalperceptual and figurative-intellectual components. It is this latter cognitive ability that Irwin and Ambron (1973), with their sample of seventy-two Kindergarten and Grade Two subjects, found was most highly related to moral judgment as assessed by Piagetian stories concerning blameworthiness, restitution, and intentionality perspectives. Yet, overall, there was a significant relationship between all the role-taking tasks (cognitive, affective and perceptual) and moral judgment. The same relationship was demonstrated by Selman (1971) in that on two role-taking tasks an overall significant relationship existed with moral maturity scores obtained from Kohlberg's Moral Judgment Instrument. In this sample of sixty subjects at the ages of eight, nine and ten, only recipricators attained the conventional level of morality. Stuart (1967) also demonstrated that decentering ability was related to moral maturity, with high decentraters rendering more mature moral judgments than low decentraters.

It would appear that role-taking ability is a significant operator of moral reasoning in that it is a necessary, but not sufficient condition for moral meturity. Kohlberg (1971, p. 50) therefore argues that opportunities for role-play activities are necessary for moral development. ... opportunities for role-taking are important and (all) morate in a similar direction by stimulating

Instructional Processes

Kohlberg postulates that movement from one stage of reasoning to the next occurs through the stimulation of cognitive conflict. Research by Turiel (1966) demonstrated that movement was created if reasoning was presented to subjects which was at one stage higher than their modal stage. This finding was based on a sample of fortyfour Grade Seven subjects placed in three treatment and one control group. Treatment subjects were exposed to reasoning at -1, +1 or +2 to their modal stage as determined by a pretest which consisted of six Kohlberg dilemmas. Following three treatment sessions spread over a three week period, a posttest was administered. This incorporated the six pretest dilemmas and the three dilemmas used during the treatment sessions. Analysis of the posttest scores on the three treatment dilemmas revealed a significant difference between the +1 and +2 groups which suggested that the +1 treatment was the most effective. However, there was no statistically significant difference between the -1 and +1 groups, although there was a slight indication (p = .10)that the +1 group was more effective. Yet, as these results were based on the analysis of the responses made to the dilemmas used during the actual treatment sessions, they could be due to "memory, learning and suggestion (Rurtines and Grief, 1974, p. 465)." Analysis of the pre/posttest sources on the other six dilemmas revealed no statistically significant treatment effects. In fact, the control group's scores decreased by 6.11 and the +1 group's by 4.51. This research therefore provides only minimal support for the theory that moral development can come through exposing students to il resconing.

In two related studies (Blatt and Kohlberg, 1973), Blatt apparently demonstrat exposure to +1 reasoning did stimulate development. In the a sample of eleven Jewish students aged between eleven ars old were exposed to reasoning at one stage the t level in twelve, one hour long, experimental s в. ng this time subjects discussed dilemmas other than the up in the pretest and the six used in the posttest (four of ere originally used in the pretest). Blatt found a mean incre moral maturity scores between pre and posttests for his experiment al group. This change was statistically significant (p = .0None of the control groups, which were taken from Turiel's 1966 and 72 experiments, increased in moral maturity scores between the press a posttests. In a follow-up test administered one year later, the pastest results were essentially unchanged. Pretest level of moral ity predicted moderately with posttest level (r = .65), while t level predicted well to the follow-up test (r = .84).

In a second study Blatt studied one hundred and thirty-two students of lower class or lower middle class status who were in Grades Eix or Tex. Three experimental groups were established in each ege prompte control group, an experimenter-led discussion group and a discussion droup without active leadership by an adult. Eighteen sendions of forty-five minutes duration each were held over a nine mark parish. The pre-and posttests were those used in Blatt's first ends. Minute indicated that although there were no statistiodily eignificant differences between the time groups on pretest scores, on the posttest the experimenter-led group increased significantly on moral maturity scores. On the follow-up test administered one year later, all groups had increased in moral maturity scores, but the experimenter-led group still differed significantly from the control group.

Tracy (Tracy and Cross, 1973) with a sample of seventy-two subjects aged between twelve and fifteen in experimental treatment and control groups, found that with three experimental treatment sessions the Stage 1 and 2 subjects moved upwards more than the Stage 3 and 4 subjects. In fact, the control group subjects at Stages 1 and 2 moved more than the experimental treatment Stage 3 and 4 subjects. This may demonstrate that the pretest is itself a manipulative device and that, for this age group, a ceiling effect exists for conventional reasoning subjects.

The above studies incorporated relatively short experimental treations sessions. Other research studies have contained experimental treatment sessions of a longer duration. Lieberman and Selman (1974) carried out an evaluation of the <u>First Things</u> (Guidance Associates, 1972) values education program. Both experimental Grade Two groups, one teacher led, the other 'expert' led, moved significantly more over an October to May period than the control group. The suthors point out that as the teacher-led group produced the greatest gains, "it is impossible to sort out 'teacher effects' from 'intermention techniques, '" and they suggest that further research be neutried out in this area.

In a similar four month long study Beck at al (1972), with a

sample of seventeen Grade Eleven students in an ethics course in which a post-conventional rationale for morality was introduced, and seventeen students in a control group, found that there were no significant pre-posttest results. However, when the follow-up test was administered one year later the experimental group scored significantly higher than the control group. Whereas on the preand posttest there were four Stage 5 subjects (two in the experimental and two in the control group), by the follow-up test two-thirds of the experimental group were scored at Stage 5. Similar results were obtained in another study by Sullivan and Beck (1974) involving fortytwo Grade Five students-twenty in an experimental, moral discussion group and twenty-two in a control group. At the end of the first semester a posttest was administered. Although neither group differed significantly from each other on moral reasoning scores, both groups scored significantly higher than on the pretest. However, on a follow-up test administered at the end of Grade Six, the experimental group had progressed significantly more than the control group. Overall, subjects in this group dropped Stage 1 reasoning and began to reason in Stage 4 terms.

0

Despite the significant differences reported above, mearchers st the Association for Values Education and Research (Association for Values Education and Research, no date) report that, in a five well mini-course introduced into Grade Eight, Nine and Ten classes, there was no statistically significant effect due to the Kohlberg treatment. However, the direction of the difference between pre and posttest scores was upward as predicted.

Overall, therefore, there is support for the view that moral development can be stimulated through cognitive conflict and the exposure to +1 reasoning. However, the treatment effect will depend upon a number of factors. Turiel (1969) posits that the age of the subject is of import—a twelve year old at Stage 4 is less likely to progress than a twelve year old at Stage 2. A second factor is the strength of fixation. The third is the manner in which higher stage reasoning is presented; Turiel (1969, p. 129) reasons that only by presenting one stage higher reasons for and against the subject's present stage reasoning and position on a dilemma will the cognitive conflict necessary for movement be created.

There are still many unknowns with regard to this treatment and its purported effects. The question can be posed as to why some subjects who are exposed to experimental treatments increase in moral maturity scores while others do not. According to Blatt (Blatt and Kohlberg, 1973), this may be a function of interest. According to Krebs (Krebs and Kohlberg, 1973), it may be a function of ego control.

Other psychological frameworks suggest that moral learning can occur through the imitation of adult models and through the use of reward and punishment stimuli. Bandura and McDonald (1963) demonstrated that subjects who were objectively oriented could, through imitating an adult model, learn to make subjective judgments, and Dorr and Fey (1974) and McCann and Prentice (1973) found that modelling effect and monetary reward could induce changes in moral responses. Since the Bandurs and McDonald (1963) study, argument has fraged as to whether the effects of modelling behavior cast doubt on

the cognitive-developmental theory. Cowan <u>et al</u> (1969) replicated the Bandura experiment and posttested two weeks after the treatment. Although confirming Bandura and McDonald's results, they stated that pretest scores, time, type of item and direction of conditioning were factors overlooked by the original experimenters. The experiment revealed that downward learning (subjective subjects exposed to objective arguments) was less stable over time than was upward learning.

What appears from this research is that it is possible to induce at least short term changes in moral reasoning through the imitation of adult models. The questions remain however as to (a) how long the induced change will endure; (b) whether or not the acceptance of change is a function of developmental stage; and (c) whether or not a response induced through an experimental treatment is a 'learned' or 'truly structural' one.

Within the Kohlberg theory it would appear, therefore, that moral development is facilitated by various processes. These include: 1. Provision of enhanced opportunities for role taking.

2. Exposure to cognitive conflict, to contradictions in one's own moral views and in their relations to the view of others.

3. Exposure to moral reasoning one stage above one's own. It would also appear that other processes can also induce changes in moral reasoning.

All the above research has been carried out mainly in isolation from the rest of the school curriculum. Fenton (Fenton <u>et al</u>, 1974, p. 6) states:

Discussions of moral dilemmas should be viewed as part of the Social Studies course, not as an additional element to be introduced on 'moral development day' twice each month.

Beck (1971) has made the same point. He shows how moral education can become an integral part of the total school curriculum. Despite possible side benefits of exposing students to moral discussions, such as the growth of inquiry and discussion skills and a greater knowledge of value related issues, research on the effect of the total school experience on the moral development of students is virtually hon-existent.

. KNOWLEDGE AND ACTION

When it comes to putting moral thought into action, Kohlberg postulates that "the influence of judgment on action should be characterizable in cognitive-structural terms (Kohlberg, 1969, p. 390)." Piaget (1932, p. 109 and p. 113) wondered what the relationship between thought and action would be and posited the theory of 'conscious realization' (p. 174). In the theoretical the individual, by reflection, makes consciously explicit the moral principles upon which he actually functions. As verbal thinking consists of a conscious realization of schemes that have been already established in action, there will be a time lag before a developmental practical change is registered at the theoretical level.

This hypothetical time lag implies several things. Firstly, it implies that practical morality may shape theoretical morality but that the theoretical does not shape the practical. The theoretical suralizing of the adult will only affect the child's theoretical

morality and that its influence will only be to retard or help this theoretical morality catch up to the practical. However, this 'conscious realization' hypothesis is speculative.

Other writers suggest that there may be little association between what one thinks one ought to do and what one actually does.

Williams (1970, p. 20) States:

It is not by any means true that, if a person believes the right thing, he will also behave in the right way. A man may believe quite sincerely that it is wrong to punch someone on the nose and yet he may physically attack people every, time they offend him. He may be said to be lacking in some quality, such as self control.

and Peters (1966, p. 203) comments:

The formulation and justification of principles is one thing, but it is still another to apply moral principles intelligently to concrete situations.

These statements are supported in studies by Beller (1949) and Morris (1958). Both found that differences existed between what a subject stated should be done and what he stated actually would be done in certain morel situations.

The classic study of moral conduct was produced by Hartshorne and May (1928). They defined morality in perms of character traits, those which society viewed as moral (like honesty). They found no pattern of development. In an important discussion in Book Two of <u>Studies is Depait</u> (Martshorne and May, 1930, p. 238), on the specificity of moral conduct, attitudes and motives, Martshorne and May reach the postiumion that makes of response are as much a function of the situation as of the individual or are part of the "total

functioning complex that includes both the individual and the

occasion." However, there is evidence from Burton (1963) that there was a general trait of honesty operating relatively independently of situational factors. This view is also supported by Brogden (1940) and by Grim, Kohlberg and White (1968). Of these studies Wright

(1971, pp. 53-54) states:

It is meaningful to talk about a general disposition of moral restraint, but in any temptation situation this is one of a number of dispositional and situational factors at work, and its role in accounting for individual differences may be relatively minor.

In 1964 Kohlberg (1964, p. 408) argued that:

Judgment does not appear to become 'moral' until early adolescence, while 'morality' of conduct appears to develop early. Individual differences in levels of moral judgment are quite general and stable; morality of conduct is more specific to the situation and more unstable over time. Moral judgment appears to develop in the same direction regardless of social groups; moral conduct appears to develop in line with specific class and peergroup norms.

He appears here to support the notion that conduct and reason develop along different lines. Lately he has come out far more strongly for the existence of a positive relationship between thought and conduct, yet evidence for this is scanty. In a study by Brown (Kohlberg, 1971a, p. 238) both principles and conventional college students were theoretically opposed to cheating, but in an experimental cheating situation only those at the principled level actually failed to cheat.

Nohibers argues that only at the principled level is there a contrac-

cash childrelies not to chest; companyional subjects on the other hand see that 'authority' shee not care sheather one chests or not, or that as president size is chesting one does also. Index (Srabe and

hundred and twenty-three Grade Six subjects. Whereas four out of five Stage 5 subjects resisted cheating (80%), as did 45% of the O Stage 4 subjects, only 25% of those in Stages 1, 2 and 3 did so. This resistence to cheating is attested by Krebs to be not only a function of stage of moral maturity but also to ego controls (intelligence quotient and attention). Krebs (Krebs and Kohlberg, 1973, P. 30) suggests that three distinct forces influence moral action in general and resistance to cheating in particular. These are:

- 1. External Social Situational Forces such as social punishment and reward, group pressures and expectations and the influence of models.
- 2. Internal Moral Judgmental Forces or Dispositions the primary variable being moral stage.
- 3. Internal Dispositions of Ego Strength or Ego Control such as intelligence, attention, delay of gratification and achievement motivation.

Kreb's research is, however, questionable in light of Kohlberg's (1973, p. 31) statement that "Stage 5 is a stage reached in adulthood, not in adolescence," and Krebs purportedly found five principled subjects in Grade Six classes. Schwartz <u>et al</u> (1969) demonstrated that college students who were above the sample median on moral maturity scores cheated significantly less than those who scored below the median.

Cheating is but one aspect of morality. It appears to be used primarily as an assessment of moral conduct because it is relatively easy to measure. Cheating is not a good indicator of moral character until the child has developed, in addissounce or later, a set of inner moral principles that modulies it.

An and a second se

action and moral values (a discrepancy due to a variety of possible deficits in ego strength or ego abilities) (Kohlberg, 1973, pp. 464-465).

Other studies on the congruence between thought and action have been sporadic. Kohlberg (Kohlberg and Turiel, 1971, p. 459) gave the moral development tests to subjects in Millgram's (1963) ", study and found that 75% of Stage 6 thinkers (four out of six) refused to administer electrical shocks to a volunteer, whereas only 13% of the twenty-four lower stage subjects did. Testing soldiers who were involved in the My Lai Massacre, Kohlberg (Kohlberg and Selman, 1972, p. 39) discovered that the soldiers who obsyed the order to fire were at Stage 4 or lower, and that only the soldier who disobeyed orders was at a higher stage.

A more sophisticated study was carried out by Haan (Haan et al, 1968). This group of researchers found that Stage 6 college subjects sat in at a Berkeley demonstration, but so too did Stage 2 instrumental relativists; only those reasoning at Stage 3 and 4 were unlikely to demonstrate.

The most providing avenue of research into the congruence between morel thought and conduct appears to be with prisoners in corrective institutions and with delinquents. Although resistance to obseting is not apparently a function of pre-conventional or operactional woral reasoning. Rohiberg, LeCross and Ricks (1971) state that repetitive stating or bullying are possible indicators of immensity in children as young as seven or eight and that these are predictive of while stating justment, delinquency and anti-social

es (1998) found that delinguency was a sign of

pre-conventional moral judgment although non-delinquency is not a sign of having reached the conventional level.

> Delinquency is the most extreme form of consistent disregard of approved behavior. In this connection it has been found that the large majority (eighty-three per cent) of fifteen to seventeen year old working class delinquents are preconventional . . . (Kohlberg and Turiel, 1971, p. 460).

Although there are many psychological and sociological factors involved, delinquency does appear to indicate immature moral development. This is evidenced by the Kohlberg, Scarf and Hicky (1972) prison study and research by Ewanyk (1973), Fodor (1972), and Hudgins and Prentice (1973), all of whom found that adult prison inmates or juvenile delinquents were invariably scored, according to the Kohlberg protocols, at the pre-conventional level.

Research on the concomitance between moral reasons and moral conduct is fragmentary. • Some evidence for conduct differences between principled and non-principled 'thinkers' exists and there is an indication that conventional and pre-conventional subjects may differ in their conduct! However, this area of research still contains many interesting questions such as whether 'milestones' are reached first in action or in moral judgment. It also remains an open question for it could be argued, as Craig (1974), does that some of our conduct may be habitual rather than based on conscious reasoning.

> Motal behavior that involves conformity to social rules is on the whole, to be explained as the result of the sum situational forces, ego variables and socialisection Sectors that determine behaviors which have no discontinue to the sectors.

> > second that Konsberg's social does not require a direct

stage by stage relationship between readon and conduct. A person may perform the same act for a variety of reasons, or persons at one particular stage of reasoning can perform different acts for the same reason. A Stage 2 child may behave well in a classroom because he wants the teacher to like him, or he's achievement oriented, or he's been trained to behave well. Another Stage 2 child might be a behavior problem because he likes doing as he pleases, he craves attention or he doesn't care for achievement goals. In fact, in 1964, Kohlberg (1964, p. 422) stated: "We are not yet able to offer a view of personal moral ideology which combines personality type and developmental considerations into a single framework." Yet in terms of conduct, Kohlberg and Turiel (1971, p. 460) postulated that:

> We may expect a typical child to have reached the conventional level of moral judgment by early preadolescence, and to reflect this level in behavior by consistently showing a decent regard for the core expectations and approval of parents, peers and outside authorities.

It is this last statement coupled with the research evidence on the level of reasoning displayed by juvenile delinquents, and the data on instructional processes within the cognitive-developmental framework that provided the theoretical and empirical basis for this study.

5. SUNMARY

This chapter has reviewed the moral development theory of Plaget and Kohlberg. It was noted that not all of the Plagetian components of morality appear to be truly development. It was

petriced out that Rolling and individe and extended the Piscetian

cognitive-developmental model was discussed in theoretical terms and the theory related to empirical research data. Factors influencing moral development were examined and the evidence for the existence of a relationship between moral reasoning and conduct was reviewed. 52

Chapter III

THE RESEARCH DESIGN

In order to investigate the problems posed in this study, the following elements of the research design were employed.

1. THE SAMPLE

Initially, three elementary schools were assigned by the Edmonton Public School Board on the grounds that these schools were likely to contain a high proportion of 'delinquent' subjects. It was hoped" that a large sample of 'delinquent' subjects would be found within one school so that inter-rater reliability on the ratings of 'delinquent' and 'non-delinquent' behavior might be obtained. However, this proved to be infeasible and the study was subject to the confines of the assigned schools. One school proved to have very few suitable subjects so another school was assigned giving a total population of approximately four hundred Grade Five and Six students from which the initial sample was drawn. Two of the schools were 'inner city' schools drawing their student population mainly from working and lower-middle class parents; the third was in a new suburb and drew its student population mainly from lower-middle and middle class parents. Evidence for socio-economic ratings was obtained from principals, teachers and from the Edmonton School Board and was based upon the type of residences found in the area serviced by the school and by the occupations of the parents whose children attended

the school. These schools were not representative therefore of the Edmonton Public Elementary School system.

In each of the three schools Grade Six teachers were requested to identify 'delinquent' and 'non-delinquent' students according to the categories identified on the Student Behavior Rating Instrument (SBRI) (Appendix C). As a sample size of forty was desired for possible statistical purposes and as initial results of the teacher ratings did not provide such a sample, Grade Five teachers in the two assigned inner-city schools were also requested to rate their students' behavior. In all, eighteen teachers provided student ratings. This provided the sample size desired.

A total of one hundred and fifty-seven students were identified as displaying one or more of the characteristics stated on the SERI. From these, twenty-three subjects were selected by the fessearcher as displaying the most extreme form of 'delinquency' in that teachers had rated them in five or more of the categories of the SERI and had rated them low on the Fairmindedness and Trustworthiness Instrument (Appendix D). Parental permission was obtained from twenty-one of these subjects; however, one subject became unavailable for study prior to commencement of the research and two subjects became unavailable for study during the course of the research and "were eliminated from the research. This left a total of eighteen "delinquent' subjects. These subjects can be identified by the prefix a efter the subject identification number throughout the rest of this

thesis.

which categories teachers rated each

		uaer	IT BE		r Ra	iting		Struit	ent (Lateg	ories	-
Subject	1		: 3	4	5	6	7	8	• 9	10		Total
1A	+				• • •	*	*	•	*	*		6
2A	*		· 🔺	*		*		· .	*			5
3A		*		*	•	*		ં 🗶 ં	• * .	, '	12	. 5
4A	*	*	*	*	*	*	*	*	*	. *	•	10
5A		· *	*	*		*	*	*	* .	* .		8
6A	*	*	· .	• • • •	*		*		, *			6
7A				*		*	*	*		*		5
8 A	*	• • •	· *	*		₹,	· ·	•	•	*		6
9 A	*	· * *	19 🔺 1	`★	*	*	`°. ★ :		*	*.		9
10 A	*		*	*	*	•	. ★ 1		* ',	° ★		8
11A	*		, *	•	• 🛣	1 🔹 1	*		*	* 2	•	8
12A		*	*	. 🛊	. 🔺	*	*	*	*	*		10
13A	. *					с, †	*	1999 - A. 1999 1999 - A. 1999 1999 - A. 1999		*		5
14A	*	. *		., ≭ .,	•	* .	• •	*	*	*		7
15 A		*	*	*	•	*	·. ·	· ·	*	*		7
16 A	*	19 1	. ¹ 👬			., ≜ °.	· *.					5
17 A	•	•			: *	*	· 🖈 💡	* ,		1 🙀 - 1	•	5
188	; / 1	* : : : : : : : : : : : : : : : : : : :	*	s - ≜ . N		*				n de Anne		5
otal	14	10	11	14	8	18	12	7	13	13	D	120

CATEGORIZATION BY TEACHERS ON THE SBRI OF 'DELINQUENT' SUBJECTS

TABLE

1. Interferes with others physically.

2. Unco-operative in work or play.

3. Interferes with others' possessions.

4. Works in discordance with others in group situations.

.

· . ·

5. Vandalizes.

15

6. Frequency of punishment.

7. Punishment has little or no effect.

8. Lacks confidence.

9. Disobeys or disregards classroom rules.

10. Lacks respect for authority.

subject. It is noteworthy with reference to this table that all 'delinquent' subjects were deemed to require frequent punishment (Category 6 of the SBRI) by teachers and the majority (n = 14) of them interfered physically with others (Category 1 of the SBRI) and worked in discordance with others in group situations (Category 4 of the SBRI). The majority (n = 13) of them also were deemed by teachers to disregard and disobey classroom rules (Category 9 of the SBRI) and to lack respect for authority (Category 10 of the SBRI). The degree to which subjects exhibited these various behaviors is questionable as seven teachers and one principal provided the ratings and each of them may have brought to bear his or her own perspective as to the degree to which each rated subject exhibited each behavior. However, it is of note that all raters defined 'delinquency' in terms of punishment, anti-social enavior and lack of respect for rules and authority.

Twenty-nine subjects were selected by the researcher as displaying the most extreme form of 'non-delinquency' in that teachers had rated them in five or more of the SBRI categories and had rated them high on the Fairmindedness and Trustworthiness components of the SBRI. Of these, parental permission was obtained for twenty-two subjects; however, two subjects became unavailable for research purposes prior to commencement of the study, leaving a total of twenty in the 'non-delinquent' group. This group is identified by the prefixing of the letter B after each subject's identification number throughout this thesis.

Table 2 indicates in which categories of the SBRI teachers

TABLE 2

CATEGORIZATION BY	TEACHERS	ON	THE	SBRI**
OF 'NON-DELI	INQUENT'	SUB	JECTS	5

	Stu	ident	Beh	avio	r Ra	ting	Ins	trum	nent	Cate	gories'	***
Subject	1	2	3	4	5	6	7	8	9	10		Total
18	*		*	*		*		*	*			6
2B	*	*		*	*	*		*	*.	*		8
3B.	*	*		*	· *,	*		*	*	*		8
4B	*		*		*	*		*	* .			6
5B	*	*	*	*	*	*		*	*			8
6B		` *	*	*	*	*		*	*	*		.9
7B	. *	*	*	*	*	*		*	*	* 1		9
8B	*	*		. *	· * .	*	-	*	*	•		7
9B			*	*		*		*	*	÷. *		6
10B	*.	•	÷ .		*	* -			*	*	۰,۰°۵	5
11B	*		*	*	*	* .		*	- 🔺	. *	· · ·	8
12B	*		່ 🖈	· 🔺	*	. *			° 🔺	*		7
13B	*	*	*	*	*	*		*	*	*		9
14B	*	*	*	*	. *	. *		. *	*	*		9
15B ີ	*		*	*	*	*			* 1	· · 🔶	1.1	
16B	*	• ★	* .	*	*	*		. 🖈	* .	*	· ·	9
17B	*	. *.,	*.	*	*	* -		÷.	*.	*		9
18B	*		. *			•		*	. ★ .	*		5
19B	*		#	*	*			*	*	- * -		7
20B			*			•	۰.	*.	*	*		5
Total	19	10	16	16	16	17	· ·.	17	20	16		147

**Category 7 did not pertain for the rating of 'non-delinquent' subjects.

*******Student Behavior Rating Categories

- 1. Doesn't interfere with others physically.
- 2. Helps other students.
- 3. Doesn't interfere with others' possessions.

4. Co-operative in group situations.

5. Doesn't vandalize.

- 6. Doesn't have to be punished.
- 8. Has respect in own ability.
- 9. Respects and obeys classroom rules
 - 10. Respects authorities.

had rated the 'non-delinquent' sample. As in the categorization of 'delinquents' (Table 1), teachers perceived 'non-delinquency' in terms of the opposites of 'delinquent' behavior; i.e., lack of punishment necessary, sociable behavior, and respect for rules and authority. However, unlike the majority of 'delinquents,' 'non-delinquents' were perceived as having confidence in their own abilities (Category 8).

The Fairmindedness and Trustworthiness scales (Appendix D)

were scored in the following way.

Fairmindedness	
Very fair	Scored as 1
Generally fair	Scored as 2
Unconcerned	Scored as 3
Selfish	Scored as 4
No sense of	
fairmindedness	Scored as 5

Trustworthiness	•
Completely trustworthy	Scored as
Reliable	Scored as
Conforming	Scored as
Unreliable	Scored as
Untrustworthy	Scored as

Table 3 indicates that the vast majority of 'non-delinquent' ratings were in terms of Very Fair and Completely Trustworthy, whereas 'delinquent' subjects were rated from Unconcerned to No Sense of Fairmindedness, and from Unreliable to Untrustworthy.

1

3

Within the total sample there were fifteen females, thirteen of them being judged as 'non-delinquent,' and twenty-three males, only seven of whom were judged 'non-delinquent' (Table 4). Table 4 also indicates that subjects ranged in age from ten years and five months to thirteen years and one month with the average age of the 'delin-

quent' group being eleven years and five months and the average age

of the 'non-delinquent' group being eleven years and four months.

Subject(1)*(2)*Subject(1)*(2)*1A 3.5 4.5 1B112A 5 4 2B113A 3 3 3 1 1 $4A$ 4 5 4 2 2 $5A$ 4 5 5 1 1 $6A$ 4 4 6 1 1 $7A$ 3 5 7 1 1 $8A$ 4 4 8 2 2 $9A$ 3 4 9 2 1 $10A$ 4 5 $10B$ 2 1 $11A$ 4 4 $11B$ 1 1 $12A$ 3 4 $12B$ 1 1 $13A$ 4 5 $13B$ 2 3 $14A$ 3.5 5 $14B$ 1.5 1 $15A$ 3 4.5 $15B$ 1 1 $16A$ 4 4 $16B$ 1 1 $18A$ 4 4 $18B$ 1 1 $18A$ 4 4 $18B$ 1 1	't	elinquents	•	'Non-Delinquents'				
2A 5 4 2B 1 1 3A 3 3 3B 1 1 4A 4 5 4B 2 2 5A 4 5 5B 1 1 6A 4 4 6B 1 1 7A 3 5 7B 1 1 8A 4 4 8B 2 2 9A 3 4 9B 2 1 10A 4 5 10B 2 1 10A 4 5 13B 2 3 11A 4 4 11B 1 1 12A 3 4 12B 1 1 13A 4 5 13B 2 3 14A 3.5 5 14B 1.5 1 15A 3 4.5 15B 1 1 16A 4 4 16B 1 1 17A 2	Subject	(1)*	(2)*	Subject	(1)*	(2) *		
2A 5 4 2B 1 1 3A 3 3 3B 1 1 4A 4 5 4B 2 2 5A 4 5 5B 1 1 6A 4 5 5B 1 1 6A 4 4 6B 1 1 7A 3 5 7B 1 1 8A 4 4 8B 2 2 9A 3 4 9B 2 1 10A 4 5 10B 2 1 10A 4 5 10B 2 1 10A 4 5 10B 2 1 11A 4 4 11B 1 1 12A 3 4 12B 1 1 13A 4 5 13B 2 3 14A 3.5 5 14B 1.5 1 15A 3	1A	3.5	4.5	18	1	1		
3A 3 3 3B 1 1 4A 4 5 4B 2 2 5A 4 5 5B 1 1 6A 4 4 6B 1 1 7A 3 5 7B 1 1 8A 4 4 8B 2 2 9A 3 4 9B 2 1 10A 4 5 10B 2 1 1AA 4 4 11B 1 1 12A 3 4 12B 1 1 13A 4 5 13B 2 3 14A 3.5 5 14B 1.5 1 15A 3 4.5 15B 1 1 16A 4 4 16B 1 1 17A 2 4 17B 1 1 18A 4 4 16B 1 1 19B 1 <td></td> <td></td> <td>4</td> <td></td> <td>· 1</td> <td>1</td>			4		· 1	1		
4A 4 5 4B 2 2 5A 4 5 5B 1 1 6A 4 4 6B 1 1 7A 3 5 7B 1 1 8A 4 4 8B 2 2 9A 3 4 9B 2 1 10A 4 5 10B 2 1 1A 4 4 11B 1 1 1A 4 5 13B 2 3 1AA 3.5 5 14B 1.5 1 13A 4 5 13B 2 3 14A 3.5 5 14B 1.5 1 15A 3 4.5 15B 1 1 16A 4 4 16B 1 1 17A 2 4 17B 1 1 18A 4 4 16B 1 1			3		ī	1		
5A 4 5 5B 1 1 6A 4 4 6B 1 1 7A 3 5 7B 1 1 8A 4 4 8B 2 2 9A 3 4 9B 2 1 10A 4 5 10B 2 1 11A 4 4 11B 1 1 12A 3 4 12B 1 1 13A 4 5 13B 2 3 14A 3.5 5 14B 1.5 1 15A 3 4.5 15B 1 1 16A 4 4 16B 1 1 17A 2 4 17B 1 1 18A 4 4 18B 1 1 19B 1 1 1 1 1		4	5		2	2		
6A 4 4 6B 1 1 7A 3 5 7B 1 1 8A 4 4 8B 2 2 9A 3 4 9B 2 1 10A 4 5 10B 2 1 11A 4 4 11B 1 1 12A 3 4 12B 1 1 12A 3 4 12B 1 1 13A 4 5 13B 2 3 14A 3.5 5 14B 1.5 1 15A 3 4.5 15B 1 1 16A 4 4 16B 1 1 17A 2 4 17B 1 1 18A 4 4 18B 1 1 19B 1 1 1 1 1	5A	4	5		1	1 '		
8A 4 4 8B 2 2 9A 3 4 9B 2 1 10A 4 5 10B 2 1 11A 4 4 11B 1 1 12A 3 4 12B 1 1 13A 4 5 13B 2 3 14A 3.5 5 14B 1.5 1 15A 3 4.5 15B 1 1 16A 4 4 16B 1 1 17A 2 4 17B 1 1 18A 4 4 18B 1 1 19B 1 1 1 1	6A	4	4	6B	1	1		
9A 3 4 9B 2 1 10A 4 5 10B 2 1 11A 4 4 11B 1 1 12A 3 4 12B 1 1 13A 4 5 13B 2 3 14A 3.5 5 14B 1.5 1 15A 3 4.5 15B 1 1 16A 4 4 16B 1 1 17A 2 4 17B 1 1 18A 4 4 18B 1 1 19B 1 1 1 1 1	7A	. 3	5	7B	1	1		
10A 4 5 10B 2 1 11A 4 4 11B 1 1 12A 3 4 12B 1 1 13A 4 5 13B 2 3 14A 3.5 5 14B 1.5 1 15A 3 4.5 15B 1 1 16A 4 4 16B 1 1 17A 2 4 17B 1 1 18A 4 4 18B 1 1 19B 1 1 1 1 1	8A	4	4	8B	2	2		
11A 4 4 11B 1 1 12A 3 4 12B 1 1 13A 4 5 13B 2 3 14A 3.5 5 14B 1.5 1 15A 3 4.5 15B 1 1 16A 4 4 16B 1 1 17A 2 4 17B 1 1 18A 4 4 18B 1 1 19B 1 1 1 1 1	9A	3	4	9B	2	1		
12A 3 4 12B 1 1 13A 4 5 13B 2 3 14A 3.5 5 14B 1.5 1 15A 3 4.5 15B 1 1 16A 4 4 16B 1 1 17A 2 4 17B 1 1 18A 4 4 18B 1 1 19B 1 1 1 1 1	10A	4	5	10B	2	1		
13A 4 5 13B 2 3 14A 3.5 5 14B 1.5 1 15A 3 4.5 15B 1 1 16A 4 4 16B 1 1 17A 2 4 17B 1 1 18A 4 4 18B 1 1 19B 1 1 1 1	11A	4	4	11B	1	1		
13A 4 5 13B 2 3 14A 3.5 5 14B 1.5 1 15A 3 4.5 15B 1 1 16A 4 4 16B 1 1 17A 2 4 17B 1 1 18A 4 4 18B 1 1 19B 1 1 1 1	12A	3	4	12B	•1	1		
14A 3.5 5 14B 1.5 1 15A 3 4.5 15B 1 1 16A 4 4 16B 1 1 17A 2 4 17B 1 1 18A 4 4 18B 1 1	138	4	5	13B	2	3		
16A 4 4 16B 1 1 17A 2 4 17B 1 1 18A 4 4 18B 1 1 19B 1 1 1	14A	3.5	5	14B ``	1.5	1		
17A 2 4 17B 1 1 18A 4 4 18B 1 1 19B 1 1 1	15A	3	4.5	15B	1	1		
. 18A 4 4 18B 1 1 19B 1 1	16A	4	4	16B	1 ~	i '		
. 18A 4 4 18B 1 1 19B 1 1	17A	2	4	17B	1 . 1	1		
	. 18A	4	4		1	1		
208 1	a an		•	19B -	1	1		
			7	20B	1	1		
		•			•			
*(1) Fairmindedness		stworthines						

RATINGS OF 'DELINQUENT' AND 'NON-DELINQUENT' SUBJECTS BY TEACHERS ON THE FAIRMINDEDNESS AND TRUSTWORTHINESS INSTRUMENT ,

•

TABLE 3 .
	'Delinquent	5'		'N	on-Delinqu	ents'	
Subject	Age Yrs.Mos.	Grade	Sex	Subject	Age Yrs.Mos.	Grade	Sex
	<u> </u>	<u> </u>	M = 1 =	10	12.0	6	Female
1A	12.0	6	Male	1B	12.0	6	Female
2A	11.4	6	Male	2B			Male
3 A	13.1	6	Female	3B	12.3	6	Female
4A	10.5	5.	Male	4 B	11.10	6.4	
5 A	11.4	5	Male	5B	11.0	• • • •	Female
6 A	10.9	5.	Male	6B	11.3	5	Female
7 A	11.0	5	Male	7B	11.3	5	Female
8A	10.11	5	Male	8B	11.0	5	Male
9A	11.9	6	Male	9 B	10.8	5	Female
10A	11.9	6	Male	10B	10.9		Male
118	13.1	6	Male	11B	11.4	6	Female
12A	10.5	5	Male	12B	12.3	6,	Female
13A	12.1	6	Female	13B	10.10	5.	Male
144	12.1	6	Male	14B	10.9	. O 5	Male
15A	11.10	6	Male	15B	12.2	6	Male
164	11.10	6	Male	16B	11.9	6	Female
17A	11.9	6	Male	17B	11.6	6	Male
188	11.7	6	Male	18B	12.1	6	Female
				19B	12.3	- 6	Female
•				20B	12.0	6	Female
X	11.55	•			11.48		

SUBJECTS BY AGE, AS OF APRIL 1ST, 1975, GRADE AND SEX

.

TABLE 4

------;;;

j

Fourteen subjects were in Grade Five classes and twenty-four were in Grade Six. Table 5 below shows how the total sample was distributed by school and by category.

TABLE 5

NUMBER OF SUBJECTS BY SCHOOL AND BY CATEGORY

School	'Delinquen	t' '	'Non-Deli	nquent'	Total
3 1	6		. 8		14
2	6		7	•	13
3	6		5	lle an Anna an Anna Anna Anna Anna Anna An Anna Anna	11
Total	18		20		38

2. INSTRUMENTATION

In order to research the problems posed, the following instruments were used:

1. The Student Behavior Rating Instrument (SBRI) (Appendix C). This was used by teachers and principals to select 'delinquent' and 'non-delinquent' subjects. Categories of behavior were supplied by the researcher and raters selected the name or names of students who fitted a particular category.

2. The Feirmindedness and Trustworthiness Instrument (Appendix D). Teachers and principals rated, on a five point scale, /the degree of fairmindedness and trustworthiness displayed by each student selected initially on the SBRI.

1. The Kohlberg Morel Judgment Instrument (Appendix 2). This

instrument consists of eight moral dilemmas and appropriate questions. Subjects were interviewed separately and responses taped. Responses were subsequently transcribed and scored according to Kohlberg's scoring manual, so that a modal stage of moral reasoning could be ascertained for each subject.

4. Flavell's Role-Taking Instrument (Appendix F). This instrument is designed to ascertain whether or not the subject can cognitively role-take. Subjects first tell a story based on seven pictures showing a boy being chased up a tree by a dog. The pictures indicating that the dog was the motive for the boy's tree climbing are then removed and the subject asked to retell the four picture story from the point of view of another observer. Responses were scored according to the degree to which the subject could shift his frame of reference from the initial seven picture story to the four picture story.

5. <u>The Canadian Lorge-Thorndike Intelligence Tests</u>. The purpose of this instrument was to gain verbal and non-verbal intelligence quotients. As it was a group test and as scores were already available for Grade Six subjects, it was deemed the most feasible instrument to use.

Each of these instruments is described in detail below. This is followed by a discussion on the reliability and validity of each instrument.

The Student Behavior Rating Instrument

this instrument (Appendix C) was used to select the 'delinquent' and 'stardallappent' samples. "Weathers and principals were

requested to identify students in their class of classes who could be categorized according to the items presented. Names of students who displayed 'delinquent' behavior were entered in the left hand column, and names of 'non-delinquents' in the right hand column. The criterion for entry was that the subject consistently displayed one or more of the behaviors listed and stood out from the rest of the student population in the display of the behavior. Teachers then ranked each student whose name had been entered, on the Fairmindedness and Trustworthiness Instrument (Appendix D).

The items on the SBRI were developed from the theoretical base that delinquent acts, at the age group being studied, were a sign of pre-conventional moral reasoning (Kohlberg, 1958; Kohlberg and Turiel, 1971, p. 460). According to Gough (Gough and Peterson, 1952), there are four clusters of pre-dispositional factors in delinquency. These are: (1) role taking deficiencies; (2) resentment towards family; (3) feelings of despondency and alienation, lack of confidence in self and others; (4) poor scholastic achievement and rebelliousness. For the purposes of this study the factors most capable of being assessed were (1); (3) and (4). Coleman (1964) also suggests that the great majority of delinquents are anti-social. He lists the following characteristics; inability to understand and accept ethical values; egocentricity; impulsiveness and irresponsibility; low frustration tolerance and poor judgment, hedonism combined with unrealistic goals, lack of ensisty and guilt; insbility to profit from mistakes; and inshilling an densing maningful social relations. Items about

which a terminer or principal would have information were included in

the SBRI. For example, teachers might not have had reliable evidence regarding a student's home life, but they were likely to know about a student's specific classroom behaviors.

Validity

The SBRI was based on both theoretical considerations and empirical evidence. It was therefore assumed to have content validity in that the items included were representative of the universe of content of classroom 'delinquency' or 'non-delinquency.' Teachers who were asked to adjudicate initial forms of the SBRI did not include any other behaviors as being apparent of 'delinquent' or 'nondelinquent' behavior although they were asked to do so. The same held true for the teachers in the major study. As Kerlinger (1973, p. 459) says, "content validation, then, is basically judgmental," and the SBRI was judged by teachers to be valid for the purposes of defining 'delinquent' and 'non-delinquent' behavior.

The criterion-related validity of the SBRI was investigated in that the major study attempted to discover whether or not moral judgment scores were predicted by selections made by teachers on the SBRI. Results of this validity check are reported in Chapter IV.

Reliability

The reliability of behavior rating instruments is always a problem. Rater bias, ambiguity of meanings, generosity error, the probable lack of equal distance between intervals on a scale and the balo effect (Meandike and Began, 1965, pp. 426-432) all have to be supen into account and controlled if possible. In order to attempt to overcome some of the shortcomings of rating instruments, behaviors were defined as specifically as possible and a substantial number of descriptions were provided. Secondly, the raters had a great deal of opportunity to observe the person being rated in situations in which he would be likely to demonstrate the characteristics on which the ratings were desired. Also the raters, as teachers, were expected to be impartial, although this point and the one above did not guarantee either. Thirdly, the instrument was designed so that it could be completed in a relatively short time. The fater did not have to rate each individual student, instead the rater provided the name of a subject, or names if there was more than one subject, who fitted a given item.

Reliability of rating was obtained in some cases in two schools where a given teacher interacted with more than one class, or the principal taught a class or classes. However, in one school reliability was not obtained as each teacher was responsible for one class alone and the principal was not willing to rate students on the grounds that he did not know the students well enough. Therefore, in the majority of instances, teacher perceptions were taken 'on face value,' as it was impossible to obtain inter-rater reliability. Table 6 indicates the reliabilities obtained on those subjects where ratings by more than one teacher and/or principal were possible. For example, in the case of subject 1%, both raters judged that he fit categories 1, 4; 3 and 10 and both agreed that he did not fit tategories 5 and 8. There was disagreement over the scores on the fairsinglements and trustmorthinges splits, giving an overall percentage PERCENTAGE AGREEMENT ON TEACHER RATING OF THE SBRI

TABLE

の理論に認定的に					ن
		Student Behavior Rating Instrument Categories			
	1	L 2 3 4 5 6 7 8 9 10	Fairminded- ness	Trustworthi- ness	Percentage Agreement
			¥ .	U T	
	2		4	- 4	58%
	F		5	P	
	N		5	4	50 8
	TO		3	с м	
	Ĉ		5	ũ	58 %
1			4	4	
			4	4.5	75\$
5			2	4	664
			4	4	
4			-	1	638
	C.		1	2	
	2				72%
	n		-		
	2				63%
				7	\$

1

TABLE 6. (Continued)

-

4

Percentage Agreement 36% 63% 54% 548 368 548 Trustworthi-1 ness 2 3 ч 1 -----Т Fairmindedness 2 **H** 1 ų e, Ļ ---. Student Behavior Rating Instrument Categories 10 * * * * × ٠ . ົ * * * 4 * # * # Ø 4 * * 4 4 1 ۰, Q ŝ * * 4 4 4 m 4 ¢ * . N ۲ * <u>, en</u>l . * • * ۲ -Reters in, 0 9 M 0 09 2 ÷ r P ť • f. 198 8

67

🖗 ; jt i

agreement of 58%. In total, reliability ratings could only be obtained for five 'delinquents' (27.7%) and for nine 'non-delinquents' (45%).

The Fairmindedness and Trustworthiness Instrument

The two items included on this test were originally used by Kohlberg (1958). The items pertain to two aspects of morality as described by Kohlberg and they were used in this study because the correlation coefficient between ratings on these items and moral judgment scores was relatively high. Kohlberg (1958) reports a product moment correlation of .45 for moral judgment and fairmindedness, and a correlation of .47 for moral judgment and trustworthiness.

Ratings on these two items were obtained for each of the students who were initially selected by teachers on the SBRI.

Kohlberg's Moral Judgment Instrument

The purpose of this instrument is to determine, from responses given to moral dilemmas, the stage(s) of reasoning displayed by a person. This instrument (Appendix E) consists of two forms (Form A and Form B) which are considered by Kohlberg to be equivalent forms for test-retest use. However, as few test-retest coefficients have been reported (Kurtines and Grief, 1974, p. 457), it was deemed advisable to use Form A of the instrument as both a pre and followup (postpost) test instrument. Form B was used as a posttest only: The dilemmas involved (see Appendix E) included the major issues (Appendix B) considered by Kohlberg to be aspects of moral judgment.

Form A and Form B of this instrument involve four major

Instruments Used in This Research

ą

Instruments Now Suggested by Kohlberg

Form A

0

	Dilemma	Issues	Dilemma	Issues
	Heinz	Life. Punishment.	Heinz	Life. Punishment.
	Joe and his father	Father-son relationship. Contract- Promise.	Joe and his father	Father-son relationship. Contract- Promise.
•	Mercy killing	Life. Punishment.	Karl and Bob	Property—Trust Conscience.

Judy and Louise Mother-daughter. Contract/Sistersister.

Punishment.

Form B

i 📢

or sick man Louise Contract/Siste sister.	Captain in Korea	Life. Governance.	Mercy killing	Life. Punishment.
Karl and Bob Property-		Life.		Mother-daughter. Contract/Sister- sister.
trust. Valjean Citizen role. Conscience. Punishment.	Karl and Bob		Valjean	Citizen role.

FIGURE 5

A COMPARISON BETWEEN THE MORAL JUDGHENT INSTRUMENTS USED IN THIS RESEARCH AND THUSE NOW SUGGESTED BY KOHLBERG dilemmas which basically involve a conflict between authoritative decree or moral law and concern for the needs and welfare of individuals. Each dilemma focuses on certain issues such as punishment, governance and the value of life (Appendix B). Kohlberg (Tracy and Cross, 1973, p. 239) has indicated that the "level of moral judgment can be reliably scored from a series of four dilemmas."

The design of this study was based on materials available to the researcher at the time. Since then Kohlberg and his associates have revised the instrument and now use three dilemmas for the pretest and three for the posttest. A comparison (Figure 5) reveals that there was probably little difference in terms of issues examined between the instruments used by this researcher and those now suggested by Kohlberg.

Administration

The instruments were administered on three separate occasions. Form A (pretest) was given first, Form B (postest) approximately six weeks later and Form A (follow-up test) again one month after the posttest. Each subject was interviewed separately and all sessions were tape recorded. Each subject was given a copy of the dilemma and the dilemma was read to the subject by the researcher. Then the questions pertaining to each dilemma (Appendix E) were asked. The researcher only deviated from the set questions if, in his estimation, the subject had mis-comprehended the dilemma, or if a subject's response needed clarification. Each subject was presented the dilemmas in the same sequence in order to facilitate ease of

transcription Each interview session lasted for approximately , thirty minutes, so that on completion of the pretest, posttest and follow-up test each subject had been interviewed for approximately ninety minutes in total.

The Interview

r

The function of the interview was to elicit, whenever possible, stage scorable responses to the questions posed to each subject. Therefore, the interviewer had to be cognizant of the structure of each stage and the probable answers given to each question according to the scoring manual, in order that the highest stage thinking of which the subject was capable, might be reflected in the answers. The function of the interview was not to obtain 'what' responses, but 'why' responses. For example, in reply to the question, "Why should a promise be kept?" Subject 13A on the pretest replied,

> Um . . . well don't make 'em if you can't keep 'em. Well . . , like if you promise somebody something you should keep the promise.

This does not answer the question as to why a promise should be kept. When asked "Why?" the subject said:

Well . . If somebody give me something and I promised somebody I would give them something back . . then I'd keep my promise. Well, because they might give me something that I liked and kind of giving them something . . . and giving me what they give me.

Promises should be kept, it appears, because of instrumental exchange reasoning-you give me something and I'll give you something back. The interviewer must therefore probe to find out, wherever possible, the structure of the reasoning used. However, one can attempt to probe too deeply and questioning ceased if a subject was unable to answer or continually repeated a prior response. 72

The interview was given in an atmosphere which was a relaxed as possible. Subjects were informed that this was not a test in which right or wrong answers were given. The Kohlberg protocols were introduced with the statement, "I'm interested in finding out what students of your age think about some problems. I'm interested in the reasons you give for your answers and there are no right or wrong answers. I'd like you to read this story and then I'm going to ask you some questions." At the end of the interview each subject was thanked for his co-operation and asked what he or she felt about the interview. Most subjects replied that the questions were "not too difficult" and some responded that they had been very easy. The majority of subjects felt that the interview had been quite enjoyable and all were willing to participate. The interviewer made sure that the interview time was suitable to the teacher and to the subject, so that the subject did not miss an activity which he or she really enjoyed.

Validity

As no universally accepted definition of morality exists and as the definition of morality changes according to which state of moral development one is discussing, validity becomes a philosophical problem. Two major concerns are of import in this section. Pirstly, does the Kohlberg instrument assess what can be regarded as <u>moral</u> reasoning? (i.e., content validity). Secondly, are the stages of moral reasoning as posited by Kohlberg valid? In this latter regard the validation of Stages 1 to 4 only will be studied as it is unlikely that subjects at Stages 5 or 6 will be found. Also a defense of the statement that Stage 6 morality is the only true morality (Kohlberg, 1971b, p. 68) was not pertinent to, and was beyond the scope of, this study.

The Kohlberg instrument is designed to elicit responses concerning a subject's conceptions of right and wrong with regard to dilemmas which pose conflicts between authoritative decree or moral law and the welfare of individuals. These dilemmas contain issues (Appendix B) which according to many philosophers [Frankena, 1963; Wilson, 1969; Plato (Rouse, 1963); and Hare, 1952] lie within the moral domain. Kohlberg argues (1971b, pp. 57-62), as do Kant and Hare, that moral judgments are based on principles-universal modes of choosing; general guides of choice which we want all people to adopt in all situations. These principles fall into categories as outlined by principled-intuitionist philosophers. For example, Kohlberg (1971b) includes, as moral components, Sidgewick's prudence (welfare consequences to the self), benevolence (welfare consequences to others), and justice (distributive equity and commutative reciprocity). The one principle Kohlberg adds is respect for authority. All these principles have two characteristics (Kohlberg, 1971b, p. 59).

> First, they are ultimate terms, they refer to states of affairs which seem right or good in themselves and are in that sense 'principles.' Second, they refer to states of affairs that are involved in all moral situations and are potentially relevant to all people.

Kohlberg had philosophical support (Rawls, 1971; Peters, 1966) on

this latter point that moral rules "involve considerations of people's happiness or welfare and consequences of equal treatment between people (Kohlberg, 1971b, p. 59)." Kohlberg states (1958, pp. 5-12) that moral judgments:

- (a) Are priented to or preceded by value judgments.
- (b) Take priority over value judgments.
- (c) Are associated with judgments of the self as being good or bad.
- (d) Based on reasons which are not limited to the consequences of a particular act in a particular situation.
- (e) Have generality, universality, consistency and inclusiveness.
- (f) Are objective to the makers—are agreed to independently of differences of personality and interest.

Therefore on the basis (1) that some schools of philosophy have identified the issues that the Kohlberg instrument assesses, as being moral issues; (2) that the principles (i.e., justice and benevolence) as formulated by some philosophers, are found in some form in each stage of reasoning; (3) that morality concerns people's welfare and principles of justice, the Kohlberg instrument can be said to have valid-

ity in that it assesses moral reasoning.

Construct Validity

There is substantial support for the existence of stages of moral development. Many researchers have identified and labelled such stages (Figure 6) and as Graham (1972, p. 182) states: "the agreement is impressive." Support for the postulates that (a) there are six stages of development and (b) these stages form an invariant sequence; are presented by Kohlberg and his associates.

1. The stages form a quasi-simplex in that adjacent stages to the subjects' dominant stage correlate more highly with each other

			4 1		75
BULL	→ Heteronomy	 Socionomy 	Autonomy		
X	 → Prudential → Authoritarian → → Reciprocity 	↓ Conscience ↓ Social ←	 Self-Concept Personal Religion 		
SINTITI	◆ Obedience	Irrational + - + Group-Oriented + + Empathy	Ego-Ideal		
KORRIS	 Self-Interest Authority Reciprocity 	Conformity to Friends	Indence		THE PATTERN OF MORAL SANCTIONS (Kay, 1970).
		Å			
		\mathbf{t}	Ļ,	n an	

than do distant stages (Kohlberg, 1958).

2. Cross-cultural research (Kohlberg, 1971b) indicates that people progress through the six stages regardless of culture. However Stage 5 and 6 thinkers were not found in some cultures.

3. Longitudinal studies (Rest et al, 1969) confirm that by late adolescence or early adulthood subjects tested had reached Stage 4, and there is some evidence (Kohlberg, 1971b, p. 36) that the stages are invariate.

4. Ranking of scores (percentage of mature responses) in a group of ten year old lower class isolates and a group of sixteen year old middle class integrates was not; according to Kendall's coefficient of concordance, due to chance (Kohlberg, 1958).

5. Stages differentiate between delinquents and non-delinquents, integrates and isolates, and between social classes; factors which previous research had deemed important in moral develop ment.

6. Subjects can only be induced to move to the +1 stage, not the stage which is +2 to their model level (Toriel, 1966)

7. Subjects are capable of translating a moral principle in terms of their modal or +1 stage, but not the +2 stage (Rest, 1968).

 In some situations the stage of reasoning positively correlates with the external oritorion of moral conduct (Haan <u>et al</u>, 1968; Erebs and Kohlberg, 1973; Hudgins and Prentice, 1973; Fodor,

- 1972).

Thepith the frequentary syldence for construct velicity and evidence presented by Durnings and Grief (1976) to expects the statement that "the value of the model remains to be demonstrated, there appears to be some support for the validity of the Kohlberg model and Moral Judgment Instrument.

Reliability

Despite the relative frequency of use of the Kohlberg instrument few references are made to which dilemmas were actually used and there is a dearth of information on test-retest reliability. Lieberman and Selman (1974, p. 5) report only a moderate correlation of .62 between their control group's pre and posttest scores. However, this correlation was based on a test of intentionality and not on the Kohlberg Instrument. Only two test-retest correlations on Kohlberg's test could be found. Blatt (Blatt and Kohlberg, 1973) reports a correlation of .84 (n = 13) and Hickey is reported by Kohlberg (1975) to have obtained 94% agreement between pre and posttest measures for control group scores (n = 20). Also internal reliability is a problem as correlations between the various dilemmas range from .31 to .75 (Kohlberg, 1969, p. 388). However, moral judgments are complex and the Kohlberg instrument does attempt to tap subjects' responses to a variety of moral issues. As Hague (1974, p. 18) states; "And until something better comes along we will continue to use the Kohlberg Moral Judgment Scale . . . The instrument, although not a highly standardized or precise measure, is a useful tool the petting to know how a person views questions that most of us would mail moral (Portar and Taylor, 1972) p. 6)."

a peliability the souring the Schiberg instrument

tarent of the Aven

relatively high. Lieberman and Selman (1974), report correlations of .84, .90 and .81 for pretest, posttest and post-posttest interjudge reliability respectively; Krebs (Krebs and Kohlberg, 1973) reports a figure of .75; Turiel (1966) reports .78 agreement between two judges,

one using the global scoring method, the other the issues scoring method; Blatt's study (Blatt and Kohlberg, 1973) produced a correlation of .89, and Hickey is reported by Kohlberg (1975) to have obtained 70% agreement between two judges on overall scores and 94% agreement on dominant stage scores.

Interjudge reliability in this study was ascertained from two judges. Each judge scored six subjects' responses to each of the pretest (Form A), posttest (Form B) and follow-up test (Form A). One subject's complete protocol was randomly selected from each of the following groups at each testing time.

'Delinquent' in the treatment group.
 'Non-delinquent' in the treatment group.
 'Delinquent' in the placebo group.
 'Non-delinquent' in the placebo group.
 'Delinquent' in the control group.
 'Non-delinquent' in the control group.

The judges consisted of two experts, both of whom were very familiar with the Kohlberg theory. One judge had, prior to this study, scored protocols for research purposes, and the other had worked with the Kohlberg materials and had received scoring training at the Center for Morel Education at Harvard University. The adges' scores were compared to those of the researcher who had also received

training at Hauverd.

and relatively black and there are in the stated with reliability

JUDGES '	SCORING ON	THE KOHLBERG	MORAL	
	JUDGMENT	INSTRUMENT		

TABLE 7

Test	Subject	Judge 1	Judge 2	Judge 3
	2 A	2(1)	2(3)	. 2
	9A	2	3(2)	2
Pre	18A	2	3	2
	7B	2(3)	2(3)	2
•	19B	2-3	2(3)	2
-	20B	2(3)	2 (3)	2
	18	2(3)	3 (2)	2 (3)
	78	. 1	1	1
Post	10A	2(1)	2(1)	2(1)
	2B	8(2)	3(2)	2(3)
4	4B	2(3)	2 (3)	-2(3)
	178	2	2 (3)	2 ,
	5A	2(1)	2(1)	2(1)
C	78	1(2)	1(2)	1(2)
Follow-up	13A	2	2(3)	2(3)
	78	2(3)	2(3)	2(3)
	98	2	2	2
	16B	2-3	2(3)	2(3)

Percentage agreements on modal stage scores.

.

Between Judge 1 and 2 = 83.338Between Judge 2 and 3 = 77.778Between Judge 1 and 3 = 94.448.

Product-scent correlations.

Between Judge 1 and 2 = .928 Between Judge 2 and 3 = .887 Between Judge 1 and 3 = .964 .79

scores reported in other studies which used versions of the Kohlberg Moral Judgment Instrument. The degree of fit between judges' scores is somewhat of a problem when this instrument is used. For example, if a scorer arrived at a total of seven Stage 2 responses, and three Stage 3 responses for a particular subject's entire protocol, this subject's overall score would be 2(3), as thirty per cent of the subject's responses are at Stage 3, and seventy per cent at Stage 2. If, however, another scorer only scored two responses at Stage 3 and the same seven at Stage 2, the two Stage 3 responses would only account for twenty-two per cent of the subject's protocol and would therefore not be counted according to the Kohlberg scoring procedure. As both judges, however, perceived the character of the protocol as being Stage 2, interjudge agreement was initially based on modal stage scores only. Results of this reliability check are displayed in Table 4.

A more thorough reliability check was carried out on the posttest and follow-up test scores. This involved comparing total scores arrived at by each judge for each subject on the post and follow-up tests. Pretest scores could not be compared this way as one judge did not score for minor stages: this judge gave an overall modal score only for each subject. Table 8 presents mesults of this analysis.

TARE 8

PERCENTAGE AGREEMENTS BETWEEN JUDGES ON OVERALL MORAL REASONING DOCERS ON THE POST AND FOLLOW-UP TESTS

0	10.1		1	2.5	5.7		1.54		e 1	1.1.	aid			من من	سفذذن	1.14			-	 						- 14.,	- e •	'
					aria Lista Lista		Ŋ														bye pd	с А		*		Ň		
		5	1	-ma		θ¥/							97795 Serae		<u>е</u>	81.00					یک منجد رست	+	-		-	•		:
							66	.6						7	54						751							
						1.12			2.10		-	C.				- 1	ç. B	in at a										4

The percentage agreements are only moderate, but as was pointed out above disagreement between judges on but a few separate responses in one particular protocol can change the overall stage score. The nature of the scoring of the instrument, resting as it does on openended responses to various question, does not always allow for 'scientifically objective! scoring. Yet, as can be seen in Table 7 the nature of disagreement between judges on overall scores is very slight. In no case is there more than one-third of a stage disagreement between judges' scores. In light of the percentage agreements reached in other studies where agreement is based on judges' scores which are within one-third of a stage of each other, the agreement reached between judges in this study is regarded as satisfactory.

Scoring

Since the conception of the six stage model in 1958, Kohlberg and his associates have refined the scoring procedures. As of June 1975 (Kohlberg, 1975) all the materials on the introduction to scoring have been rewritten; only the actual scoring manual has not yet been revised. As was pointed out in Chapter II, the original thirty aspects of moral judgment had been collapsed to ten issues with elswen moral concerns, of which seven (self-interest, altruism, duty, conscience, rules, role-taking and justice) have been identified as central to the way in which a person appears to view an issue. However, as the 1973 scoring guide was the most facent available reference, this was used as the criterion for scoring, with clarification

cought from the 1975 Materiality

1 of moding was no find evidence of the etimoture of

the Subject's reasoning, with the manual acting as the structural criterion or 'independent variable.' A response was only scored if it corresponded to a point in the manual. This guarded against scoring what appeared to be cognitively complex responses at a higher stage than they actually were, or relying on content (what behavior the subject thought was right) as the sole indicator of stage of reasoning. Whereas one can predict, to an extent, choice from the stage of reasoning, one cannot predict stage from choice; e.g., ninety per cent of Stage 5 subjects think it would be right to steal frug in the Heinz dilemma, but then so do ninety per cent of Stage 2 thinkers. Therefore, a subject's response was only scored when it 'fitted! a response in the manual. For example, Subject 2A, on the pretest, replies to the question "Should Heinz steal the drug even if he didn't love his wife?"

Well, no, because what would he steal the drug for if he didn't love his wife?

This fits criterion concept B5 on page thirteen of the Form A scoring manual which reads, "The husband has no obligation or reason to risk stealing the drug if he does not love his wife." This sometimes meant that a few protocols had very sparse scores. Overall, however, scores were obtained for all subjects on each of the pre, post and follow-up tests.

The standard scoring procedure was used as this is the most widely accepted method and as the scoring manual is designed to faciliate this method.

Procedure

1. The scorer was familiar with the stages and had read through the scoring manual to orient himself to the "structure" for each issue.

2. The scorer read through the subject's responses to the entire dilemma and particular issue being scored.

3. Having ascertained the stage at which the subject appeared to be reasoning, the scoring manual was used to ascertain the stage responses given for each issue. Usually, this meant that the scorer started looking at the manual for Stage 1 criteria. If two Stage 1 responses were found these were entered in the scoring sheet. If only one could be found, this was entered and Stage 2 criteria were then examined. This process through the stages was continued until all responses had been scored or had been deemed unscorable. The number of concepts that can be scored at any one stage is limited to two because:

> There are individual differences in the variety with which a subject can express his or her underlying moral structure. We do not, for example, want to consider a subject who expresses six different ideas, all at Stage 3, to be any more 'solidly Stage 3' then another subject whose repertoire is at the same moral stage but is more limited in its variety. The standard of two concepts at a.stage by a compromise which is large enough to make it likely over the subject really possesses the structure of that stage thout requiring too great a demonstration (Kohlberg, Let III, 1975, p. 23).

Therefore, a given criterion concept is only used once per subject. For example, Subject 16A, on the pretest, replied to the question, "Should Joe give his father the money?"

> Mell . . . no, because Joe has earned hard earned money and he's the one who worked for it so it belongs to him.

In response to the question "Does his father have the right to ask Joe for the money?" the subject replied:

> Well no . . . probably . . . well he has the right . . but if I was in Joe's position I wouldn't give it up. I'd earned it, so I wouldn't give it up.

Both these responses fit the criterion B5 on page 69 of the Form A scoring manual which reads "Joe should refuse to give his father the money because he earned it, it's his money." It was, therefore, only scored once.

On completion of the scoring sheet for one issue, the next issue was scored until the entire test had been completed. The scoring proved to be very time consuming and often difficult. Two problems existed; one, because of the dilemmas used in this research, and two, because of the scoring manual. The posttest used contained the Captain in Korea (Instrument V, Appendix E) and the Troublemaker or Sickman (Instrument VI, Appendix E) dilemmas. The 1973 scoring manual does not contain these dilemmas so the previous manual had to be used and this does not contain examples of actual responses.

Therefore, the scoring of these latter two mentioned dilemmas had to be carried out on the basis of knowledge of the stages and on criteria in the old manual which may be out of date in light of the new knowledge gained since its publication. Secondly, although the scoring manual contains criteria for the conscience issue is and Bob dilemma (Instrument VII, Appendix E), it was noted by the researcher at the workshop held at Harvard in June 1975, that the examples given were not considered to be particularly valid. So

the conscience issue was not scored.

4. Having obtained scores for each of the issues an overall score is computed. All responses at a particular stage are totalled. If all responses are at one stage, the score is at that stage; e.g., Subject 8A on the pretest (Appendix K) who only has Stage 2 responses. If there is a mixed stage score then percentage weightings are used to determine major and minor stages. For example, Subject 1A on the pretest (Appendix K) has two Stage 1 responses, seven at Stage 2 and four at Stage 3. As the two Stage 1 responses only account for 15.4% of the total number of responses (2/13), they are rejected and Stage 2 becomes the dominant stage with Stage 3 (4/13 or 30.8%) becoming the minor stage.

5. Finally a moral maturity score (MMS) can be computed. This takes into account each stage scorable response and the percentage of those responses at each stage. This percentage at each stage is then multiplied by the stage number itself; e.g., Subject 1A on the posttest (Appendix K) has eleven Stage 2 responses (73.33%) and four Stage 3 responses (26.66%). Multiplying 73.33% by two and 26.66% by three realizes a moral maturity score of 226.

Flavell's Role-Taking Instrument

According to both theory and empirical research (Piaget, 1932; Kohlberg, 1958; Stuart, 1967; and Selman, 1971) role-taking ability is positively related to both cognitive and moral development. Selman (1971) posited that reciprocal role taking was a necessary condition for the development of conventional moral thought. The relationship should be as follows:

Moral Stage

Pre-conventional level

Conventional level

Role Taking Ability

Stage 1 Egocentric (non-reciprocal) Stage 2 Possibility of reciprocity Stage 3 Reciprocity necessary Stage 4 (Selman, 1971, adapted).

Administration

This instrument (Appendix F) was originally devised by Flavell (1968).

1. Each subject was shown the ordered series of seven pictures showing the story of a boy being chased by a dog, running to and climbing an apple tree with the dog nipping at his heels, then eating an apple as the dog trots away.

2. Each subject told an appropriate story about the pictures.

3. Three pictures were then removed from the set leaving those showing the boy walking, running to the apple tree, climbing it and eating an apple. Although the dog remains in the last but one

picture the 'fear of the dog' motive is removed.

4. Each subject was then asked to tell a story as if his teacher were looking at this four picture set. The subject was then asked, "Why would Mr./Mrs. A think the boy climbed the tree?"

Each subject was administered this test during the pretest session and each subject was tape recorded.

A categorical system of scoring was used to determine whether or not cognitive role taking ability had been demonstrated.

<u>Category 1</u> (egocentric). The subject could not perform any transformation of the original story. In both stories the angry dog

remained as the motivational force behind the boy's tree climbing.

Category 2. The subject could either:

(a) Tell a perceptually correct four picture story but reverted to the 'fear of dog' motive on questioning; or

(b) Use the 'fear of the dog' motive in the story but on questioning could indicate one or more other motives for the boy's tree climbing.

<u>Category 3</u> (reciprocal). The subject could tell a four picture story that the observer might tell by suppressing the original seven picture motivational scheme. On questioning, the subject could provide other than the 'fear of the dog' motive, or use this motive as one of many possible alternatives.

Validity

.

According to Piaget, the ability to decenter—to 'see' something from the viewpoint of another person—is a necessary condition for concrete and formal operational thought. According to Selman (1971) and Kohlberg (1958), this ability is also necessary for moral development. As was pointed out in Chapter II, role-taking ability can take a number of forms, one of which is cognitive role-taking ability. In this test the subject has to disassociate from his seven picture viewpoint in order to tell the four picture story from an external observer's viewpoint. The task "thus creates a gap between two roles or perspectives which the subject must bridge (Flavell, 1968, p. 76)." On this basis, the Flavell instrument would seem to have content validity, in that it assesses the ability to see something from other than one's own perspective.

88

Reliability

Interjudge reliability was ascertained by three judges, one of whom was the researcher, another a recent PhD graduate of the University of Alberta and the third was a graduate student in the Department of Elementary Education at the University of Alberta. Table 9 indicates the percentage agreement between the three judges.

TABLE 9

	ERJUDGE AGREEMEN DLE-TAKING SCORES		
 Judges 1 and 2	Judges 2 and 3	Judges 1 and 3	
94.73%	92.11%	97.37%	

Test-retest reliability was determined by readministering the instrument during the follow-up test period. Table 10 indicates the preand follow-up test role taking scores for each subject. The product moment correlation between scores on these two tests was .68.

The Intelligence Test

All subjects in Grade Six, as part of the Edmonton Public School Board testing program, had been administered the Canadian Lorge-Thorndike Intelligence Tests (Form D) just prior to the commencement of this research. Verbal and non-verbal scores were therefore collected from the subject's cumulative record. Grade Five subjects were administered Form C of this test in a group situation by the researcher just after commencement of the research. This particular test was chosen because some results were already available, and

•(

Subject	Test	Retest	S	ubject	Test	Retest
				1B	3	3
1A	2	3	· · ·	2B	2	2
2A	3	3		3B	3	· · · · 3
3A	3	3	•			3
4A	, 3	3		4B	2	3
5A	ຸ 3	3		5B	ຸ ວ	3
6A	1	1	· · · ·	6B ·		3
7A .	3	3	•	7B	j ∧3 ÷	3
8A.	3	,3		8B	3	
9A	3	3	Y	9B /	3	
10A	1	3		10в	3	s n
11A	3	3		11B	3	3
12A	2	2		12B	3	· 3
13A	3	3	1 - 1	13B	3	3,
14A	3	3 /		14B	3	3
15A	3	3		15B	ຸ3ຸ	\$
16A	3	. 3		16B	3	3
17A ,	3	3		17B	3	3
	3	3		18B	3 *	3
18A 🚬	5		· · ·	19B	. 3	3
				20B	2	2
er (1997)			•	· · · ·		

TABLE 10

ŝ

TEST-RETEST ROLE-TAKING SCORES FOR EACH SUBJECT

.

89

in terms of time, appropriate. Subjects had already spent at least half an hour in individual testing sessions and it was felt undesirable to take up any more subject or researcher time in this endeavour. As the test is standardized and is widely used, considerations of validity and reliability are not discussed here. Details can be obtained from the manual (Lorge, Thorndike, Hagen and Wright, 1967).

Summary

In total five different instruments were used in this study, one of which—the SBRI—was researcher designed. Figure 7 summarizes the purpose of the instrument, its validity and reliability and the type of data obtained.

3. TREATMENTS

Subjects were randomly assigned to one of the following three experimental groups.

1. Treatment. 2. Placebo. 3. Control. Table 11 indicates in which experimental group and in which school each subject was placed. All treatment and placebo sessions were researcher conducted.

The Treatment Group

Thirteen subjects were involved in six, one-half hour sessions held once per week in each of the three schools, over a six week period. Five of these sessions comprised the viewing of one of the <u>First Things</u> (Guidance Associates, 1972) filmstrips (Appendix G), followed by a discussion, the purpose of which was to stimulate moral reasoning. Each subject was encouraged to arrive at the best reason for saying that the character in the filmstrip dilemma should pursue

Instrument	Purpose	Administration	Validity	Reliability	Data Gainëd
Student Pehavior Rating	Selection of delinquent' and 'non- delinquent' subjects.	Individual Teacher	Sontent	Some inter- judge.	Names of students who could be classified in one Or more categories.
Fairmindedness and Trustworthi- ness Scale	Selection of ' delinquent' and 'nôn- delinquent' sùbjects.	Individual Teacher	Content	Some inter- judge.	Quantitative score.
Moral Judgment	To determine stage of moral reasoning.	Individual Student	Content Construct.	Inter-judge. Test/Retest.	 Qualitative responses categorized into quantitative stage scores.
Flavell's Role- Taking	To determine cognitive role-taking ability.	Individual Student	Content	Inter-Judge Test/Retest.	Qualitative responses categorized according to reciprocal or non- reciprocal levels.
Canadian Lorge- Thorndike Intelligence	To determine I.Q.	Group	•		Quantitative score.
			FIGURE 7		-

6

.

91

.'

•	Expe	rimental Groups,	· ·	
School	Treatment	Placebo	Control	
		4	2A	•
1	1A 6A	4A 5A	3A	•
		•		¢
.*	3 B	18	2B	
•	8B	4B	5B	
1	¢		99 78	
<u></u>				
2	9A	10A	<i>†</i> A	
-	11A	· •	8A	
	12A	•	•	
ъ. <u>.</u> .	9B	12B	14B	
	10B	13B		
· · · · · · · · · · · · · · · · · · ·	11B	15B		
		°	14A	
3	13A -	15A 18A	14A 16A	•
		104	17A	
and the second s			N	
Йа р а	17B	16B	188	· · ·
	20в	19B		
Totals				•
'Delinquent	s' 6	5	7	
'Non-deling	uents' 7	7	6	

COMPOSITION OF THE THREE EXPERIMENTAL GROUPS BY STUDENT IDENTIFICATION NUMBER AND CATEGORY, AND BY SCHOOL

TABLE 11

.

١

1

a certain course of action. The format followed was:



Throughout the session the researcher attempted to stimulate conflict, to point out alternatives, to clarify subject's responses and to reason at the +1 stage both for and against initial subject responses. The following types of questions were employed:

1. Issue specific. E.g., Should you always keep any promise?

2. Inter issue. E.g., Which is more important, a brother's birthday present or keeping a friendship?

3. One stage higher. E.g., What would happen if all parents acted like that?

4. Role switch. E.g., How would John feel? A concentrated effort was made to encourage subjects to argue amongst themselves, but despite this determination to remain in the background most questions and answers were directed to and through the researcher. However, as the sessions progressed, the subjects became more capable of carrying on a discussion amongst themselves and the researcher only initiated himself into the discussion when arguments ran out or were constantly regurgitated. As freedom of expression and a non-judgmental atmosphere regarding the dilemmas were emphasized and as subjects became better acquainted with each other, the discussions became more student organized.

The final session involved the discussion of dilemmas which subjects felt were relevant to their own lives. Most of the situations mentioned involved rules so discussion focused upon the usefulness and fairness of certain rules. These included the rule of being at home by a certain time at night, the rule often imposed by teachers that if one person in a class has performed a mis-demeanor and no one will own up then the whole class is punished, and rules prohibiting stealing; damaging property, lying and fighting. In School 3 the subjects attempted to arrive at one rule which they thought should in no circumstances be broken. They eventually justifiable exception. One was, 'Don't litter' and the other was 'Don't poison babies.'

The Placebo Group

Twelve subjects were involved in six, one-half hour sessions held once per week, in each of the three schools, over a six week period. The intent of these sessions was to interact positively with these subjects so that any possible researcher effect on the treatment group might be controlled on the posttest. This group played various Social Studies games (Appendix H), none of which involved issues regarding what was morally right or wrong.

The Control Group

Thirteen subjects were assigned to the control group. This group continued with normal classroom activities during the treatment and placebo group sessions. As far as can be ascertained, none of the control subjects was involved in the formal discussion of moral issues in classtime.

1. ANALYSIS OF THE DATA

The initial problem posed in this research involved the relationship between moral reasoning and moral conduct. In order to investigate this problem two groups were selected, one consisting of students deemed 'delinquent,' the other of those deemed 'nondelinquent.' Despite the fact that subjects were found in different classes in three different schools and were selected by different teachers, it was decided, for analysis purposes, that 'delinquents'
and 'non-delinquents' could be treated as non-equivalent groups. The following rationale was used.

1. The existence of some inter-judge reliability on student ratings on the SBRI.

2. The degree of commonality perceived in the various raters use of the SBRI categories. For example, reference to Table 1 will show that every 'delinquent' subject selected was labelled as requiring frequent punishment and nearly every subject (fourteen out of eighteen) was noted as interfering physically with others and as working in disharmony with others in group situations. In the case of all 'non-delinquent' subjects, all raters (Table 2) deemed the respecting and obeying of classroom rules as a criterion for selection, and nearly all subjects (nineteen out of twenty) were considered not to interfere physically with others. In other words, a common perception of 'delinquent' and 'non-delinquent' behavior appeared to exist despite the number of raters and despite the number of schools used. This is also borne out in the results of the Fairmindedness and Trustworthiness Instrument. Only one 'delinquent' subject (17A) was deemed to be high in fairmindedness (he was rated as a 2), and only one 'non-delinquent' (13B) was rated higher than a score of two in both the fairmindedness and trustworthiness components. For details see Table 3. Whereas the specific quantitative degree of similarity between each individual 'delinquent' or o'nondelinquent' subject is open to question, on a global basis it would appear that these two groups can be treated as discrete for analysis

purposes, and as on all items of the measures used there were

significant differences between the two groups.

Statistical Procedures

1

The following statistical procedures were used in order to attempt to gain answers to the specific problems posed in the study.

1. In order to attempt to answer the question of whether or not the 'delinquent' group differed from the 'non-delinquent' group on moral maturity scores, an analysis of variance test was applied to the pretest scores, derived from the Kohlberg Moral Judgment Instrument, for each of these two groups. The same procedure was used on these two groups' posttest and follow-up test moral maturity scores.

A <u>t</u> test was used to ascertain whether or not 'delinquents' and 'non-delinquents' differed on role-taking scores derived from the Flavell instrument.

2. Product-moment correlations were applied to the moral maturity scores and role-taking scores and the moral maturity scores and intelligence quotients of the entire sample in order to attempt to discover the relationship existing between these variables. This correlation procedure was also applied separately to the data from the 'delinquent' and 'non-delinquent' groups-in order to attempt to discover whether or not relationships among the above mentioned variables were similar for the two groups.

3. In order to test the effect of the treatments a threeway analysis of variance test was applied to pre, post and follow-up test moral maturity scores of 'delinquents' and 'non-delinquents' in the experimental, placebo and control group treatments. All statistical procedures were carried out on the advice of, and under the direction of the Educational Research Services in the Faculty of Education at the University of Alberta.

5. THE PILOT STUDY

The purposes of the pilot study were to:

1. Ascertain the validity and reliability of the research instruments.

2. Determine appropriate methods of administering the instruments.

3. Field test the experimental treatment materials. The pilot study was carried out in an Edmonton elementary school situated in a low socio-economic area. All subjects were in Grade Five (n = 4) and Grade Six (n = 5) and all were between the ages of ten years and four months and eleven years and six months. Four subjects (one in Grade Five and three in Grade Six) were deemed 'delinquent' by teacher raters, and five were deemed 'non-delinquent' (two in Grade Five and three in Grade Six).

Instrumentation

The Student Behavior Rating Instrument. This instrument in its initial form was unclear to teachers as it appeared that only 'delinquent' subjects had to be rated. Therefore the instrument was modified to ensure that both 'delinquents' and 'non-delinquents' would be rated (Appendix C). The categories displayed on the SBRI were regarded as valid by teachers and no other characteristics of 'delinquent' or 'non-delinquent' behavior were added by the pilot study raters or by other teacher judges contacted by the researcher. Teachers initially found the instrument difficult to use, but when it was pointed out by the researcher that he only required subjects who consistently displayed one or more of the characteristics stated on the SBRI, and that it was the individual teacher's perceptions which were of importance, then teachers were more willing to rate students. The only real difficulty noted by teachers was cases when a student vacillated between 'delinquent' and 'non-delinquent' behaviors. In these cases teachers were asked to use their own judgments. If a student was rated as both 'delinquent' and 'nondelinquent,' the researcher did not include the subject in the sample. Reliability of the teachers discussed their choices with each other. Therefore, in the major study efforts were made to prevent teacher/ teacher or teacher/principal interaction.

The Fairmindedness and Trustworthiness Instrument. This instrument (Appendix D) was judged by teachers to be relatively easy to use and was deemed effective in differentiating one student's behavior from that of another.

Flavell's Role-Taking Instrument. This instrument was deemed to be appropriate for the age level of the pilot sample. Only four out of the nine subjects could, according to this test, cognitively roletake. The test was therefore assumed to have discriminative value. A question posed regarding what the dog in the third picture of the four picture series was doing proved to be ineffective in noting differences in subjects' role-taking ability. Students said, "It's walking away." This question was therefore eliminated from the task. Test/retest agreement over a three week period was one hundred per cent . 99

and scoring, according to the criteria formulated, was deemed reliable.

The Kohlberg Moral Judgment Instrument. The interview technique was found to be the most appropriate form for collecting the data. Written comments were either incomprehensible or were too brief for scoring purposes. It was also evident that, by only posing the questions as stated in Appendix E, unclear or garbled responses were often obtained. Therefore, clarification was sought and questions regarding the reason for a particular response were asked. For example, a response of, "Stealing is wrong" was followed by the interviewer asking, "Why is it wrong?" On the basis of the pilot experience, the above procedure of seeking clarification, or probing ' responses, was adopted.

Scoring of the responses to the dilemmas proved initially to be difficult. The standard scoring procedure was found to be the most appropriate as it utilized all classifiable data. Although each question posed during the interview focused upon one particular issue (value of life, property, punishment and so on), student responses did not necessarily focus upon the particular issue under examination. The complete protocol had therefore to be examined. It was found that the majority of student responses could be categorized as to a stage level and the pilot experience of interviewing and scoring proved to be invaluable to the researcher.

Field Testing of the Experimental Treatment Materials. The field testing of the experimental treatment materials demonstrated that the selected First Things filmstrips (Guidance Associates, 1972)

were appropriate for Grade Five and Six students. They provoked argument and it appeared that students related to the dilemmas posed. Group discussions were effective but it was apparent that teacher facilitation was necessary at times to stimulate further discussion.

Summary

On the basis of the pilot study experiences the following observations were noted:

1. The SBRI was shown to be effective in describing student behavior, however, the format was modified to make it more comprehensible to raters and more stringent reliability measures were seen to be needed.

2. The Role-Taking and Kohlberg Moral Judgment Instruments were shown to be appropriate instruments for collecting the data needed. Revisions were made to the testing formats and the scoring methods were determined.

3. The experimental treatment materials were chosen and were found to be appropriate in content.

6. SUMMARY

This chapter discussed the selection of the sample, the instrumentation used, the analysis of data procedures and the pilot study.

The major study sample consisted of eighteen subjects in three schools who were deemed to display 'delinquent' behavior by teacher and/or principal raters, and twenty subjects deemed 'nondelinquent,' chosen by the same teacher and/or principal raters in TABLE 12

						<u>.</u>	
		Age			*Ex.	I.Q.	I.Q.
ubject	Sex	Yrs.Mos.	School	Grade	Group	Verbal	Non-verbal
1A	м	12.0	1	6	Е	91	88
2A	M	11.4	1	6	C	73	80
3A	F	13.1	1	6	C	88	99
4A -	M	10.5	1	5	P	94	96
5A	M	11.4	1	5	P	102	117 *
6A	M	10.9	1	5	E	120*	113
7A	M	11.0	2	5	• C	93	97
8A	M	10.11	2	5	C `	94	101
9A	M	11.9	2	6	E	107	119
10A	M	11.9	2	6	P	140	129
11A	M	13.1	2	6	E	78	91
12A	M	10.5	2	5	E	80	83
13A	F	12.1		6	E	81	66
14A	M	12.1	*3 `3	6	C	77	70
15A	M	11.10			P	110	117
16A	M	11.10	3 3	k	C	104	90
17A	M	11.9		• 6	C	84	78
18A	M	11.7	3	6	P	86	121
18	F	12.0	1	6	P	105	125
2B	F	11.11	1	6	c	105	₹ 95
2 <i>B</i> 3B	M	12.3	- 1	6	Ē	119	99
3B 4B	F	11.10	ī	6	P	126	127
5B	F	11.10	1	5	Ċ	135	130
5B 6B	F	11.3	1	5	C	130	108
ов 7в	F	11.3	î	5 5	° C	117	111
8B	M	11.0	1	5	E	121	128
9B	F	10.8	2	5		104	121
10B	M	10.9	2	5	E	104	97
10B	F	11.4	2	6	E	135	137
11B 12B	F	12.3	2	.6	P	80	104
12B 13B	M	12.3		5	P	95	113 .
	•	10.10	2 2	5	C	94	119
14B	M	10.9	2	6	P	83	84
15B		12.2	3	6	P	116	111
16B	F		3	6/		110	103
17B	M	11.6 12.1		Q. 1	C E	119	105
18B	F		3	6	P	122	129
19B	F	12.3	3	16 16			129
20B -	F	12.0	3	•	E	110	167

SUMMARY OF THE CHARACTERISTICS AND LOCATION OF THE SAMPLE

= Experimental treatment. = Placebo: E

P

C

1

= Control.

the three schools. All subjects were aged between ten years and five months, and thirteen years and one month, with the average age of the entire sample being approximately eleven years and five months. All subjects were in Grade Five or Six classes. Table 12 consolidates information regarding the sample. Each subject was administered the Kohlberg Moral Judgment Instrument (Form A) and Flavell's Role-Taking Instrument, and an intelligence quotient was obtained from the Canadian Lorge-Thorndike Intelligence Tests. All subjects were then randomly placed in one of three groups: (1) An experimental treatment group in which moral dilemmas were discussed; (2) A placebo group in which Social Studies games were played; (3) A control group which carried on with normal classroom activities. On completion of. the six week intervention, a posttest consisting of Form B of Kohlberg's Moral Judgment Instrument was administered. This was followed a month later by the readministration of Form A of Kohlberg's Moral Judgment Instrument and Flavell's Role-Taking Instrument.

Chapter IV

FINDINGS OF THE STUDY

Introduction

This exploratory study focused upon the relationship between moral reasoning and conduct among Grade Five and Six students who had been rated by teachers, according to a researcher designed Student Behavior Rating Instrument, as 'delinquent' or 'non-delinquent.' A second concern related to the effect of participation in moral discussions upon the moral reasoning of 'delinquent' and 'non-delinquent' subjects. Eighteen 'delinquents' and twenty 'non-delinquents,' in three Edmonton schools, were administered, in an interview situation, Kohlberg's Moral Judgment Instrument (Form A) and Flavell's Role-Taking Instrument. Intelligence quotients, based on the Canadian Lorge-Thorndike Intelligence Tests, were obtained from Grade Six subjects! cumulative records, and Grade Five subjects were administered the tests in a group situation. Following the administration of these instruments all subjects were randomly assigned to one of three groups: (a) an experimental treatment group which participated in moral discussions; (b) a placebo group which played games related to Social Studies education; and (c) a control group which carried on with normal classroom activities during the experimental treatment and placebo group sessions. Both the experimental treatment and placebo groups, which were conducted by the researcher, were involved in six, one-half hour sessions spread over a six week period. At

the end of this time all subjects were administered Form B of Kohlberg's Moral Judgment Instrument. A month later Form A of the Kohlberg Instrument and Flavell's Role-Taking Instrument were readministered, the latter test being included for test-retest reliability purposes.

This chapter reports the findings of the study. Each question posed in the study is reported separately, and a discussion of each finding follows.

1. FINDINGS OF THE STUDY

Question 1

Do Grade Five and Six 'delinquents' and 'non-delinquents' differ in the scores obtained on Kohlberg's Moral Judgment Instrument

Table 13 displays the pretest results of the scoring of responses made on the Kohlberg Moral Judgment Instrument (Form A), with the stage of response and the moral maturity score (mms) for each subject in the 'delinquent' and 'non-delinquent' groups being shown. For example, Subject 20B obtained a stage score of 2(3). This indicates that this subject had a modal Stage 2 orientation (60% of her responses), and, as 40% of her responses were scored at Stage 3, the overall score is displayed as 2(3). This, when quantified, yielded a moral maturity score of 240. Table 13 also indicates that 'delinquents' in the sample tend to reason more in Stage 1 and 2 terms than do 'non-delinquents.' Only two 'delinquent' subjects (1A and 4A) demonstrated 25% or more Stage 3 reasoning according to

'D	elinquen	t' Group	'Non	'Non-Delinquent' Group					
Subject Stage	Moral Maturity Score	Subject	Stage	Moral Maturity Score					
1A	2(3)	215	18	2(3)	240				
2A	2(1)	177	2B	3(2)	273				
3A	2	183	3B	2(3)	227				
4A	2(3)	231	, 4B	2	221				
5A	1-2	150	5B	3(2)	280				
6A	2	208	6B	2(3)	227				
7A	1(2)	130	7 B	2(3)	230				
8 A	2	200-	8B	3(2)	272				
9A	2	206	9B	2	179				
10A	2	184	10B	2(3)	210				
11A 3	2	191	11B	2(3)	228				
12A	1-2	- 150	12B	2	184				
13A	2	182	13B	2	200				
14A	2(1)	163	14B	2	206				
15A	2	191	15B	2	208				
16 A	2	206	16B	3(2)	273				
178	2	212	17B	2	192				
18A	5 2	206	18B	2(3)	221				
	1		19B	2-3	250				
			20B	2 (3)	240				

1.

· ...

ł

TABLE 13

scored responses on the Kohlberg instrument. Yet four 'nondelinquent' subjects (2B, 5B, 8B and 16B) demonstrated Stage 3 reasoning as a modal stage, and nine subjects (1B, 3B, 6B, 7B, 10B, 11B, 18B, 19B and 20B) apparently had 25% or more Stage 3 thinking as part of their repertoire. Conversely, whereas five 'delinquent' subjects (2A, 5A, 7A, 12A and 14A) used some Stage 1 reasoning, no 'non-delinquent' subject used enough Stage 1 reasoning to account for 25% of their total score, or they used no Stage 1 reasoning at all. This is indicated in Appendix K which displays the number of scored responses at each moral stage for each subject.

An analysis of variance was performed on the 'delinquent' and 'non-delinquent' moral maturity scores. Table 14 displays the results of this analysis. As is indicated, the 'delinquent' and 'nondelinquent' groups differed significantly from each other. This statistically significant difference was maintained on the posttest and follow-up test results as Table 14 indicates.

Discussion

 S_{i}

As the inter-judge scoring reliability was relatively high (Tables 7 and 8), it can be stated with some confidence that, for this sample of study the group perceived by teachers as being 'delin" quant' differed in moral reasoning, as assessed by the Kohlberg Moral Judgment Instrument, from the group perceived as being 'nondelinquent.' This finding is notable in light of the research studies on the moral reasoning of legal delinquents (Kohlberg, 1958; Ewanyk, 1973) in which delinquents were found to reason predominantly at the preconventional level. Despite the difference in terms of age TWO-WAY ANALYSIS OF VARIANCE RESULTS ON PRE, POST AND FOLLOW-UP TEST MORAL MATURITY SCORES OF 'DELINQUENTS' AND 'NON-DELINQUENTS' IN THE THREE TREATMENT GROUPS

TABLE 14

Pretest Source Sum of Squares Degrees of Freedom F. Ratio Probability Α 15139.70 1 17.83 .001 В 89.14 · 2 .05 .94 Test for additivity. p = .41Test for homogeneity of variance. p = .99Posttest Source Sum of Squares Degrees of Freedom F. Ratio Probability A 23792.90 · • . 1 21.37 .001 B· 1328.68 2 .59 .55 Test for additivity. p = .27Test for homogeneity of variance. p = .56Follow-up test Source Sum of Squares Degrees of Freedom F. Ratio Probability Ä 18351.30 1 20.81 .001 B 658.54 2 . 37 .69 Test for additivity. p = .42Test for homogeneity of variance. p = 1.76

Source A = Groups; 'Delinquent': 'Non-delinquent' Source B = Treatments; Experimental: Placebo: Control.

and status, this study apparently demonstrated that upper elementary school students who were perceived in terms of delinquency by teachers also tended to reason predominantly at the preconventional level.

It is also notable that 'non-delinquents,' in the sample studied, tended to reason, at least in some of their responses, in Stage 3 terms. This, to an extent, supports Kohlberg and Turiel's (1971, p. 460) conclusion that typical children should be reasoning at the conventional level by early preadolescence and to reflect this level of reasoning in their conduct. It would appear, from teacher ratings on 'non-delinquents,' that these students may have shown "a decent regard for the core expectations and approval of parents, peers and outside authorities (Kohlberg and Turiel, 1971, p. 460)."

However, non-delinquency is not a sign of having reached the conventional level, as is indicated by the seven Stage 2 thinkers in the 'non-delinquent' group. The findings of this study do not necessarily demonstrate a cause and effect relationship between moral reasoning and conduct as there may be, as is pointed out by Krebs and Kohlberg (1973), a variety of other factors which influence behavior.

This study seemed to indicate that there may be possible differences in the moral reasoning structures of 'delinquents' and 'non-delinquents.' This finding may form the basis for building hypotheses as to the relationship between reason and conduct among Oper elementary school children. Question 2 .

Will a situation in which a random selection of 'delinquents' and 'non-delinquents' are exposed to moral reasoning at one stage above their pretest stage influence their reasoning to the extent that their stage of reasoning is higher on the posttest than it was on the pretest?

Question 2A

Will any changes be reflected on a follow-up test administered one month after the posttest?

Question 2B

Will there be any differences between pre, post and follow-up test scores on the Kohlberg Moral Judgment Instrument between 'delinquents' and 'non-delinquents' in experimental, placebo and control groups?

These questions were investigated together as findings for each question were based on a three-way analysis of variance test.

Reference to Table 14 will indicate that, when 'delinquents' and 'non-delinquents' were randomly placed into the three treatment groups, the probability of any difference existing between these three groups on pretest moral maturity scores was .95. This suggests that the randomization process was effective?

It was decided that, to answer the questions pertaining to treatment effect, posttest moral maturity scores would not be taken into account. This was done for the following reasons. Firstly, reference to Table 16 will indicate that a decline in mean moral maturity scores seemingly occurred between the pre and posttest for all groups and treatments except for the 'non-delinquents' in the control group and 'non-delinquents' in the experimental treatment group. For example, experimental treatment 'delinquents' tended to score lower on the posttest ($\overline{x} = 171.8$) than on the pretest ($\overline{x} = 192.00$), whereas pre and follow-up test scores were very similar; the follow-up mean score being 193.33. In overall terms, twenty-eight subjects decreased in moral maturity scores between the pre and posttest, whereas nine increased and one subject had the same pre and posttest score (Table 15).

Secondly, the product-moment correlations between the pre, post and follow-up tests (Table 17) indicate that the most positive correlations were found between the pre and follow-up tests. This was borne out for the entire sample and for 'delinquents' and 'nondelinquents' in the three treatment groups taken separately. The only exception was in the correlations found between control group 'delinquents' pre, post and follow-up test scores, in which the most positive correlation was between the post and follow-up test (r = .95)'.

Thirdly, the nature of the posttest itself may have been a factor in explaining the vacilations in correlation coefficients and mean scores. The posttest was Form B of the Kohlberg instrument, whereas Form A was used as the pre and follow-up instrument, and Form B deals with three issues (Property-Trust: Governance and

TABLE 15

1.1

PRE, POST AND FOLLOW-UP TEST MORAL MATURITY SCORES FOR 'DELINQUENT' AND 'NON-DELINQUENT' SUBJECTS IN THE EXPERIMENTAL TREATMENT, PLACEBO AND CONTROL GROUPS

S.			'.	ELINQUENI	'S'					
Tests*	-	rimental atment		Pla	Placebo			Control		
	Subject	Stage	MMS	Subject	Stage	MMS	Subject	Stage	MMS	
1 2 3	1A	2 (3) 2 (3) 2 (3)	215 226 225	4A	2 (3) 2 2 (3)	231 182 226	2A	2(1) 2(1) 2(1)	177 163 170	
1 . 2 3	6A	2 2 (1) 2	208 182 2 3 2	5A `	1-2 2(1) 2(1)	150- 170 180	3A	2 1-2-3 2	183 198 198	
1 2 3	98	2 2(1) 2	206 160 190	104	2 2(1) 2	184 182 214	7A	1(2) 1 1(2)	130 118 138	
1 2 3	11A	2 2(1) 2(1)	191 154 180	15A	2 2(1) 2	191 154 190	8 8	2 2(1) 2	200 163 176	
1 2 3	12 A	1-2 1 (2) 1-2	150 135 150	18 A	2 2 2	206 219 198	14A	2(1) 2(1) 2(1)	163 157 162	
1 2 3	13A	2 2(1) 2	182 174 182				16A	2 2 2	206 198 206	
1 2 3		•			<u>y</u>		17 A	2 2 2	212 210 198	

Test 1 = Pre

Test 2 = Post

Test 3 = Follow-up

			'NC	N-DELINQU	ENTS '		· · ·		
Tests*		rimental atment		Placebo			Control		
	Subject	Stage	MMS	Subject	State	MMS	Subject	Stage	MMS
1		2(3)	227		₹ 2(3)	240		3(2)	27:
2	3B	2(3)	2,27	•1B	2	193	2B	3(2)	239
3	•	3(2)	297		2(3)	236		3(2)	274
1	• · · ·	3(2)	272	•	2	221		3(2)	28
2	8B	2(3)	240	4B	2(3)	241	5B	3(2)	254
3		3 (2)	287		2(3)	237		3(2)	270
1	0	2	179		2	184		2(3)	22
2	9B	2	198	12B	2(1)	165	6B	2-3	23
3		2	206		2	188		2(3)	214
-1		2(3)	210		2	200		2 (3)	230
2	10B	2	208	13B	2(1)	154	7B	3(2)	28
3		2(3)	225		2	210	•	2(3)	23:
1		2(3)	228		2	208		2	20
2	11B	2(3)	240	15B	2	205	14B	2	21(
3		2(3)	240		2	207		2	208
1 .	, ···	2	192		3 (2)	273		2(3)	22]
2	17B	2	181	16B	3	297	188	2(3)	24(
3		2	191		3 (2)	268	``````````````````````````````````````	2(3)	227
1	1″	2(3)	240	• • • • • • • • • • • • • • • • • • • •	2-3	250			
2	20B	3(2)	264.	, 19 8	2(3)	204		•	•
3		3(2)	255		2(3)	243			

TABLE 15 (Continued)

٥

host 1 s Pre

Test 2 = Post

Test 3 = Follow-up

TAE	ILE	16	•

4

•

ç.

MEAN	MORAL	MATU	RITY S	COL	ÆS F	ÒR	'DELI	INQUENT	S' AND
1	NON-DEI	LINQU	ENTS '	IN	EACH	OF	THE	TREATM	ENT
	* GROU	JPS A	r each	I TE	ST A	DMI	NISTR	NOITAS	

	Experimental Treatment	Placebo	Control	
'Delinquent'	192.00	192.40	181.57	
'Non-delinquent'	221.14	225.14	239.50	Pretest
			· · · ·	
'Delinquent'	171.83	181.40	172.42	
'Non-delinquent'	222.57	208.42	243.33	Posttest
		t	•	
Delinquent'	193, 33	201.60	180.85	Follow-up
Non-delinquent'	243.00	227.00	236.66	test
	······································			

• 114

,

PRODUCT-MOMENT CORRELATIONS BETWEEN PRE, POST AND
FOLLOW-UP TEST MORAL MATURITY SCORES FOR THE
ENTIRE SAMPLE AND FOR 'DELINQUENTS' AND
'NON-DELINQUENTS' BY TREATMENT GROUP

TABLE 17

Group*	Treatment	** Test	Pretest(1)	Posttest(2)	Follow-up test(3)
Entire	e Sample	1 2 3	1.00	.78 1.00	.88 .77 1.00
D.	1	1	1.00	.75 1.00	.89 .83 1.00
Non-D.	1	23	1.00	.79 1.00	.82 .70 1.00
D.	2	1 2 3	1.00	.37 1.00	.78 .25 1.00
Non-D.	2	1 2 3	1.00	.78 1.00	.97 .83 1.00
D.	3	1 2 3	1.00	.87 1.00	.79 .95 1.00
ion-D.			1.00	.35 1.00	.97 .36 1.00

'Delinquent.' Non-D = "Non-delinquent.' 1 = Experimental. 2 = Placebo. 3 = Control.

Citizen Role) which are not major issues in Form A. As was indicated in Chapter III the posttest caused scoring problems in that two of the dilemmas on the test ('The Captain in Korea' and 'The Troublemaker and the Sickman') were not included in the most recent, ~revised, scoring manual. The former scoring manual had to be used and this manual is not as thorough as the new one. Also one of the posttest issues (conscience) was not scored for due to doubts by Kohlberg and his associates on the scoring of this particular issue.

The pilot study apparently demonstrated that the posttest was relatively easy to score. This was probably due to the fact that the pilot study subjects all appeared to reason in Stage 1 or 2 terms. Their responses also tended to be short as the researcher, who was gaining experience in the interviewing technique, asked few probing questions. However, scoring difficulties were encountered in the major study. Subjects tended to say more, probably because the researcher was a little more sophisticated in his interviewing technique and in the use of probing questions.

On this rationale, therefore, it was decided that the most satisfactory measure of any treatment effect would consider only pre to follow-up test scores. In this regard an analysis of variance was applied to the pre and follow-up test moral maturity scores of 'delinquents' and 'non-delinquents' in the three treatment groups. Results of this analysis are displayed in Table 18.

As is indicated the probability that 'delinquents' and 'nondelinquents' differ on pre and follow-up test scores combined was .001, This lends support to the finding indicated for Question 1.

TABLE	18 [·]	
-------	-----------------	--

ANALYSIS OF VARIANCE. STATISTICS ON PRE AND FOLLOW-UP TEST MORAL MATURITY SCORES FOR 'DELINQUENTS' AND 'NON-DELINQUENTS' IN THE EXPERIMENTAL, PLACEBO AND CONTROL GROUPS · · .

Sum of Source Squares				Probability	
A	32667.21	1	20.27	.001	
B	97.07	2	.03	.97	
AB	2462.69	2	.76	.47	
С	490.04	1	3.78	.06	
AC	63.55		.49	.49	
BC	558.66	2	2.16	.13	
ABC	684.58	2	2.64	.09	

.

ŝ

ł

A = 'Delinquents' and 'Non-delinquents' B = Experimental, Placebo and Control Treatments

C = Pre and follow-up test moral maturity scores . .

However, when treatment effect was considered, the probability that there was any difference between the three treatment groups, and pre and follow-up combined moral maturity scores for both 'delinquents' and 'non-delinquents,' was .97. Neither was there any significant difference (p = .47) when 'delinquents' and 'non-delinquents' were separated. When pre to follow-up test scores were analysed, there was the probability of .06 that the entire sample differed on these two measures, but when 'delinquents' and 'non-delinquent' scores were analysed separately the probability of any pre to follow-up test changes was .49, and when treatment effect is considered there was a probability of .13 that treatments influenced pre to follow-up test scores of the entire sample. As there was no significant interaction effect (p = .09) it would appear that the treatments had no significant effect on the moral reasoning scores of either 'delinquents' or 'non-delinquents.'

The only slight indication of any treatment effect was in the difference existing between experimental treatment 'non-delinquents'' pre to follow-up test scores. Reference to Table 16 will indicate that they made the largest gain in mean scores (pretest X = 221.14; follow-up test X = 243.00). This indication of change was borne out; to an extent, by the two experimental treatment 'non-delinquents' (Table 15) who apparently changed from a pretest stage score of 2(3), to a follow-up stage score of 3(2). The only other changes were in the placebo group where one subject (5A) was scored at 1-2 on the pre-test, and 2 on the follow-up test; and subject 4B who apparently moved from a Stage 2 moors of 2(3).

Discussion

The finding that an experimental treatment, which involved subjects in moral discussions, seemingly had little or no effect on follow-up test moral reasoning scores is not in keeping with some previous research studies, but is in keeping with others. Turiel (1966) apparently showed that exposure to +1 reasoning, during three experimental treatment sessions, leads to gains in moral maturity scores. Tracy and Cross (1973) found the same phenomenon and Blatt and Kohlberg (1973) discovered gains after twelve hours of experimental treatment sessions. However, Beck, Sullivan and Taylor (1973) found no substantial gains on their posttest measure; gains occurred one year after the treatment sessions. One can only question as to why there are these differences in findings.

Of interest in this study is the slight indication that 'non-delinquents' appeared to increase in moral maturity scores between the pre and follow-up test, whereas no other group showed much or any substantial increase. However, it should also be noted that one subject (17B) did not make a pre to follow-up test gain.

Question 3

Is role-taking as assessed by the Plavell Role-Taking Instrument related to stage of moral reasoning? For 'delinquents'? For 'nondelinquents'?

Table 19 displays the pre and follow-up test role-taking scores and the moral mesodalay accres for the "Selingment' and "hom-delinquent' groups: 20 is indicated the majority of subjects

'Delinquent' Group					'Non-delinquent' Group					
Role-taking					·	Role-	taking			
Subject	Stage	MMS	Test 1	Test 2	Subject	Stage	MMS	Test 1	Test 2	
la	2/21	015				·····				
	2(3)	215	2	3	18	2(3)	240	3	3	
2A	2(1)	177	3	3	2B	3(2)	273	2	2	
3 A	2	183	3.	. 3	3B	2 (3 the	: 227	3	3	
4A	2(3)	231	· 3	3	4B	2	221	3	3	
5A	1-2	150	3.	3	5B	3(2)	280	3	3	
6A	2	208	1	1	6B	2(3)	227	3	3	
7A	1(2)	130	3	3	7B	2(3)	230	3	. 3	
8 a	2	200	- 3	3	• 8B	3(2)	272	3	-3	
98	2	206	3	3	- 9B	2	179	3	3	
10A ·	2	184	1	3	10B	2(3)	210	3	3	
11A	2	191	3	3	11B	2(3)	228	3	3	
12A	1-2	150	2	2	12B	- 2	184	3.		
13A	2	182	3	3	13B	2	200	3	3	
14A	2(1)	163	3	3	14B	2	200	3	3	
15A	2	191	3	3	· 15B	2	208		3	
16A	2	206	3	3	16B	· · · · · · · · · · · · · · · · · · ·		3	3	
178	2	212	3	3	10B 17B	3(2)	273	3	3	
18A	2	206	3	3		2	192	3	3	
an the second	-			.	18B	2(3)	221	3	3	
					19B	2-3	250	3	3	
		e. A series de la companya de la company			20B	2(3)	240	. 2	3	

1

5-

PRE AND FOLLOW-UP TEST ROLE-TAKING SCORES AND MORAL MATURITY SCORES FOR THE 'DELINQUENT' AND 'NON-DELINQUENT' GROUPS

TABLE 19

(thirty-two out of thirty-eight) were apparently able to cognitively role-take (scored as 3). Most subjects, it seemed, realized, on the Flavell instrument, that an observer to the four picture set would not have the same information that they had having seen the seven picture set. Only two subjects (6A and 10A) could not, it seemed accomplish this task and four subjects (1A, 12A, 2B and 20B) could not be categorized as clearly showing the ability or lack of ability to cognitively role-take.

The product-moment correlation coefficient between roletaking scores and pretest moral judgment scores for the entire sample was r = .02, indicating that a subject's score on the roletaking instrument was likely to be anywhere within the total sample range of moral judgment scores. Figure 8 graphically demonstrates this. As might be expected from the above correlational analysis on the entire sample, product-moment correlations between the 'delinquent' group's moral reasoning and role-taking scores, and the 'non-delinquent' group's moral reasoning and role-taking scores, were close to zero. For the 'delinquent' group,the r = -.07, and for the 'non-delinquent' group the r = -.32.

Question 3A

Do 'delinguents' and 'non-delinquents' differ on role-taking ability?

A t test was applied to the mean sole-taking scores of the 'delinqueut' and "non-delinquent' groups. The t value redlized was 1.32. As a solution palos of 2.02 is required for df 40 at the .05 level of confidence it can be concluded that 'delinquents' and



THE RELATIONSHIP BETWEEN ROLE-TAKING SCORES AND MORAL MATURITY SCORES

'non-delinquents,' in this study, did not significantly differ from each other on role-taking ability.

Discussion

The pilot experience apparently demonstrated that the Flavell Role-Taking Instrument had discriminative value in that four out of the nine pilot study subjects could seemingly not cognitively roletake. However, in the major study the majority of subjects demonstrated cognitive role-taking ability and the relationship between moral reasoning and role-taking ability was close to zero (r = .02). The cognitive-developmental theory of moral development posits that role-taking ability is related to the stage of moral development and empirical findings support this view. One, it appears, has to demonstrate reciprocal role-taking ability in order to reason at Stage 3

only to assess of another perso perceive that an the 'simultaneous' that I think'-it d think that he think children who reason

on Kohlberg's

r, the Flavell instrument would appear seeing something from the perspective not assess the subject's ability to would have any knowledge of the subjecttaking ability of 'I think that he thinks asesses the ability of perceiving that 'I This latter shility may be demonstrated by Stage 2 terms. Therefore the Flavell test did not discriminate between Stage 2 and Stage 3 thinkers. Selman (1971), using the same Playell test, came to the same conclusion. The anomaly in this study was that there were three subjects (1A,

28 and 208) who could not be scored on the role-taking task and yet constanted some Stags 3 reasoning. However, on the retest these

subjects apparently demonstrated role-taking ability (Table 18). Only two subjects (6A and 10A) could not seemingly cognitively role-take and one of these (10A) scored 3 on the retest. The only subjects who could not be categorized on the retest were subjects 12A and 2B, the former being a dominant Stage 2 thinker and the other a dominant Stage 3 thinker. In light of the theory and empirical findings, this latter result is hard to explain. The latter subject did not mention "the fear of the dog motive in his story but on being questioned as to why the observer would think the boy climbed the tree, he said "Maybe he wanted an apple and also he saw that dog." On the basis of the statement, "that dog," the subject was scored at 2. In order to probe the statement made by the subject, the researcher could have questioned the subject further. This was unfortunately not done.

Question 4

Is intelligence quotient as assessed by the <u>Ganadian</u> <u>Lorge-Thorndike Intelligence Tests</u> related to stage of moral development? For 'delinquents'? For 'nondelinquents'?

Table 20 displays subject's moral reasoning scores and intelligence quotients, both verbal and non-verbal. The mean intelligence quotients, both verbal and non-verbal, were different for both groups. The range in scores was, however, similar for both 'delinquents' and 'non-delinquents.' Verbal intelligence quotients ranged between 73 and 140, and non-verbal ranged between 70 and 129 for the 'delinquent' group, whereas, for the 'non-delinquent' group, the verbal intelligence gnotient ranged between 83 and 135, and

MORAL MATURITY SCORES AND VERBAL AND NON-VERBAL	
INTELLIGENCE QUOTIENTS FOR THE 'DELINQUENT'	
AND 'NON-DELINQUENT' GROUPS	

TABLE 20

'Delinquent' Group					'Non-delinquent' Group					
	Intelligence Quotient				•.	Intelligence Quotient				
Subject	Stage	MMS		Non- verbal	Subject	Stage	MMS	Verbal	Non- verbal	
18	2 (3)	215	91	88	18	2(3)	240		125	
2A	2(1)	177	73	80	2B	3(2)	.273	105	95	
3A	2	183	88	99	3B	2(3)	227	119	. 99	
4A	2(3)	231	94	96	4B	2	221	126	127	
5 A	1-2	150	102	117	o 5B °	3(2)	280	135	130	
6A	2٠	208	120	113	6B	2(3)	227	130 🦈	108	
7A	1(2)	130	93	97	7B	2(3)	230	* 117 📜	111	
8A	2	200	94	101	8B	3(2)	272	121	128	
9 A	2	206	107	119	9B	2	179	104	121	
10A	2	184	140	129	10B	2(3)	210	104	97	
11A	2	191	78	91	118	2(3)	228	135	137	
12A	1-2	150	80	83	12B	2	184	80	104	
13A	2	182	81	66	13B	2	200	95 •	113	
147	2(1)	163	77	70	14B	2	206	94	119	
15A	2	191	110	117	15B	2	208	83.	.84	
16A	2	206	104	90	16B	3(2)	273	116	111	
17Å 👘	2	212	84	78	17B	2	192	114	103	
188	2	206	86	121	18B	, 2 (3)	221	119	106	
			= 04 FF	07 50	19B	2-3	250	122	129	
		••••	x 94.55	97.50	20B	2(3)	240	110	129	
		•	0		c		x	111.7	113.8	

non-verbal between 84 and 137. Product-moment correlations between moral reasoning scores and intelligence quotients for the entire sample (n = 38) are displayed in Table 21. The correlation coefficient between pretest moral reasoning scores and verbal intelligence quotient was .54. This was significant (p = .001). The correlation with non-verbal intelligence quotient was .41. This also was significant (p = .01). This positive correlation indicates that the higher the intelligence quotient, the higher the moral reasoning score, and conversely the lower the intelligence quotient the lower the moral reasoning score.

Correlational analysis of the relationship between 'delinquents' moral redsoning scores and intelligence quotients realized product-moment correlations of .17 with verbal intelligence and .13 with non-verbal, indicating that for this group the relationship was more of a random nature; a subject was just as likely to be high on moral reasoning and low on intelligence quotient as vice versa.' However, the correlations for the 'non-delinquent' group were .53 ' and .29, indicating a more positive relationship.

Discussion

The relationship between moral reasoning scores and intelligence quotient for the entire sample is in keeping with the findings from other studies (Harris, 1970; Lydiat, 1973; Kohlberg, 1969). As Kohlberg (1968, p. 1030) says, if a subject is high in moral reasoning he tendarto be high in intelligence quotient, but the converse is not necessarily true. This was demonstrated in the 'delinquent' sample where a high intelligence quotient did not necessarily mean a high

score in moral reasoning. The finding of this study would, in the main, support the idea that intelligence is a necessary, but not sufficient condition for moral reasoning.

2. CONCEUSIONS

ĝ.

Findings in this study indicated that 'delinquents' and 'nondelinquents' differed significantly from each other on pretest moral maturity scores ($\oint = .001$). This difference was maintained on the post and follow-up test measures. Moral reasoning was found to be significantly related to both verbal and non-verbal intelligence quotients for the entire sample (r = .53 and r = .41 respectively). Role-taking ability was not related to moral maturity and it was hypothesized that, despite the role-taking task's apparent discriminative ability in the pilot study, the instrument did not discriminate in the main study because it did not assess reciprocal role-taking ability. This latter ability is seemingly necessary for Stage 3 reasoning, whereas the Flavell instrument only assessed the ability of seeing something from one other person's viewpoint; an ability which Stage 2 thinkers can demonstrate. 'Delinquents' and 'nondelinquents' did not differ from each other on role-taking.

As the posttest instrument caused problems, the most satisfactory change assessment was regarded as differences between pre and follow-up test moral maturity scores. In this regard the only indication of any treatment effect was found in the experimental treatment 'non-delinquents.'

Each finding was discussed in light of the cognitive-

		Enti	re Sample	(n = 38)	• · · · · ·	
•			MMS*	101**	IQ_***	•
	MMS	•	1.00	.54	.41	
:	101	. 0		1.00	.73	
	102	•		•	1.00	
	•				•	
	•	'Deling	uent Sample	e (n = 18)		Ŷ
			MMS*	10 ₁ **	102***	
	MMS		1.00	.17	.13	
	101			1.00	.76	
	102	•	••••		1.00	
	•	'Non-deli	nquent' Sam	ple (n = 20))	
			MRIS*	10 ₁ **	102***	
	MMS		1.00	.53	.29	•
	IQ ₁			1.00	.53	•
	102				1.00	

PRODUCT-MOMENT CORRELATIONS BETWEEN MORAL MATURITY

TABLE 21

Verbal Intelligence Quotient. Non-verbal Intelligence Quotient. 10

10,

128

د



developmental theory and previous research. Questions were raised concerning the findings. Chapter V deals with these questions in more detail.

3. SUMMARY

This chapter reported the findings of the study. Descriptive

1. 'Delinquents' and 'non-delinquents' appeared to differ significantly on moral maturity scores.

2. The treatments apparently had no effect on the follow-up test moral maturity scores of 'delinquents' or 'non-delinquents.' The only slight indication of any change was in the scores of 'nondelinquents' in the experimental treatment situation.

3. Cognitive role-taking was seemingly not related to moral maturity scores and 'delinquents' and 'non-delinquents' did not differ on this role-taking ability.

4. Both verbal and non-verbal intelligence quotients correlated positively with moral maturity scores, however correlations between these variables were lower for the 'delinquent' group than for the 'non-delinquent' group.

Chapter V summarizes the study, discusses implications of the findings and concludes with suggestions for further research.

Chapter V

SUMMARY, IMPLICATIONS AND SUGGESTIONS FOR FURTHER RESEARCH

Introduction

In the last decade, the literature on Social Studies Education has emphasized student exploration of moral issues. In this regard, the cognitive-developmental theory of moral development, as formulated by Lawrence Kohlberg, has received a great deal of theoretical and empirical attention. Kohlberg and his associates have apparently demonstrated that, given certain conditions, moral reasoning relates to conduct, and that moral reasoning can be advanced through exposure to reasoning at a higher stage. Yet, the congruence between the stage of moral reasoning and conduct remains questionable, and it is with conduct, as well as with reasoning, that moral education must be concerned. Therefore, the major purpose of this study was to explore the relationship between moral reasoning and conduct, with a second concern relating to the effect of participation in moral discussions on moral reasoning. The study was regarded as exploratory because, as far as was known, it was unique in that it studied the access meaning of Grade Five and Big 'delinquants' is a public schede matches. Although studies had been carried cut on the material communication of the deliversents, the relationship a service and Within school delinguents contract while the constrained that a movie while the second as and the high second a boot allog as to the particulation between moral
judgments and action among Grade Five and Six children. Although previous research had studied the effects of exposure to higher stage moral reasoning on legal delinquents and 'normal' elementary school children, this study attempted to explore the effects of participation in moral discussions on 'delinquent' and 'non-delinquent' students. Results might then form a basis for discovering the relationship between conduct and the effects of participation in moral discussions. This might lead to possible hypotheses which might form the framework for future research.

This chapter begins with a summary of the research design procedures and findings of this study. Implications of the findings follow and the chapter concludes with suggestions for further research.

1. SUMMARY OF THE STUDY

This summary is divided into two parts: (1) a brief report on the research procedures, and (2) a summary of the findings as reported in Chapter IV.

SHI .

In some is applied the selectionship between moral reasoning and conduct, complete in these Simulation Elementary Schools, chosen by for Secondary Medicia Meneri Manual ware asked to select Grade Five and Six applieds when electropic event "delinquent" or 'non-delinquent' because. Medical mere appendix to the station to be behavior because. Medical mere appendix to the station of behavior

the of 'all mount' or 'son delingent' behavior.

These categories were included in the Student Behavior Rating Instrument (Appendix C). Also teachers rated each subject, who was identified on the Student Behavior Rating Instrument, on the Fairmindedness and Trustworthiness Instrument (Appendix D).

From a total of one hundred and fifty-seven students identified by teachers, twenty-three students were selected by the researcher as displaying the most extreme form of 'delinquency' in . that teachers had rated them in five or more of the SBRI categories, and twenty-nine were selected as displaying the most extreme form of 'non-delinquency' in that teachers had rated them in five or more of the SBRI categories. These selected 'delinquents' also rated low on the Fairmindedness and Trustworthiness Instrument, whereas 'nondelinquents' rated high. Forty subjects received parental permission to be involved in the study, but with some students becoming unavailable for study either prior to, or during, the research, the final sample consisted of eighteen 'delinguents' and twenty 'nondelingments.' Subjects ranged in age from ten years and five months to thirteen years and one month, with the average age being eleven years and five months. Within the total sample there were fifteen females, thirtness of whom were judged 'non-delinquent,' and twentytimes sains, only never of when were judged 'non-delinquent.' All subjects attended achools in low or low-middle class areas of Binonton.

All subjects were wiministered Form & of the Kohlberg Moral Judgment Instrument and Flevill's Bale-Tabing Instrument in an interwine situation. Constitut Announcementation Institutence quotients were obtained from the cumulative records of Grade Six subjects. Grade Five subjects were administered the <u>Canadian Lorge-Thorndike</u> <u>Intelligence Tests</u> in a group situation.

The 'delinquent' and 'non-delinquent' subjects were then randomly assigned to one of three groups: experimental treatment, placebo or control. The experimental treatment consisted of six, one-half hour sessions held over a six week period, in which moral dilemmas were discussed and the researcher attempted to argue for and against student responses to the dilemmas at the stage which was one higher to the initial student responses. Five sessions comprised the viewing of one of the dilemmas included in First Things-Values (Guidance Associates, 1972), and the last session consisted of a discussion on dilemmas which subjects felt were relevant to their own lives. The purpose of this treatment was to attempt to raise the students' level of moral reasoning. The placebo group, under the direction of the researcher, played Social Studies games during six, one-half hour sessions held once per week over a six week period. The purpose of the inclusion of this group in the research design was to attempt to control any possible researcher effect on the experimental group's postcest scores. The control group continued with normal classroom activities during the experimental and placebo group esseions. As far as can be ascertained, no control group subject was engaged in the type of activity practised by the experimental treatment group.

Of comparison of the six whit intervention, all subjects

A month later all subjects were administered Form A of the same instrument and were retested, for test-retest reliability purposes, on Flavell's Role-Taking Instrument.

Interjudge agreements on role-taking and moral reasoning scoring were obtained. The percentage of interjudge agreement between three judges, one of whom was the researcher, on the Flavell Role-Taking Instrument scores was over 90%. On modal stage scores derived from the Kohlberg Moral Judgment Instrument at the pre, post and follow-up test administrations, on a random sample of eighteen subjects, judges one and two had a percentage agreement of 83%, judges two and three had a 77% agreement, and judges one and three were in agreement in seventeen out of eighteen cases (94%).

2. Summary of the Findings

Question 1

Do Grade Five and Six 'delinquents' and 'nondelinquents' differ in the scores obtained on Kohlberg's Noral Judgment Instrument?

It was found that 'delinquents' and 'non-delinquents' differed significantly (p = .001) from each other in moral maturity scores at the pre, post and fellow-up test administrations of the Nohlberg instrument. 'Delinquents' tended to reason more in Stage 1 and 2 terms than did 'non-delinquents' unereas the latter group demonstrated more Stage 3 thinking then the 'delinquent' group.

Will a mittalion in which a random subsction of which

Five and Six 'delinquents' and 'non-delinquents' are exposed to moral reasoning at one stage above their pretest stage influence their reasoning to the extent that their stage of reasoning is higher on the posttest than it was on the pretest?

2A. Will any changes be reflected on a follow-up test administered one month after the posttest?
2B. Will there be any differences between pretest, posttest and follow-up scores on the Kohlberg Moral Judgment Instrument between 'delinquents' and 'non-delinquents' in the experimental, placebo and control groups?

It was decided that the posttest instrument (Form B of Kohlberg's Noral Judgmed Instrument) was not a satisfactory indicator of any change following treatment. The most recent scoring manual does not include two of the dilammas so problems in scoring were encountered. One issue (conscience) in the Earl and Bob dilemma (Instrument VII, Appendix E) was not scored owing to doubts by Rohlberg on the velidity of the scoring manual on this issue. Product-manient correlations indicated that the posttest did not competent well with the pre sid follow-up test, and analysis; will make stores suggested that the posttest difficulties may have also have due to the increased sophistication of the researcher in interviewing technique. Mersus is the pildt study for probing questions.

to asked, and this usually lad to very short responses, probing

questions were asked in the major study. This frequently meant longer and more complex res ficulties. Also, in Stage 1 and 2 Stage 4 respon found, were more owing to inadequi

A three-w

was no statistical moral maturity scor when compared to 'de treatments, made the La 'men-delinquent' expe score of 2(3),

Constitut 3

Bole Making Instrument related to stage of moral reasoning? For 'delinguents'? For 'non-delinguents'? Do "Melinguente' and 'nen-delinguents' differ on

Mile Meturity scores

subject increased in follow-up test moral maturity scores. Is role-taking ability as assessed by Flavell's

turn this created more scoring difudy, subjects had reasoned in the major study Stage 3 and some ted. These latter responses, it was to score than Stage 1 or 2 responses, the old scoring manual.

sis of variance test indicated that there gnificant treatment effect on follow-up test The only slight indication of any change was in the 'non-delinque experimental treatment group. This group, mints' in the same treatment, or when compared to either 'delinquents' er 'non-delinquents' in the placebo or control : Mean score change. Two subjects in the reatment group changed from a pretest follow-up test score of 3(2), and all but

correlations were obtained. This lack of relationship was attributed to the nature of the test itself. Whereas the Flavell instrument assessed the ability of 'putting oneself in the shoes of another person' and perceiving things from another's point of view, it does not assess the ability of mature reciprocal role-taking in which one can understand that others can take one's own perspective simultaneously with one's taking of others' perspectives. The former ability can be demonstrated in Stage 2 reasoning, however, the latter appears to be more related to a Stage 3 orientation. As the majority of subjects apparently demonstrated the ability to cognitively roletake, no significant difference was found between 'delinquents' and 'non-delinquents' on this measure.

Question 4

Is intelligence quotient as assessed by the <u>Canadian</u> <u>Lorge-Thorndike Intelligence Tests</u> related to stage of moral development? For 'delinquents'? For 'nondelinquents'?

The product-moment correlation between the entire sample's moral reasoning scores and verbal and non-verbal intelligence quotients was .54 and .41. This deconstrated a positive relationchip between sound maximing and intelligence quotients and this was maintained by the lang-deligence's same does 'delinquent' and 'nonstatement' score fact and this correlation between and intelligence the lange and the correlation between and intelligence the lange and intelligence quotient for the 'non'delinquent' group were lower, being .17 and .12 respectively.

General Summary of the Findings

The following findings seem to apply to this study:

1. 'Delinquents' and 'non-delinquents' appeared to differ significantly on moral reasoning scores.

2. The treatments apparently had no effect on follow-up test moral maturity scores of 'delinquents' or 'non-delinquents.' The only slight indication of any change between pre and follow-up test scores was in the 'non-delinquent' experimental treatment group.

3. Cognitive role-taking was seemingly not related to moral maturity scores and 'delinquents' and 'non-delinquents' did not differ in role-taking ability.

4. Both verbal and non-verbal intelligence quotients correlated positively with moral reasoning, however, correlations between these variables were lower for the 'delinquent' group than for the 'non-delinquent' group.

2. IMPLICATIONS

This section is divided into four sections: (1) Infligations for the cognitive developmental theory of moral development; (2) Implications for elapsroom minimum (3) Implications for Social Studies Execution; (4) Implications for Social Studies

This study was an employed on the tit was designed as an activity preliminary to hypothesis-tenting. As Karlinger (1973,

It is well to recognize, though, that there are activities preliminary to hypothesis-testing in scientific research.

He goes on to say that "research aimed at discovering, or uncovering relations, is indispensable to scientific advance in the social sciences (ibid.)." Therefore, remarks which follow are based upon the exploratory nature of this study, in that hipotheses are generated, and questions posed which may be fruitful in future research.

1. Implications for the Cognitive-Development Theory of Moral Development

The finding that 'delinquents' and 'non-delinquents' apparently differ on moral reasoning scores derived from the Kohlberg Moral Judgment Instrument, may have theoretical implications. Although studies on legal delinquents and adult inmates have been carried out (Kohlberg, 1958; Ewanyk, 1973), and findings indicated that the majority of these subjects reasoned in preconventional terms, this study suggests that children of the age group studied, who are deemed 'delinquent' by teachers, also tend to reason in preconventional terms, whereas those deemed 'non-delinquent' reason more in Stage 3 terms. This does not necessarily indicate a cause and effect relationship as there may be a variety of factors which affect behavior. These may include external, social situational factors, such as social punishment and reward, or internal dispositions, such as ego strength (Krebs and Kohlberg, 1973). Neither does this study suggest, even for this sample, that the relationship. between moral reasoning and conduct is firmly positive as 'delinquents'

were found who seemingly had some Stage 3 thinking as part of their repertoire, and several 'non-delinquents' were found who reasoned predominantly in Stage 2 terms. Yet with regard to the 'nondelinquent' group there appears to be some support for Kohlberg's view that by early preadolescence, children will have reached the conventional level of moral reasoning, and that they will reflect this level in behavior "by consistently showing a decent regard for the core expectations and approval of parents, peers and outside authorities (Kohlberg and Turiel, 1971, p. 460)."

The findings of this study would suggest that, even by early preadolescence, there are apparently very sociable children who reason in Stage 3 terms, and some who still reason within a Stage 2 orientation. This raises the question of whether these Stage 2 children rationalize their 'good' behavior in terms of instrumental. exchange (you be nice to me and I'll be nice to you), or whether some other variable, such as parental upbringing, has influenced their behavior. Conversely one can ask why there were three 'delinquents' who had some Stage 3 reasoning as part of their repertoire. Are there other factors, such as ego strength or external social forces, which apparently influence their 'delinquent' behavior?

Overall, however, this study indicated that 'delinquents' and 'non-delinquents' differ on moral reasoning scores. One might, therefore, hypothesize that, for children in this age range who display 'delinquent' or 'non-delinquent' characteristics, there is a relationship between moral reasoning and conduct.

Rabberry (2070, p. 204) states that meral development is

encouraged by "creating dissatisfaction about [the student's] present knowledge of the good." Apparently, if any dissatisfaction was generated in the 'delinquents' in the experimental treatment group, it did not seemingly lead to moral development. The tentative finding in this study was that only 'non-delinquents' were slightly affected by the experimental treatment. Why the experimental treatment, apparently, only affected 'non-delinquents' can only be speculated. Maybe, as Piaget (1932, p. 174) theorized, it is the practical morality which shapes the theoretical and the practical morality of 'non-delinquents,' according to teacher ratings, was in terms of co-operation and respect for rules and authorities, whereas for the 'delinquent' sample the opposite appeared true. One could ask the question as to whether 'delinquents' are intransigent, or fixated, in their moral development. Maybe the environment in which they live is not conducive to moral development. Their reality may consist of frequent punishment because of efforts to do what they want to do,

rather than what others would wish them to do.

In light of other research studies which apparently demonstrated that exposure to +1 moral reasoning led to posttest or postposttest gains, this study may suggest that it is easier to create movement in some children because their conduct is already in accord with higher stage reasoning, whereas it is more difficult with other children because their conduct is not congruent with higher stage their stage.

The finding that intelligence quotients were apparently related to morel responing scores was in keeping with other studies

(Kohlberg, 1968; Lydiatt, 1973). Of possible note in this study was that, although there was a significant relationship between intelligence quotients and moral reasoning scores over the entire sample, when 'delinquents' and 'non-delinquents' were separated for analysis purposes, correlation coefficients for the former group were only slightly positive. It would appear that 'delinquents' could have high intelligence quotients and yet be quite low on moral reasoning. If intelligence is a necessary but not sufficient condition for moral development, the overall correlation indicated in this study would seemingly support this claim. Yet the correlations indicated in the 'delinquent' group may suggest that there are other factors which impinge on this relationship, one of which, conduct, has already been discussed.

The indication that role-taking ability might not be related to moral reasoning stages is probably not surprising in light of the instrument used. The only implication that it might be possible to draw from this study is that the Flavell cognitive role-taking instrument does not have discriminative value in differentiating between the preconventional and conventional stages of moral development. Moreas on the pilot study this test apparently differentiated between subjects at Stages 1 and 2, the major study involved subjects who reasoned mainly in Stage 2 and 3 terms. According to the cognitive-developmental theory, a test which assesses mature reciprocal role-taking is necessary to differentiate between these two stages. The Flavell instrument, however, does not assess mature reciprocal role-taking is that it does not assess mature

perceive that others can take one's own perspective simultaneously with one's taking of others' perspectives.

2. Implications for Classroom Management

The indication that 'delinquents' and 'non-delinquents' differ in moral reasoning ability may have implications for the ways in which teachers can help 'delinquent' children. Kohlberg (1973d,

p. 179), when writing about cheating behavior, says:

The effort to force a child to agree that an act of cheating is very bad when he does not really believe it will only encourage morally immature tendencies toward expedient outward compliance. In contrast a more difficult but more valid approach involves getting the child to examine the pros and cons of his conduct in his own terms (as well as introducing more developmentally advanced considerations).

This would suggest that teachers should attempt to reason with a student at the student's own stage of moral reasoning or at the stage one above, and not at a level which is developmentally removed. This, as Kohlberg points out, is not easy. Even if a child who is causing problems, is reasoned with and accepts the reasons offered, this may not necessarily mean that the child will act on them, for the child may be dedicient in an ego control, such as the will to act on 'good' reasons. Yet, a teacher could attempt to reason at the child's level. For example, a student with a Stage 2 orientation could be reasoned with in his own terms by informing him that 'good' behavior will lead to activities which he enjoys, or he could be reasoned with in Stage 3 terms by encouraging him to take the perspective of another person in a conflict situation. In this way

not only may behavior be based more on reason, but moral development

may also be stimulated. Whereas it is realized that some classroom order must be maintained if learning is to occur, rigid authoritarianism may lead to compliance, but it may not help moral development.

A classroom in which appraisive and prescriptive (what should be, and what should be done) inquiry is encouraged, and rules are based on reasons which students comprehend, may be quite effective in stimulating moral development. In fact, Lieberman and Selman (1974) found that the class, in which the teacher encouraged moral discussions, made greater gains in moral maturity scores over a one year period, than did a class which was involved in discussions led by an outside 'expert.'

This study indicates that upper elementary school students reason mainly in Stages 2 and 3. In terms of classroom management this may imply that reasoning used by teachers, in order to help 'behavior problem' children, should be couched in Stage 2, 3 or 4 terms.

3. Implications for Social Studies Education

The Alberta Social Studies program, <u>Experiences in Decision</u> <u>Making</u> (Alberta Department of Education, 1971) suggests that elementary school children should examine social issues. These issues, as Jeffrey (1968, p. 40) points out, are "fundamental moral problems." The question can be raised as to shether elementary school children are sognitively and morally (as the two appear to be linked) nature shough to emplies certain issues. Contain publicas may be too abstract for elementary school children. But, <u>Emplicances in Decision</u>

Anterior of Alexandras, 1998, S. 2017 appoints that

children should examine the concept of justice in terms of:

. reasoned consideration for others . . . fair play, security, what ought to be, impartiality, equality, reasonableness, legitimacy, rightfulness One wonders how a child who reasons in terms of instrumental relativism (Stage 2) could comprehend this 'adult notion' of justice. One might also wonder how children with a Stage 3 roletaking perspective, which does not include a generalized 'system' perspective, could consider the question of their responsibilities as Canadian members of the international community (Alberta Department, of Education, 1971, p. v-3), except in terms of their own viewpoint. There are other references in the Alberta program where children are expected to have the ability to role-take. For example, one objective in a Grade Three unit states: "Students should attempt to understand the perspective of the contemporary Eskimo . . . (ibid., p. 111-113)." Yet in this research study there were two subjects who seemingly could not cognitively role-take. It would appear that the Alberta Social Studies program has not taken into account the role-taking abilities of elementary school children.

Experiences in Decision Making (ibid.) also emphasizes the Raths valuing model. There is little stress in this values clarification approach on moral reasoning. Yet this research study suggests that reasoning abilities differ smoog Grade Five and Six students, a point which is apparently neglected by the Alberta Depastment of Education.

This research may imply that Social Studies programs already in existence be exemined in light of the Mohlberry theory and

specifically in light of the moral reasoning demonstrated by children. It may also imply that program developers should be cognizant of children's moral reasoning so that new programs are more closely matched to the child's moral reasoning abilities. Specifically, programs should be introduced which match the stages of reasoning, displayed by elementary school children, to both the content and process of instructional programs. For example, social issues which are examined should be within the child's world view. It would appear ineffectual to ask a child to create solutions to problems, such as the world energy crisis, with which he has had no, or little, experience. It would also appear that the Grade One to Six sequence of home, to community, to province, to country, to world (the expanding horizons theory) may have to be re-examined in light of the roletaking and social perspective taking abilities of elementary school children. A country or world view would imply the necessity of a generalized 'system' perspective. This perspective, according to the cognitive-developmental theory, is not attained until Stage 4 reasoning is reached, and this study indicated that students at the Grade Five and Six levels demonstrate very little, if any, Stage 4 reasoning.

This study, however, does not imply that all Social Studies teachers must fully comprehend the Mohlbery theory, and be able to interview students on moral dilemens and then score the responses. This remearcher found it cost meanly two pears to come to his understanding of the theory, and his skillty to imperview students and

The second of second the second

of the theory and can probably be trained to implement moral discussions and other value-clarifying activities in their classrooms. A danger is that, if teachers become cognizant of the theory, they may label students for the wrong reasons; i.e., the label becomes an evaluative stereotype. The reason for understanding the theory is not to praise or blame but to understand the child better so that moral development can be stimulated.

This study indicated that students, in the main, enjoyed the experimental treatment and were not hesitant in bringing forth their reasons and arguments. This would seem to indicate that this procedure should be capitalized on, and that activities which stimulate moral development should be incorporated in the Social Studies curriculum. However, it is this writer's opinion, that moral discussions should be supplemented by inquiry procedures in which pertinant and representative data is examined and analysed.

Implications for Curriculum

Horal education has recently received attention in educational literature, especially with regard to Kohlberg's theory. One of the instructional implications of this theory, as has already been indicated, requires that students be exposed to moral reasoning at a stage higher than their own stage. Yet as has been pointed out by Penton (1974), the type of experimental treatment carried out in this research, involving, as it did, only six, half-hour sessions, may not be very effective in stimulating the moral development of some students. Despite a previous research study which indicated that three stabilions were maningly effective in reising moral maturity scores (Turiel, 1966), it would appear in this particular study that six one-half hour sessions were not effective, especially with 'delinquent' children. As Beck (1971) points out, moral development does not necessarily occur in isolation from the rest of the school curriculum. If moral education is to be one of the major objectives of the school, then the curriculum, and practice of that curriculum, have to be viewed as a vehicle, and the curriculum, to be effective, needs to relate to the needs, abilities and interests of students. This study has indicated that two or three stages of moral development might be in operation in Grade Five and Six students. If a curriculum is to be resonant with the moral development stages of children, then this finding may need to be taken into account.

This would imply, firstly, that the school be operated on principles of justice and fairness; a school in which the rules and regulations are seen, by children, to be reasonable. Secondly, existing instructional programs often include value related issues, and these should be capitalized upon. For example, in children's stories there are often conflicts involving people; these should be discussed as moral issues. Thirdly, the classroom climate should be conducive to moral development in that children are encouraged to reason about moral issues and to act in morally mature ways.

3. SUGGESTICUS FOR FURMER RESEARCH

The findings of this study and the questions posed in light of these findings suggest the need for further research studies.

e would includer

1. The formal testing of the hypothesis that moral reasoning is related to the conduct of 'delinquent' and 'non-delinquent' school children.

2. A study which explores the effect of student participation in moral discussions on future student conduct.

3. An examination of the relationship between stages of moral reasoning and the ways in which a student investigates social problems in Social Studies classes.

4. An investigation of the influence of exposing childrento moral discussions which attempts to ascertain the variables which stimulate moral development.

5. A further examination of the relationships between inteligence, role-taking ability and moral reasoning of 'delinquent' children.

6. An examination of the effect of various instructional procedures, such as problem-solving, value clarification activities and role-playing, on the moral development of children.

7. An examination of the effect of reasoning, with 'delinquent' children, at the child's own stage of reasoning and at the stage which is one above the child's stage.

8. An investigation into children's perceptions of, and attitudes towards, classroom or school rules.

9. A study which investigates the effects of learning about, and using the Kohlberg theory, on teacher classroom behavior. BIBLIOGRAPHY

BIBLIOGRAPHY

- Alberta. Department of Education. Experiences in decision making. Edmonton: Department of Education, 1971.
- Association for Values Education and Research. <u>Interim report of a</u> <u>study in moral education at Surrey, British Columbia</u>. Vancouver: University of British Columbia, Association for Values Education and Research, no date.
- Association for Values Education and Research. <u>Moral education: A</u> <u>bibliography</u>. Vancouver: University of British Columbia, Association for Values Education and Research, 1972.
- Bandura, A., & McDonald, F. The influence of social reinforcement and the behavior of models in shaping children's moral judgment. Journal of Abnormal and Social Psychology, 1963, 67, 274-281.
- Beck, C. <u>Moral education in the schools</u>. Toronto: Ontario Institute for Studies in Education, 1971.
 - Beck, C., Sullivan, E., & Taylor, N. Stimulating transition to postconventional morality: The Pickering High School study. Interchange, 1972, 3, 28-37.
- Beller, E. Children's attitudes towards honesty: Two attitude components in younger boys. Journal of Social Psychology, 1949, 29, 137-151.
- Black, A. Co-ordination of logical and moral reasoning in adolescence. Abstract of paper presented at Western Psychological Association, 1973.
- Blatt, M., & Kohlberg, L. The effects of classroom moral discussion upon ghildren's level of moral judgment. In L. Kohlberg, <u>Collected papers on moral development and moral education</u>. Cambridge, Massachusetts: Harvard University, Center for Moral Education, 1973.
- Boehm, L. The development of conscience: A comparison of American children of different mental and socio-economic levels. <u>Child</u> Development, 1962, 33, 575-590.
- Brennan, W. Belation of monial adaptation, emotional adjustment and moral judgment to intelligence in primary school children. British Journal of Educational Psychology, 1962, 32, 200-204.
- Brogien, H. A Captor analysis of forty character tests. Psychological Manadomates 1940, 20, 3.

- Burton, R. The generality of honesty reconsidered. <u>Psychological</u> <u>Review</u>, 1963, 70, 481-499.
- Center for Moral Education. Form A-Standard scoring manual. Cambridge, Massachusetts: Harvard University, Center for Moral Education, 1973. (a)
- Center for Moral Education. Form B-Standard scoring manual. Cambridge, Massachusetts: Harvard University, Center for Moral Education, 1973. (b)
- Coleman, J. <u>Abnormal psychology and modern life</u>. Glenview: Scott Foresman, 1964.
- Cowan, P., Langer, J., Heavenrich, J., & Nathanson, M. Social learning and Piaget's cognitive theory of moral development. Journal of Personality and Social Psychology, 1969, 2, 261-274.
- Craig, R. Lawrence Kohlberg and moral development: Some reflections. Educational Theory, 1974, 24, 121-129.
- Dewey, J. Moral principles in education. New York: Philosophical Library, 1959.

Dewey, J. Democracy and education. New York: Free Press, 1966.

- DOFr, D., & Fey, S. Relative power of symbolic adult and peer models in the modifications of children's moral choice behavior. Journal of Personality and Social Psychology, 1974, 29, 335-341.
- Edwards, J. Some studies of the moral development of children. Educational Research, 1965, 7, 200-211.
- Ewanyk, D. Disequilibrium as a source of inducing higher moral reasoning in delinquents. Unpublished master's thesis, University of Alberta, 1973.
- Fenton, E., Colby, A., & Speicher-Dubin, B. Developing moral dilemmas for Social Studies classes. Unpublished manuscript. Harvard University, Center for Moral Education, 1974.
- Flavell, J., Botkin, P., Fry, C., Wright, J., & Jarvis, P. The development of role taking and communication skills in children. New York: Wiley, 1968.
- Fodor, S. Delinquency and susceptability to social influence among adolescents as a function of level of moral development. Journal of Social Psychology; 1972, 86, 257-260.
- Freenkel, J. Belping students think and value. Englewood Cliffs, New Jersey: Prentice-Hall, 1973.

Frankena, W. Ethics. Englewood Cliffs, New Jersey: Prentice Hall, 1963.

Good, C. (Ed.). Dictionary of education. New York: McGraw-Hill, 1973.

Gough, H., & Peterson, D. The identification and measurement of predispositional factors in crime and delinquency. Journal of Consulting Psychology, 1952, 16, 207-212.

Graham, D. Moral learning and development. London: Batsford, 1972.

- Grim, P., Kohlberg, L., & White, S. Some relationships between conscience and attentional processes. Journal of Personality and Social Psychology, 1968, 8, 239-252.
- Guidance Associates. First things: Values. Pleasantville, New York: Guidance Associates, 1972.
- Haan, N., Smith, M., & Block, J. Moral reasoning of young adults. Political-social behavior, family background and personality correlates. Journal of Personality and Social Psychology, 1968, 10, 183-201.
- Hague, W. The gospel according to Kohlberg. <u>Religious Studies and</u> <u>Moral Education Council of the Alberta Teachers' Association</u> <u>Newsletter</u>, 1974, <u>1</u>, 16-18.

Hare, R. The language of morals. Oxford: Clarendon Press, 1952.

- Harris, H. Development of moral attitudes in white and negro boys. Developmental Psychology, 1970, 2, 376-383.
- Hartshorne, H., & May, M. Studies in the nature of character. Vol. 1. Studies in deceit. New York: Macmillan, 1930.
- High School Geography Project. The game of farming. In <u>Geography</u> in an urban age, part II. London: Macmillan, 1969.
- Hudgins, W., & Prentice, N. Moral judgments in delinquent and nondelinquent adolescents and their mothers. Journal of Abnormal <u>Psychology</u>, 1973, 82, 145-152.
- Irwin, D., E Ambron, S. Moral judgment and role taking in children ages three to seven. Paper presented at biennial meeting of The Society for Research in Child Development, Philadephia, 1973. [ERIC ED084033]

Jantz, R. An investigation of the relationship between moral development and intellectual development in male elementary school students. Theory and Research in Social Education, 1973, 1, 75-81.

- Jeffreys, M. <u>Personal values in the modern world</u>. Harmondsworth: Penguin, 1968.
- Johnson, R. A study of children's moral judgments. <u>Child Develop-</u> ment, 1962, 33, 327-354.
- Joyce, B. <u>New strategies for social education</u>. Chicago: Science Research Associates, 1972.
- Kay, W. Moral development. London: George Allen and Unwin, 1970.
- Kellmer-Pringle, M., & Edwards, J. Some moral concepts and judgments of junior school children. British Journal of Social and Clinical Psychology, 1964, 3, 196-215.
- Kerlinger, F. Foundations of behavioral research (2nd ed.). New York: Holt, Rinehart and Winston, 1973.
- Kirman, J. <u>Canada's prairie wheat game</u>. Edmonton, Alberta: Author, no date.
- Kohlberg, L. The development of modes of moral thinking and choice in the years ten to sixteen. Unpublished doctoral dissertation, University of Chicago, 1958. [University of Chicago Microfilm No. 4397.]
- Kohlberg, L. Moral development and identification. In H. Stevenson (Ed.), <u>Child psychology</u>. The Sixty-second Yearbook of the National Society for the Study of Education. Chicago: University of Chicago Press, 1963.
- Kohlberg, L. Development of moral character and moral ideology. In M. Hoffman & L. Hoffman (Eds.), <u>Review of child development</u> research. Volume One. New York: Russell Sage Foundation, 1964.
- Kohlberg, L. Moral development. <u>International Encyclopaedia of the</u> Social Sciences. <u>Macmillan</u>, 1968.
- Kohlberg, L. Stage and sequence: The cognitive-developmental approach to socialization. In D. Goslin (Ed.), <u>Handbook of</u> <u>socialization theory and research</u>. New York: Rand McNally, 1969.

Kohlberg, L. Education for justice: A modern statement of the Platonic view. In N. Sizer & T. Sizer (Eds.), <u>Moral education</u>: <u>Five lectures</u>. Cambridge, Massachusetts: Harvard University Press, 1970.

Kohlberg, L. Brom is to ought. How to commit the naturalistic fallacy and get away with it in the study of moral development. In T. Mischel (Ed.), <u>Cognitive development and epistemplogy</u>. New York: Agademic Press, 1971. (a)

- Kohlberg, L. Stages in moral development as a basis for moral education. In C. Beck, B. Crittenden & E. Sullivan (Eds.), Moral <u>education: Interdisciplinary approaches</u>. Toronto: University of Toronto Press, 1971. (b)
- Kohlberg, L. Moral education in the schools: A developmental view. In D. Purpel & M. Belanger (Eds.), <u>Curriculum and the cultural</u> <u>revolution</u>. Berkeley, California: McCutchan, 1972.
- Kohlberg, L. Continuities and discontinuities in childhood and adult moral development revisited. In L. Kohlberg, <u>Collected</u> <u>papers on moral development and moral education</u>. Cambridge, <u>Massachusetts: Harvard University</u>, Center for Moral Education, 1973. (a)
- Kohlberg, L. Stages and aging in moral development—some speculations. The Gerontologist, 1973, 13, 497-502. (b)
- Kohlberg, L. The development of children's orientations towards a moral order: Sequence in the development of moral thought. In
 E. Lord & C. Bailey (Eds.), A reader in religious and moral education. London: SCM, 1973. (c)
- Kohlberg, L. The moral atmosphere of the school. In L. Kohlberg, <u>Collected papers on moral development and moral education</u>. Cambridge, Massachusetts: Harvard University, Center for Moral Education, 1973. (d)
- Kohlberg, L. Moral stage scoring manual. Part I, Introduction to interviewing and scoring. Part II, Tables. Part III, Summary and description of standard scoring. Cambridge, Massachusetts: Harvard University, Center for Moral Education, 1975.
- Kohlberg, L., Kauffman, K., Scharf, P., & Hickey, J. A just community approach to corrections: A manual, part II. Cambridge, Massachusetts: Harvard University, Moral Education Research Foundation, 1974.
- Kohlberg, L., & Kramer, R. Continuities and discontinuities in childhood and adult moral development. <u>Human Development</u>, 1969, <u>12</u>, 93-120.
- Kohlberg, L., LaCrosse, R., & Ricks, D. The predictability of adult mental health from childhood behavior. In B. Wolman (Ed.), <u>Handbook of child psychopathology</u>. New York: McGraw-Hill, 1971.
- Kohlberg, L., & Mayer, R. Development as the aim of education. Hervard Educational Review, 1972, 42, 449-495.
- Bablery, L., Scharf, P., & Mickey, J. The justice structure of the primes: A theory and an intervention. <u>The Prison Journal</u>, 1972, <u>51</u>, 3-14.

Kohlberg, L., & Selman, R. Preparing school personal relative to values. A look at moral education in the schools. National Centre for Educational Communication, 1972. [ERIC ED058153]

٤.,

- Kohlberg, L., & Turiel, E. Moral development and moral education. In G. Lesser (Ed.), <u>Psychology and educational practice</u>. Chicago: Scott Foresman, 1971.
- Krebs, R., & Kohlberg, L. Moral judgment and ego controls as determinants to resistance to cheating. Unpublished manuscript, Harvard University, Center for Moral Education, 1973.
- Krupa, W. The land use game. Middletown, Connecticut: Education Ventures, 1971.
- Kugelmass, S., & Breznitz, S. The development of intentionality in moral judgment in city and kibbutz children. Journal of Genetic Psychology, 1967, <u>111</u>, 103-111.
- Kurtines, W., & Grief, E. The development of moral thought: Review and evaluation of Kohlberg's approach. <u>Psychological Bulletin</u>, 1974, 81, 453-470.
- Lee, L. The concomitant development of cognitive and moral modes of thought. A test of selected deductions from Piaget's theory. Genetic Psychology Monograph, 1971, 83, 93-146.
- Leming, J. An empirical examination of key assumptions underlying the Kohlberg rationale for moral education. Paper presented at the annual meeting of the American Education Research Association, Chicago, 1974. [ERIC ED093749]

Lester Smith, W. Education. Harmondsworth: Penguin, 1964.

Lesuring Communication. Trading post. San Francisco: Lesuring Communication, no date.

Lieberman, M., & Selman, R. An evaluation of a cognitive-developmental values curriculum for primary grade children. Paper presented at annual meeting of the American Educational Research Association, Chicago, 1974.

Lorge, I., Thorndike, R., Hagan, E., & Wright, E. <u>Canadian Lorge-</u> Thorndike Intelligence Tests. Toronto: Nelson, 1967.

Loughran, R. 'A pattern of development in moral judgements made by adolescents, derived from Piaget's scheme of its development in childhood. Educational Deview, 1967, 19, 79-98.

Lydiat, M. The development of moral judgments in children. <u>Journal</u> of Moral Education, 1973, <u>3</u>, 367-377. MacRae, D. A test of Piaget's theories of moral development. Journal of Abnormal and Social Psychology, 1954, 49, 14-18.

May, P. Moral education in the school. London: Methuen, 1971.

- McCann, D., & Prentice, N. The facilitating effects of money and moral set on the moral judgment of adolescent delinquents. Journal of Abnormal Psychology, 1973, 82, 81-84.
- McKechnie, R. Between Piaget's stages: A study in moral development. British Journal of Educational Psychology, 1971, <u>41</u>, 213-217.
- Milgram, S. Behavioral study of obedience. Journal of Abnormal Social Psychology, 1963, 67, 371-378.

Morris, J. The development of adolescent value judgments. British Journal of Educational Research, 1958, 28, 1-14.

- Newman, F., & Oliver, D. <u>Clarifying public controversy</u>. Boston: Little, Brown, 1970.
- Oliver, D., & Shaver, J. <u>Teaching public issues in the High School</u>. Boston: Houghton Mifflin, 1966.
- Peters, R. Athics and education. London: Allen and Unwin, 1966.
- Phillips, J. The origins of intellect: Piaget's theory. San Francisco, W. H. Freeman, 1969.
- Piaget, J. The moral judgment of the child. London: Routledge and Kegan Paul, 1932.
- Piaget, J. The general problems of the psycho-biological development of children. In J. Tanner & B. Inhelder (Eds.), <u>Discussions on</u> <u>child development: Proceedings of the World Health Organization</u> <u>study group on the psychobiological development of the child.</u> Vol. IV. New York; International Universities Press, 1960.
- Porter, N. Kohlberg and moral development. Journal of Moral Education, 1972, 2, 123-128.
- Porter, N., & Taylor, N. How to assess the moral reasoning of students. Toronto: The Ontario Institute for Studies in Education, 1972.
- Porteus, B., & Johnson, R. Children's responses to two measures of conscience development and their relation to sociometric nomination. <u>Child Development</u>, 1965, <u>36</u>, 703-711.
- Raths, L., Harmin, M., & Simon, S. Values and teaching. Columbus, Ohio: Charles E. Herrill, 1966.
- Rawls, J. <u>A theory of justice</u>. Cambridge, Massachusetts: Harvard University Press, 1971.

- Rest, J. The hierarchical nature of moral judgment. A study of patterns of comprehension and preference of moral stages. Journal of Personality, 1973, <u>41</u>, 86-109.
- Rest, J., Turiel, E., & Kohlberg, L. Relations between level of moral judgment and preference and comprehension of the moral judgment of others. Journal of Personality, 1969, <u>37</u>, 225-252.
- Rouse, W. (Trans.). Great dialogues of Plato. Toronto: Mentor, 1963.
- Schwartz, S., Feldman, K., Brown, M., & Heingartner, A. Some personality correlates of conduct in two situations of moral conflict. Journal of Personality, 1969, <u>37</u>, 41-57.
- Selman, R. The relation of role taking to the development of moral judgment in children. <u>Child Development</u>, 1971, <u>42</u>, 79-91.
- Selman, R., Gordon, A., & Damon, W. The relation between stages of social role-taking and stages of justice conception in children ages four to ten. Paper presented at the Eastern Psychological Association, Washington, D.C., May, 1973.

Simon, F. <u>A reconstructive approach to problem-solving in the Social</u> <u>Studies</u>. Calgary, Alberta: Author, 1970.

Snow, C. <u>Two cultures and a second look</u>. Cambridge: Cambridge University Press, 1962.

Stuart, R. Decentration in the development of children's concepts of moral and causal judgment. Journal of Genetic Psychology, 1967, 3, 59-68.

- Sullivan, E., & Betk, C. A developmental approach to assessment of programs. Journal of Moral Education, 1974, 4, 61-66.
- Thorndike, R., & Hagan, E. <u>Measurement and evaluation in psychology</u> and education (3rd ed.). New York: Wiley, 1969.
- Tomlinson-Keasey, C., & Keasey, B. Formal operations and moral development: What is the relationship? <u>Proceedings of the</u> <u>Annual Convention of the American Psychological Association</u>, 1972, 7, 113-114.
- Tracy, J., & Cross, H. Antecedents of shift of moral judgment. Journal of Personality and Social Psychology, 1973, 26, 238-244.

Turiel, E. An experimental test of the sequentiality of developmental stages in the child's morel judgments. Journal of <u>Personality and Social Psychology</u>, 1966, <u>3</u>, 611-618. Turiel, E. Developmental processes in the child's moral thinking. In P. Mussen, J. Langer & M. Covington (Eds.), <u>New directions</u> <u>in developmental psychology</u>. New York: Holt, Rinehart and Winston, 1969.

Ł

Turiel, E., & Rothman, G. The influence of reasoning on behavioral choices at different stages of moral development. <u>Child</u> <u>Development</u>, 1972, 43, 741-756.

Warnock, G. Contemporary moral philosophy. London: Macmillan, 1967.

Whiteman, P., & Kosier, K. Development of children's moralistic judgments-age, sex, I.Q. and certain personal-experiential variables. Child Development, 1964, 35, 843-850.

Williams, N., & Williams, S. <u>The moral development of children</u>. London: Macmillan, 1970.

Wilson, J., Williams, N., & Sugarman, B. Introduction to moral education. Harmondsworth: Penguin, 1969.

Wright, D. The psychology of moral behaviour. Harmondsworth: Penguin, 1971.





DEFINITION OF MORAL STAGES (Kohlberg, 1971b, pp. 86-88)

I. Preconventional level.

At this level the child is responsive to cultural rules and labels of good and bad, right or wrong, but interprets these labels in terms of either the physical or the hedonistic consequences of action (punishment, reward, exchange of favors) or in terms of the physical power of those who enunciate the rules and labels. The level is divided into the following two stages:

Stage 1: The punishment and obedience orientation. The physical consequences of action determine its goodness or badness regardless of the human meaning or value of these consequences. Avoidance of punishment and unquestioning deference to power are valued in their own right, not in terms of respect for an underlying moral order supported by punishment and authority (the latter being Stage 4).

Stage 2: The instrumental relativist orientation. Right action consists of that which instrumentally satisfies one's own needs and occasionally the needs of others. Human relations are viewed in terms like those of the market place. Elements of fairness, of reciprocity and equal sharing are present, but they are always interpreted in a physical pragmatic way. Reciprocity is a matter of "you scratch my back and I'll scratch yours," not of loyalty, gratitude or justice.

II. Conventional level.

uite.

At this level, maintaining the expectations of the individual's family, group, or nation is perceived as valuable in its own right, regardless of immediate and obvious consequences. The attitude is not only one of <u>conformity</u> to personal expectations and social order, but of loyalty to it, of <u>actively</u> maintaining, supporting, and justifying the order and of identifying with the persons or group involved in it. At this level, there are the following two stages:

Stage 3: The interparsonal concordance or "good boy-nice girl" orientation. Good behaviour is that which pleases or helps others and is approved by them. There is much conformity to stereotypical images of what is majority or "natural" behaviour; Behaviour is frequently judged by intentions-"he means well" becomes important for the first time. One earns approval by being "nice."

Stage 4: The "law and order" origination. There is orientation toward anthority, fixed rales, and the definitenance of the social order. Right behaviour consists of doing one's duty, showing respect for authority and maintaining the given social order for it's own

III. Post-Conventional, Autonomous, or Principled Level

🕸 _ 5. t

At this level, there is a clear effort to define moral values and principles which have validity and application apart from the authority of the groups or persons holding these principles and apart from the individual's own identification with these groups. This level again has two stages:

Stage 5: The social-contact legalistic orientation. Generally with utilitarian overtones. Right action tends to be defined in terms of general individual rights and in terms of standards which have been critically examined and agreed upon by the whole society. There is a clear awareness of the relativism of personal values and opinions and a corresponding emphasis upon procedural rules for reaching consensus. Aside from what is constitutionally and democratically agreed upon, the right is a matter of personal "values" and "opinion." The result is an emphasis upon the "legal point of view," but with an emphasis upon the possibility of changing law in terms of rational considerations of social utility, (rather than freezing it in terms of Stage 4 "law and order"). Outside the legal realm, free contract is the binding element of obligation. This is the "official" morality of the American government and Constitution.

Stage 6: The miversal ethical principle orientation. Right is defined by the decision of conscience in accord with self-chosen <u>ethical principles</u> appealing to logical comprehensiveness, universality, and consistency. These principles are abstract and ethical, (the Golden Rule, the categorical imperative) they are not concrete moral rules like the Ten Commandments. At heart, these are universal principles of justice of the reciprocity and equality of the human rights and of respect for the dignity of human beings as individual persons.



THE TEN MORAL ISSUES (Kohlberg, Part II, 1975, p. 1)

- 1. <u>Punishment and Blame</u>. Should someone be punished or not? What is fair punishment?
- 2. <u>Property</u>. Should someone give, take or exchange property? What are property rights?
- 3. <u>Affiliation Roles</u>. Should someone help another or maintain the other's expectations in a personal relationship? What are the motives and obligations of a good family member or friend?
- 4. Law. Should someone obey or maintain the law? What are the characteristics of a good law?
- 5. Life. Should someone save a life or not? What makes life valuable?

7.

- 6. <u>Truth</u>. Should someone tell the truth or allow the truth to be disclosed or not? What defines truth-telling and why is it valuable?
 - Governance. Should someone obey or accept the authority of another person or of a government or rule-making group? What are the characteristics of a good governor and a good citizen?
 - <u>Civil Rights and Social Justice</u>. Should someone violate or uphold the political, economic, and social rights of another person or group? What are the basic political, economic and social rights?
- . Sex. Should one have a sexual relationship or not? What is the nature of a good erotic relationship and why is it valuable?
- 10. Morality and Mores. Should one follow one's moral opinion or conscience when it conflicts with law, love or self-interest? What is the nature of morality and what is the basis of its validity?


or names of students who consistently display any of the following behaviors. should stand out from the rest of the student population in their display of Students named Write the name behaviors.

Directions:

For each student you have identified on this instrument please complete the Fairmindedness and Trustworthiness Scale.

	Displays	Displays	
	Behavior	Behavior	
1. Pushes, hits, trips or otherwise			Doesn't hit, trip or otherwise
interferes with others physically.			interfere with others physically.
These actions usually lead to			•
complaints from others:			
2. Doesn't help other students.			Helps other students. Loans
Doesn't loan materials, help			materials, helps others in their
others in their work or other-			work and cooperates with others
vise cooperate with others in			in either work or play.
either work or play.			
3. Interferes with others'			Doesn't interfere with others'
possessions. Takes them or	•		possessions. Doesn't take them
disturbs them thereby		· · ·	or disturb them thereby prevent-
preventing others from using	, , ,		ing others from using them.
them. These actions usually	Ċ.		•
lead to complaints from others.		-	
•			

THE STUDENT BEHAVIOR RATING INSTRUMENT



Appendix C (Contd.)



Appendix C (Contd.)

J



No Sense of	Fairmindedness	that he Doesn't seem to only know his own rights about - or the rights of or turn.	Untrustworthy	oe Always trying to upon get away with rules something. e is check.	τ°
•	Selfish	Does what he wants, only thinks about rights when his share or turn.	Unreliable	Can only be depended upon to follow rules if someone is around to check	
	Unconcerned	Usually knows what the fair thing is but doesn't care too much about others' rights or his own.	Conforming	Follows rules mostly in order to keep out of trouble and/or to win approval.	
	Generally Fair	Tends to share and respect others' rights and stick up for his own succept when encorpt when encorpt when encorpt or excited or wants com- thing hedly.		Usually does' what's right even when there is no one around to check on him.	
	Vary Fair		Completely 2mitteorthy	Can always be depended on to do whatever he knows right regardless of what he wants or desires.	

1

THE FAIRMINDEDNESS AND TRUSTWORTHINESS INSTRUMENT

 χ^{2}

•



THE KOHLBERG MORAL JUDGMENT INSTRUMENT

Form A

Instrument I

In Europe, a woman was near death from a specific kind of cancer. There was one drug that the doctors thought might save her. It was a form of radium that a druggist in the same town had recently discovered. The drug was expensive to make, but the druggist was charging ten times what the drug cost him to make. He paid \$200 for the radium and charged \$2,000 for a small dose of the drug. The sick woman's husband, Heinz, went to everyone he knew to borrow the money, but he could only get together about \$1,000 which is half of what it cost. He told the druggist that his wife was dying, and asked him to sell it cheaper or let him pay later. But the druggist said, "No, I discovered the drug and I'm going to make money from it." So Heinz got desperate and broke into the man's store to steal the drug for his wife.

1.1 Should Heinz have done that? Was it actually right or wrong? Why?

- 1.2 Is it a husband's duty to steal the drug for his wife if he can get it in no other way? Would a good husband do it? Why?
- 1.3 Did the druggist have the right to charge that much when there was po law actually setting a limit on the price? Why?

If the subject thought Heinz should steal the drug the following questions were asked:

- 1.4(a) If the husband did not love his wife should he steal the drug? Why?
- 1.5(a) Suppose it wasn't Heinz's wife who was dying but it was Heinz's best friend. His friend didn't have any money and there was no one in his family who was willing to steal the drug. Should Heinz steal the drug for his friend? Why?

If the subject thought Heinz should not steal the drug the following questions were asked:

1.4 (b) Mould you steal the drug to save your wife's life? Why?

1.5(b) If you were dying but were strong enough would you steal the . drug to save your own life? Why?

1.6 Beins broke into the store and stoke the drug for his wife but he was caught by the paline. He was brought to court. Should the judge and heise to jail or should be let his go free?

Instrument II

Joe a fourteen-year old boy wanted to go to camp very much. His father promised him he could go if he saved up the money for it himself. So Joe worked hard at his paper route and saved up the \$40 it cost to go to camp and a little more besides. But just before camp was going to start, his father changed his mind. Some of his friends decided to go on a special fishing trip, and Joe's father was short of the money it would cost. So he told Joe to give him the money he had saved from the paper route. Joe didn't want to give up going to camp, so he thought of refusing to give his father the money.

2.1 Should Joe refuse to give his father the money? Why?

2.2 Does his father have the right to tell Joe to give him the money? Why?

2.3 Does giving the money have anything to do with being a good son? Why?

2.4 Which is worse, a father breaking a promise to his son or a son breaking a promise to his father? Why?

2.5 Why should a promise be kept?

Instrument III

There was a woman who had very bad cancer, and there was no treatment known to medicine that could save her. Her doctor knew that she had only about six months to live. She was in terrible pain, but she was so weak that a good dose of a pain-killer like ether or morphine would make her die sooner. She was delirious and almost crasy with pain, and in her calm periods, she would ask the doctor to give her enough ether to kill her. She said she couldn't stand the pain and she was going to die in a few months anyway.

3.1 Should the doctor do what she asks and give her the drug that will make her die? Why?

3.2 When a pet animal is badly wounded and will die, it is killed to put it out of its pain. Does the same thing apply here? Why?

3.3 Would you blame the doctor if he gave her the drug? Why?

3.4 Do you think the woman should have the right to decide to die or should the right to decide he up to her husband, to the doctors or so the courts? Why?

3.5 The dector Lills the works and is brought to court. He is

found guilty of murder. The usual sentence is life imprisonment. What should the judge do? Why?

Instrument IV

Judy was a twelve-year-old girl. She had saved up from babysitting and lunch money for a long time so she would have enough money to buy a ticket to a special out-of-town rock concert that was coming to her town. She had managed to save up the \$5 the ticket cost plus another \$3. Her mother had promised her that she could go to the rock concert if she saved the money herself. Later her mother changed her mind and told Judy that she had to spend the money on new clothes for school. Judy was disappointed, and decided to go to the concert anyway. She bought a ticket and told her mother that she had only been able to save \$3. That Saturday she went to the performance and told her mother that she was spending the day with a friend. A week passed without her mother finding out. Judy then told her older sister, Louise, that she had gone to the performance and had lied to her mother about it.

- 4.1 Should Louise, the older sister, tell their mother that Judy had lied about the money or should she keep quiet? Why?
- 4.2 What would be the best reason for Louise to keep quiet? Why?
- 4.3 Louise has to think about how it would influence Judy in the future if she told. What influence on Judy's future should Louise consider? Why?

4.4 Why should a promise be kept?

Form B

Instrument V

In Korea, a company of Marines was way outnumbered and was retreating before the energy. The company had crossed a bridge over a river, but the energy were mostly still on the other side. If someone went back to the bridge and blew it up, with the head start the rest of the men in the company would have, they could probably then escape. But the man who stayed back to blow up the bridge would probably not be able to escape aliver there would be about a four to one chance he would be killed. The captain himself is the men who knows best how to lead the retreat. He asks for volunteers, but no one will volunteer. If he goes himself, the men will probably not get back safely and he is the only man who knows how to lead the retreat.

5.1 Mould the captain order a map to go on this very dangerous mission or should be go himself? Why?

- 5.2 What is the best reason for saying it is right to send someone besides himself?
- 5.3 What is the best reason for saying it is wrong to send someone else?
- 5.4 Does the captain have the right or the authority to order a man if he thinks it best to? Why?

5.5 Would a man have the right to refuse such an order? Why?

Instrument VI

The captain finally decided to order one of the men to stay behind. He thought he could pick one of his two demolition men. Both of these men were trained to use dynamite to blow up bridges or fortifications at the least risk to themselves. One of the demolition men had a lot of strength and courage but was a bad troublemaker. He was always stealing things from the other men, beating them up and wouldn't do his work around camp.

The second demolition man he thought of had got a bad disease in Korea and was likely to die in a short time anyway, though he was strong enough to do the job.

6.1 Sould the captain send the troublemaker or the sick man? Why?

6.2 Whose life is worth more, the troublemaker or the sick man? Why is that?

- 6.3 The captain knows that the sick man has a family, the troublemaker doesn't. Should that enter into his decision. Why?
- 6.4 The captain knows that the troublemaker was a scientist before he joined the army. He was working on a cure for cancer. Should that enter into the decision? Mhy?

Instrument VII

Two young men, brothers, Mid got into minimum trouble. They were secretly leaving town in a harry and mesded money. Karl, the older one, broke into a store and stole \$500. Bob, the younger one, went to a retired old man who was known to help people in town. Bob told the man that he was very stak and he needed \$500 to pay for the operation. Really he wasn't sick at all, and he had no intention of paying the man back. Although the man didn't know Bob wary well, he lowned him the money. So Bob and Karl shipped town, each with \$500.

T.1 If you had to say she did worse, would you say that Karl did

worse by breaking into the store and stealing \$500, or did Bob do worse by borrowing the \$500 with no intention of paying it back? Why?

- 7.2 Would you feel like a worse person, stealing like Karl or cheating like Bob? Why?
- 7.3 Why shouldn't someone steal from a store anyhow?
- 7.4 Who would feel worse, the store owner who was robbed or the man who was cheated out of the money? Why?
- 7.5 Which should the law be more harsh or strong against, stealing like Karl, or cheating like Bob? Why?

Instrument VIII

In a country in Europe, a poor man named Valjean could find no work, nor could hisesister and brother. Without money, he stole food and medicine that they needed. He was captured and sentenced to prison for six years. After a couple of years, he escaped from the prison and went to live in another part of the country under a new name. He saved his money and slowly built up a big factory. He gave his workers the highest wages and used most of his profits to build a hospital for people who couldn't afford good medical care. Twenty years had passed when a tailor recognized the factory owner as being Valjean, the escaped convict whom the police had been looking for back in his home town.

- 8.1 Should the tailor report Valjean to the police? Would it be right or wrong to keep it quiet? Why?
- 8.2 Is it a citizen's duty to report Valjean? Would a good citizen? Why?
- 8.3 If Valjean was a good friend of the tailor, would that make a difference? Why?
- 8.4 Should Valjean be sent back to jail by a judge? Why?



THE ROLE-TAKING INSTRUMENT

This test was originally devised by Flavell (1968).

1. Each S. is shown an ordered series of seven pictures showing a story of a boy being chased by a dog, running down a street, and climbing a tree to eat an apple as the dog trots away.

2. Each S. tells an appropriate story about the pictures.

3. Three pictures are removed from the set leaving those showing the boy walking, running to the apple tree, climbing it and eating an apple. Although the dog remains in the last but, one picture the 'fear of the dog' motive is removed.

4. Each S. tells a story with regard to how an observer to this new set would tell it. The S. is then asked how the observer to this set would answer the question, "Why did the boy thimb the tree?"



THE FILMSTRIP DILEMMAS USED WITH THE EXPERIMENTAL TREATMENT GROUP

1

These filmstrips were taken from First Things: Values (Guidance Associates, 1972).

The Trouble with Truth. Part I

Patrick and six of his friends visit a fishing area with Dave, their camp counsellor. As a treat they are all to take a boat ride with a fisherman. When Dave and the fisherman leave to check weather conditions at the Coast Guard station, Patrick is left in charge with strict orders that nobody board the boat until they get back, or nobody gets a boat ride. However three children climb aboard, getting off just before Dave and the fisherman return. Patrick has to decide whether or not to tell Dave and the fisherman that three children boarded the boat.

You Promised. Part I

Alex has been visiting Russell on his farm for a week during school vacation. Everyday Russell has been riding King and Alex has been riding Midget, a little pony. Russell has promised Alex that he can ride King, the big horse, on the final day of his visit. However Russell's older brother, Zack, needs King to compete in a horse show. Russell has to decide whether to honor his promise to Alex or whether to allow Zack to have the horse for the competition.

But It Isn't Yours. Part I

Jason is taking a brand new sledge home for his brother's birthday. He decides to enter in a sledging race, despite his friend's Lionel's objections. Losing the race he tries the sledge on Suicide Hill and breaks it. At the community hall he cannot find any suitable wood with which to repair the sledge until Karen and Matt find an appropriate piece which belongs to Lionel. Jason has to decide whether or not to use Lionel's wood.

But It Isn't Yours. Part II

Diama and Connie are helping Diama's older brother Doug clean out the basement. Larry, Doug's younger brother, is asked to help but decides to play instead. He is told that anything considered junk, even if it is his, will be thrown put. Diama and Connie discover a bike that Doug had given to Larry on the condition that he repair it. It is not repaired but Diama and Connie mend it and ride it when Larry sees the girls riding 'his' bike. Doug has to decide to whom the bike really belongs.

What Do You Do About Rules? Part I

Cheetah, a superhero, is a member of the Cat People. The Cat People aid those in trouble and fight crime. Each member of the Cat People is forbidden to reveal his true identity. Cheetah is really Sam Wilson, but not even his family know he is Cheetah. One evening when waiting for his son Marcus at a bank, Sam notices a robbery taking place. As Cheetah he captures and ties up the robbers but hearing police sirens he changes back to Sam Wilson and leaves the bank through a side window. Marcus sees him and thinks that a robber is escaping but then realizes that the 'thief' is his father. Sam has to decide whether or not to reveal his Cheetah identity. 183

)



GAMES PLAYED BY THE PLACEBO GROUP

1. Kirman, J. <u>Canada's Prairie Wheat Game</u>. Edmonton, The University of Alberta: Author, no date.

185

Players move around a board, buying land, seed, fertilizer and crop insurance. They experience monetary loss throuch such hazards as drought or hail, or monetary reward through the sale of commodities. Finally they sell their wheat at current market prices, the player making the most money being the winner.

2. Krupa, W. The Land Use Game. Middletown, Conn.: Education Ventures Inc., 1971.

Groups of players build new housing developments on a map. The map is zoned, each zone losing so many environmental points if it is built upon. The team building the subdivision with the least number of environmental points is the winner.

3. The NASA Space Game. (Author and publisher unknown.)

Each player pretends that he has crash landed on the moon and has to reach a rescue ship which is two hundred miles away. Each player ranks fourteen items in terms of the necessity to him for survival. The players then must arrive at a group consensus regarding the ranking of each item within a given time period. Rankings are then compared to those provided by NASA.

4. Lesuring Communication. Trading Post. San Francisco: Lesuring Communication, no date.

Players receive a role card and have to provide basic commodities for themselves, trading their own produced commodities for those needed for survival during the winter. Players move around a board receiving needed commodities. The first to collect all necessary commodities is the winner.

5. High School Geography Project. The Game of Farming, in <u>Geography</u> <u>in an Urban Age</u>, Unit 2, Manufacturing and Agriculture. London: <u>Macmillan</u>, 1969.

Players pretend that they are ploneers in western Kansas in the period 1880-1882. Each player has to decide how to manage a farm. To do this he is given \$1,500, \$500 of which is allocated to living expenses and the rest is spent on various crops or livestock. When all the money has been spent or put in savings an outcome card is collected. This card, which gives a multiplier for each crop or livestock, reflects variations in weather, plant or animal disease and insect damage, as well as variation in market price. The player computes his total income for the year and then discusses his outcomes with the other players in order to ascertain the factors which led to profit or loss situations. Each player then utilizes his 1880 income to the 1881 farming year. Following the 1881 outcome, a discussion, and the allocation of money to the 1882 year, a final outcome is reached in 1882 at which time the player with the most money is declared the winner. This game was played during two sessions.

APPENDIX I

SAMPLE PRETEST RESPONSES BY STAGE OF RESPONSE TO THE ISSUES OF LIFE AND PUNISHMENT IN THE HEINZ DILEMMA

SAMPLE PRETEST RESPONSES BY STAGE OF RESPONSE TO THE ISSUES OF LIFE AND PUNISHMENT IN THE HEINZ DILEMMA

188

Life

Stage 1. No, because if he got caught he'd be put in jail. (Subject 5A)

Stage 2.

If he loves his wife it's his duty to get (the drug) like . . if he really likes her then he can get it but if he really doesn't care for her, well, it's his opinion if he wants to get it and save his wife's life then he can get the drug but if he really doesn't really love her or care for her then guess it isn't if he doesn't want to. (Subject 18B)

Stage 3,

. I think I would have done the same thing (steal the drug) but actually he shouldn't have done it because it's still breaking the law. I think he was right but a judge might think he was wrong, because if someone's going to die and you do something to save their life even if it means you break into something or crash you still saved a person's life and that's all that matters. (Subject 18A)

Punishment

- Stage 1. Send him to jail if he tried to steal it. Well, he committed. a crime, he stole. What he if stole the wrong drug too? (Subject 5A)
- Stage 2. Let him go, but on bail, but not that much. He should sentence the druggist because he was the one that raised the price so much. (Subject 8A)



SAMPLE RESPONSES TO THE FLAVELL ROLE-TAKING INSTRUMENT

Category 1. Subject 10A

Question: What kind of story would Mr. A. tell if he saw these four pictures?

Response: I think he'd say there was a boy named Jack walking down the street, he climbed up a tree because there was the dog after him. While he was waiting for the dog to go he eats an apple.

Question: Why would Mr. A. think Jack climbed the tree?

Response: He was afraid the dog might attack him.

Category 2. Subject 20B

Question: What kind of story would Mr. A. tell if he saw these four pictures?

Response: Tim was walking along a street and he saw a tree and a dog and he was scared of the dog so he climbed up the tree and ate apples.

Question: Why would Mr. A. think Tim climbed the tree?

Response: Because he sees the dog or maybe he's looking for somebody. No, he's scared of the dog, I think. Or he wants to eat apples, or . . .

Category 3. Subject 7A

Question: What kind of story would Mr. A. tell if he saw these four pictures?

Response: The boy was walking and saw an apple tree, he climbed up it and got an apple.

Question: Why would Mr. A, think the boy climbed the tree?

Response: Maybe he was hungry.

NUMBER OF SCORABLE RESPONSES AT EACH MORAL JUDGMENT STAGE FOR EACH SUBJECT ON THE PRE, POST AND FOLLOW-UP KOHLBERG MORAL JUDGMENT TESTS

APPENDIX K

1 2 7 4 2(3) 215 1 1 6 4 2(3) 226 2 11 4 2(3) 226 5 2 11 4 2(3) 226 5 2 4 7 2(1) 177 1 2A 4 4 4 1.72-3 198 2 2 10 2 183 1 2 183 1 4A 2 10 2 183 1 2 182 2 4 2 10 2 182 2 3 1 2 182 2 1 5 10 1 2 206 1 2 1 1 182 2 3 1 1 18 3 1	À	Subject	Stage 1	LOW-UP K		age 3	· · ·	·	Stage Score	MMS*	Test**	- ,
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		1A (6	~		t .		2(3)	226	. 2	-
3A 4 4 4 1-2-3 198 2 1 9 6 2 (3) 231 1 4A 2 10 2 182 2 6 6 2 (3) 226 3 5A 3 7 2 (1) 170 2 5A 3 7 2 (1) 170 2 5A 3 7 2 (1) 180 3 6A 3 .8 1 2 (1) 180 3 6A 3 .8 1 2 (208 1 7A 9 .2 .2 208 1 7A 9 .2 .1 182 2 7A 9 .2 .1 118 2 8 5 .1 .2 130 1 7A 9 .2 .1 1188 2 8 .1 .2 .2 .2 .2 .2 9A .1 .2 .2 .2<		2A	4	7	• •	1			2(1)*.	163	2	3
4A 2 10 2 182 2 6 6 7 2(3) 226 3 5A 3 7 2(1) 170 2 5A 3 7 2(1) 170 2 5A 3 7 2(1) 180 3 6A 3 .8 1 2(1) 180 3 6A 3 .8 1 2(1) 180 3 6A 3 .8 1 2(1) 182 2 6A 3 .8 1 2(1) 182 2 6A 3 .8 1 2(1) 182 2 6A 3 .8 1 118 2 7A 9 2 .1 118 2 8 5 .1(2) 138 3 8A 4 7 2 200 1 9A 7 4 2 2 1060 2 9A 7	•	3A	2 4 2	· 4 ·	•	-			1-,2-3	198		>
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	• •	. 4A (1 2	10	¢		•		2	182		1
10 1 2 200 1 $6A$ 3 $.8$ 1 $2(1)$ 182 2 11 3 1 2 232 3 9 $.4$ $1(2)$ 130 1 $7A$ 9 2 $1(2)$ 130 1 $7A$ 9 2 $1(2)$ 130 1 $8BA$ 4 7 2 200 1 $8A$ 4 7 2 200 1 $9A$ 7 4 2 200 1 $9A$ 7 4 2 200 1 $9A$ 7 4 2 $2(1)$ 160 2 $9A$ 7 4 2 $2(1)$ 160 2 $9A$ 7 4 2 $2(1)$ 182 2 $10A$ 3 8 1 $2(1)$ 182 2		5 A	3	7	•	1	•		2(1)	170-	2	
7A 9 2 1 118 2 8 5 1(2) 138 3 $8A$ 4 7 2 200 1 $8A$ 4 7 2(1) 163 2 3 10 2 2(1) 163 2 $9A$ 7 4 2 2(1) 160 2 $9A$ 7 4 2 2(1) 160 2 3 10 1 2 184 1 $10A$ 3 8 1 2(1) 182 2		6A	3	.8	•	1 1 3	e- 1		2(1)	182	2	
8A 4 7 2(1) 163 2 3 10 2 176 3 12 12 2 2 206 1 $9A$ 7 4 2 2(1) 160 2 $2A$ 7 2 2(1) 160 2 2 $9A$ 7 4 2 2(1) 160 2 2 7 2 2 190 3 3 10 1 2 484 1 $10A$ 3 8 1 2(1) 182 2	i.	7A	9	-	•				1	118	2.	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		8A •		7			•		2(1)	163 •		
10A 3 8 1 2(1) 18 2 2	•	9A #	1/* 7 2	12 4 7		2 2 2			2(1)	160	1 2 3	и
		10A	*3 3 •	8		1 1 3			2(1)	182	1 2 3	

•

 $_{I}$

.

.

÷.

. •

• • •

.

Subjec	t Stage 1	Stage 2	Stage 3	Stage 4	Ștage Score	MMS*	Test**	
	2	10	11	×	·2	191	. 1	<u>_</u>
11A	5 ້	6			2(1)	154	2	
	6	9		•	2(1)	180	3.	•
مر	-	-		*		150		
.	5 7	5	_		1-2		1	,
12A	- 6	4 6	•		1(2) 1-2	135 150	2 - 3	· •
	0	U	•		1-2	100	.	
	•2	10			2	182	1	
13A	3,	. 4	1		2(1)	174	2	
	2	10	•		2	182	3,	
•	• · · · ·			· ,				
	4	11	I	•	2(1)	163	1	
14A	5	7		•	2(1)	157	2	
	, 6	10			2(1)	162	3	
					<u>_</u>	101	3 /	
15A	1 5	11 6	-	1	2	191 15′4	$\frac{1}{2}$	
IDA	ن ب	10			2(1) 2	194 190	3	
,	, 1					170		
	1	11	2	•	2	206	1	
16A	3	7	3	<u>s</u>	· 2	198	2	•
	1	11	2		2 . ື	206	3	•
	3 ¹	10	-		.			
17A	1	12	3		2 2	212 210	1 2	
1.10	.3	7	.3		2	198	3	
	·	· · · · ·			_		_	
, ,	2	• 10 9	3		.2	206	1	•
184			ູ 2 🕐	· .	2	219	2	
•	1	10	1		2	198	3	•
		7	E		2(2)	240	1	¥
_1B	•	. 10	5	•	2(3) 2	193	1 2	. .
, ~~ .		7	4	• • • •	2(3)	236	3	
на на 1910 — До			•					
	• •	4	7	2	3(2)	273	1	
2B	ì	5	7	1	3(2)	239	2	•
		6	10	2	3(2)	274	j 3. w.	•
		••						
35		10	. 4	•	2(3)	227	1	:
3B		7 5	5		2 (3) 3 (2)	227 297	⊲2 3	
		.				231		1
* Mora	1 maturity	score		1	4 .			
								•
	Pretest + Posttest				•			:
	Follow-up't		• • • • • • • • • • • • • • • • • • •			1		

	Subject	Stage 1	Stage 2	Stage 3	Stage~4	Stage Score	MMS*	Test**
		, ,	11	3		2	221	1
	4B		7	5		2(3)	241	
			8	4	•	2(3)	237	2 3
		•	3	7		3 (2)	280	1
ŧ.	5B		4	5			• 254	, 2
	s.		3	8	· ·	3 (2)	270	3
		1	6	4 -	•	2 (3)	227	1.
	6B	1	6	6		2-3	,237	2
	-	2	. 7	4		2(3)	214-	3
			11	5		2 (3)	230	· 1
	7B		2	.9		3(2)	281	2
	•		8	. 4	· · · · · · · · · · · · · · · · · · ·	2(3)	232	3
			3	8	1	3(2)	272	1
	8B			6		2(3)	240	1 2
		j.	8 2	10		3(2)	287	3
	:	2	.11	1		2	179	1
	9B	3	ל	3		2	198	2
		. 1 ,	11	2	•	2	206	3
	•	1	9	3		2(3)	210	1 .
	10B 🕈	1. 1	8	2 3		2	208	2
	· • •		9	3	e.	2(3)	,225	3
	к. т. А		8	5	· · ·	2(3)	228	1
	(11B		6 9	4	*	2(3)	240	2
	•		9	6		2(3)	240	3
	· · · · ·	2	11		•.	2	184	1
	128• >	3 .	6	· · ·		2(1)	165	් 2 3 ්
	•	2	6	1		2	188	3
	1 7 -	1	8	1	•	2	200	1
	13B	•	5	6		2(1)	154	2
			9	, ,	14. 14 24. 14 24.	2	210	3
	1.45	2	9	3			206	1
	14B	,	• 9	1		2	210	2
4		1)	8	4 .	• •	2	208	3
		aturity	score	•••				3
	** 1 = Pre		•	• • •	- - -	. •		•
		ittest low-up t				et i i i		

•

.

•

-

1

• 5

æ

0

4

•1

,

: . \land ٠,

Subject	Stage 1	Stage 2	Stage 3	Stage 4	Stage Score	MMS*	Test**
	1	7	2	an a	2	208	1
15B		10	2		. 2	205	2
		11	1		2	207	3
		6	7		3 (2)	273	1
16B		2	10		3	297	2
		5	11	٢	3(2)	268	3
	2	11	1		2	192	1
17B	.2	10	1		2	181 -	2
	1	11			2	191 -	3
	L	11	4	1	2(3)	221	1
18B		6	4	2	2(3)	240	2
۲.		10	4		2(3)	227	3 🦏
		7	7		2- <u>3</u>	250	· 1
19B	2	8	4		2(3)	204	2
	•	8	6	·	2(3)	243	3
		9	6		2(3)	240	1.
20B		4	8	· •	3(2)	264	2 /1
		7	9		3(2)	255	• 3 ,

Ð

Ţ.

* Moral maturity score

** 1 = Pretest

ť

2. = Posttest

3 = Follow-up test

195

3.